

THESIS

Continuous Offline Handwriting Recognition using Deep Learning Models

Author

Jorge Juan Sueiras Revuelta

Information technologies and Communications program

Spain

2021

Acknowledgements

Una tesis es un largo viaje, quiero dar las gracias aquí a las personas que me han acompañado y apoyado a lo largo del mismo.

En primer lugar, quiero dar las gracias a mis directores de tesis José Vélez y Ángel Sánchez, sin los cuales este trabajo no hubiera sido posible. El buen talante siempre presente y la libertad que me disteis para perseguir mi curiosidad y mis intereses, han sido fundamentales para que este viaje haya sido una gran experiencia. José, gracias por compartir tu sabiduria sobre tantas áreas, pero especialmente en visión por computador, por tus consejos y por estar siempre ahí. Ángel, gracias por el apoyo y la crítica constructiva que tanto han ayudado en la publicación de los resultados de esta tesis.

Gracias a mi mujer Lua, que ha sido mi consejera y confidente y me ha apoyado incondicionalmente desde el principio hasta el final del viaje. Gracias también a mis hijos Mario, Diego y Tello que han aportado el contrapunto necesario al trabajo.

Gracias también a Victoria, con la que compartí una parte de este camino al inicio. Juntos aprendimos a resolver los problemas que van apareciendo durante el desarrollo de una tesis.

Gracias a Daniel Vélez, por presentarme a su hermano José. Ese fue realmente el punto de inicio de este viaje.

Gracias a Rocío por aportar su experiencia investigadora que me ha ayudado a enfocar mejor mi trabajo.

Gracias también a todas aquellas personas que he conocido durante esta andadura y que me han hecho el trabajo mas fácil y ameno, en particular a Juanjo.

Abstract

Handwritten text recognition is an open problem of great interest in the area of automatic document image analysis. The transcription of handwritten content present in digitized documents is significant in analyzing historical archives or digitizing information from handwritten documents, forms, and communications. The problem has been of great interest since almost the beginning of the development of machine learning algorithms. In the last ten years, great advances have been made in this area due to applying deep learning techniques to its resolution.

This Thesis addresses the offline continuous handwritten text recognition (HTR) problem, consisting of developing algorithms and models capable of transcribing the text present in an image without the need for the text to be segmented into characters. For this purpose, we have proposed a new recognition model based on integrating two types of deep learning architectures: convolutional neural networks (CNN) and sequence-to-sequence (seq2seq) models, respectively. The convolutional component of the model is oriented to identify relevant features present in characters, and the seq2seq component builds the transcription of the text by modeling the sequential nature of the text.

For the design of this new model, an extensive analysis of the capabilities of different convolutional architectures in the simplified problem of isolated character recognition has been carried out in order to identify the most suitable ones to be integrated into the continuous model. Additionally, extensive experimentation of the proposed model for the continuous problem has been carried out to determine its robustness to changes in parameterization. The generalization capacity of the model has also been validated by evaluating it on three handwritten text databases using different languages: IAM in English, RIMES in French, and Osborne in Spanish, respectively.

The new proposed model provides competitive results with those obtained with other well-established methodologies and opens the door to new lines of research focused on applying seq2seq models to the continuous handwritten text recognition (HTR) problem.

Resumen

El reconocimiento de texto manuscrito es un problema abierto y de gran interés en área del análisis automático de documentos. La transcripción del contenido manuscrito que aparece escrito en documentos digitalizados es especialmente importante en el análisis de archivos históricos o en la digitalización de la información de documentos, formularios y comunicaciones manuscritas. El problema ha tenido un gran interés desde prácticamente los inicios del desarrollo de algoritmos de aprendizaje automático. En estos últimos 10 años se han producido grandes avances en el mismo a raíz de la aplicación de técnicas de aprendizaje profundo a su resolución.

En esta Tesis se aborda el problema del reconocimiento de texto manuscrito continuo, consistente en desarrollar algoritmos y modelos capaces de transcribir el texto presente en una imagen sin necesidad de que dicho texto esté segmentado en caracteres. Para ello, se ha propuesto un nuevo modelo de reconocimiento basado en la integración de dos tipos de arquitecturas de aprendizaje profundo, las redes convolucionales y los modelos sequence-to-sequence (seq2seq), respectivamente. El componente convolucional del modelo se orienta a identificar las caracteristicas relevantes de los caracteres presentes en el texto y el componente seq2seq construye la transcripción del mismo modelando la naturaleza secuencial del texto.

Para el diseño de este nuevo modelo se ha realizado un extensivo análisis de las capacidades de distintas arquitecturas convolucionales en el problema simplificado de reconocimiento de caracteres aislados con el objetivo de identificar las más adecuadas para integrarlas en el modelo continuo. Adicionalmente, se ha realizado una extensiva experimentación del modelo propuesto para el problema de reconocimiento de texto continuo a nivel de palabras, lo que ha permitido determinar su robustez frente a cambios en la parametrización. También se ha validado la capacidad de generalización del modelo mediante la evaluación del mismo usando tres bases de datos de texto manuscrito en diferentes idiomas: IAM en inglés, RIMES en francés y Osborne en español, respectivamente.

El nuevo modelo proporciona resultados competitivos con los obtenidos aplicando otras metodologías bien establecidas y abre nuevas puertas a líneas de investigación centradas en la aplicación de modelos seq2seq al problema del reconocimiento de texto manuscrito continuo.

Contents

1	Intr	oducti	ion	1
	1.1	Motiv	ation	1
	1.2	Resear	rch objectives	6
	1.3	Thesis	soutline	7
2	Ana	alysis c	of the offline handwritten text recognition problem	9
	2.1	Overv	iew of the problem	9
	2.2	Handy	written text image preprocessing	13
		2.2.1	Remove the noise \ldots	14
		2.2.2	Slope correction	14
		2.2.3	Slant correction	15
		2.2.4	Normalize ascenders and descenders regions	15
		2.2.5	Crop and resize input images	16
	2.3	Data a	augmentation strategies	16
	2.4	Decod	ing predictions	17
	2.5	Evalua	ation metrics	18
		2.5.1	Metrics to evaluate character recognition models	18
		2.5.2	Metrics to evaluate word recognition models	21
3	Har	ndwriti	ing Databases	23
	3.1	Handy	writing characters databases	23
		3.1.1	NIST database	23
		3.1.2	MNIST database	25
		3.1.3	TICH database	26
		3.1.4	New COUT database	26
	3.2	Gener	al handwritten text databases	28
		3.2.1	IAM Database	29
		3.2.2	RIMES database	31
		3.2.3	Osborne database	32
	3.3	Datab	ases summary	34

4	Rev	riew of	Handwriting recognition approaches	37
	4.1	Introd	luction	37
	4.2	Data	normalization background	39
		4.2.1	Noise removal	40
		4.2.2	Slope correction	41
		4.2.3	Slant correction	42
		4.2.4	Normalize ascenders and descenders	43
	4.3	Data a	augmentation	43
	4.4	Classi	cal approaches based on HMM	44
		4.4.1	Overview of Hidden Markov Models (HMMs)	45
		4.4.2	Hidden Markov Models in handwriting recognition	46
	4.5	Neura	l networks and deep learning architectures	47
		4.5.1	Historical background	48
		4.5.2	Multilayer perceptrons	49
		4.5.3	Recurrent Neural Networks	52
		4.5.4	Convolutional Neural Networks	57
		4.5.5	Multidimensional Recurrent neural netwoks	60
		4.5.6	The CNN-LSTM-CTC Architecture	63
	4.6	Conne	ectionist Temporal Classification (CTC)	65
	4.7	Soluti	ons based on Nearest Neighbors	67
	4.8	seq2se	eq architectures	68
		4.8.1	Overview of seq2seq architectures	68
		4.8.2	seq2seq Encoder	69
		4.8.3	seq2seq Decoder	70
		4.8.4	seq2seq Attention Mechanism	72
		4.8.5	seq2seq in handwriting	74
	4.9	Synth	etic data and domain adaptation	76
	4.10	Langu	age models background	77
5	Dee	p arch	itectures for isolated offline character handwriting recognition	79
	5.1	Introd	luction	79
	5.2	COUT	Γ character database generation	81
		5.2.1	Extraction of characters from UNIPEN online	81
		5.2.2	Transformations applied to original images	82
	5.3	Convo	lutional architectures	83

		5.3.1	LeNet architecture
		5.3.2	VGG architecture
		5.3.3	ResNet architecture
	5.4	Experi	mental results
		5.4.1	Baseline results on MNIST database
		5.4.2	Comparative results on NIST database
		5.4.3	Comparative results on TICH database
		5.4.4	Results on the new COUT database
	5.5	Discus	sion of results
0	Л	,	·· · · · · · · · · · · · · ·
0	Dee tion	p arch	itectures applied to continuous offline handwriting recogni- 105
	6.1	Model	architecture
		6.1.1	Detailed schema of the model
		6.1.2	Convolutional reader
		6.1.3	Encoder
		6.1.4	Decoder
		6.1.5	Attention mechanism
	6.2	Experi	mental setup
		6.2.1	Source code and software
		6.2.2	Implementation details $\ldots \ldots 116$
		6.2.3	Image preprocessing
		6.2.4	Data augmentation $\ldots \ldots 122$
		6.2.5	Lexicons
	6.3	Experi	ments
		6.3.1	Baseline experiments
		6.3.2	Image segmentation: step size
		6.3.3	Image segmentation: patch width $\ldots \ldots 129$
		6.3.4	Convolutional architectures
		6.3.5	Encoder architecture $\dots \dots \dots$
		6.3.6	Image resolution
		6.3.7	Normalization and data augmentation
		6.3.8	Teacher forcing
		6.3.9	Final model
		6.3.10	Error analysis

	6.4	On the	e difficulties of comparing results	. 143
		6.4.1	Choice of train, validation and test partitions $\ldots \ldots \ldots \ldots$. 144
		6.4.2	Filter transcription errors in IAM database	. 146
		6.4.3	Character set selected for decoding $\ldots \ldots \ldots \ldots \ldots \ldots$. 147
		6.4.4	Data augmentation	. 148
		6.4.5	Data normalization	. 149
		6.4.6	Decoding with lexicons and language models	. 149
	6.5	Compa	arative of results	. 150
		6.5.1	Best practices recommendations to results comparative	. 150
		6.5.2	Historical results comparative for RIMES database $\hfill \ldots \hfill \hfill \ldots \hfill \ldots \hfill \hfill \ldots \hfill \hfill \ldots \hfill \hfill \ldots \hfill \hfill \ldots \hfill \hfill \hfill \ldots \hfill \hfi$. 151
		6.5.3	Historical results comparative for IAM database \hdots	. 153
	6.6	Discus	sion	. 156
7	Con	clusio	ns	159
	7.1	Future	e work	. 160
	7.2	Public	ations	. 162
Bi	bliog	raphy		163
In	dex			185

List of Tables

3.1	Digits frequency in the NIST train and test partitions	24
3.2	Letters frequency in the NIST train and test partitions	25
3.3	Character frequency in the train partition of TICH database	27
3.4	Character frequency in train partition of COUT database	28
3.5	Distributions of authors, pages, lines and words in the IAM official partition.	30
3.6	Distributions of pages, lines and words in the IAM Aachen partition. $\ . \ .$	30
3.7	Frequency of punctuation symbols that appears as a segmented words in the IAM database.	31
3.8	Frequency of accented vowels and punctuation symbols in the train partition of the RIMES database.	32
3.9	Character frequency in the Osborne database	34
3.10	Summary for the isolated character databases	35
3.11	Summary for the continuous handwriting text databases	35
5.1	Characters contained in the COUT database.	82
5.2	Configurations of the LeNet architecture used in the different databases. $\ .$	85
5.3	Summary of layers, output shape and number of parameters for the pro- posed LeNet architecture applied over MNIST database.	85
5.4	Accuracy of the VGG architecture over the COUT upper and lower case database. Comparative over the number of blocks	87
5.5	Summary of layers, output shape and number of parameters for the selected VGG architecture applied on upper and lower case letters of the COUT	07
-	database.	87
5.6	Accuracy with the ResNet model over the NIST upper case database. Com- parative over the number of blocks.	88
5.7	Summary of layers, output shape and number of parameters for the ResNet architecture applied over the NIST upper case database.	90
5.8	Accuracy by architecture over the MNIST database	91
5.9	Accuracy by architecture over the NIST upper case dataset	93
5.10	Accuracy by architecture over the NIST lower case dataset	94

5.11	Accuracy by architecture over the NIST upper and lower case letters dataset. 96
5.12	Accuracy by architecture over the TICH database
5.13	Accuracy by architecture over the new COUT upper case dataset 99
5.14	Accuracy by architecture over the new COUT lower case dataset 99
5.15	Accuracy by architecture over the new COUT upper and lower case dataset. 100
5.16	Accuracy by architecture over the new COUT database
5.17	Accuracy by input resolution on the COUT test partition 102
6.1	Software employed in the experiments
6.2	List of excluded pages of the LOB corpus used in the IAM database. $\ . \ . \ . \ 124$
6.3	Baseline results with no lexicon
6.4	Baseline results using standard lexicon
6.5	Baseline results using the test lexicon
6.6	Step size experiments results (baseline in step size of 2). Best result, by database, is presented in bold
6.7	Patch width experiments results (baseline in patch width of 10). Best result by database in bold
6.8	Convolutional architectures experiments results (baseline in Lenet and patches). Best result by database in bold
6.9	Encoder size experiments results (baseline in encoder size of 256). Best result by database in bold
6.10	Number of layers and bidirectionally experiment results (baseline in 2 bidi- rectional layers). Best result for each database in bold
6.11	RNN encoder type experiments results (baseline in LSTM bidirectional). Best result by database in bold
6.12	Input image resolution experiments results (baseline in 48x192). Best result by database in bold
6.13	Normalization and data augmentation experiments results (baseline with normalization and no data augmentation). Best result by database in bold. 138
6.14	Input image resolution experiments results (baseline with teacher forcing). Best result by database in bold
6.15	Optimal results without lexicon
6.16	Optimal results using a standard lexicon
6.17	Optimal results using test lexicon
6.18	Cross table of lines count between Official and Aachen partitions of IAM database
6.19	Results comparative between the Official and Aachen IAM partitions 146
6.20	RIMES word level results. Visual model. No lexicon or LM

6.21	RIMES word level results. Lexicon or LM not including test data	152
6.22	RIMES word level results. Test data included in lexicon or LM. $\ .\ .\ .$.	153
6.23	IAM-Aachen test partition results. Word level results. Visual model, no lexicon or LM	154
6.24	IAM-Aachen test partition results. Word level results. Lexicon or LM not include test data.	154
6.25	IAM-Aachen test partition results. Word level results. Test data included in lexicon or LM.	154
6.26	IAM-Official test partition results. Word level results. Visual model, no lexicon or LM	155
6.27	IAM-Official test partition results. Word level results. Lexicon or LM not include test data	155
6.28	IAM-Official test partition results. Word level results. Test data included in lexicon or LM.	155

List of Figures

1.1	Historical writing examples. Maya (12th century AC), Egyptian hiero- glyphs (30th century BC) and Georgian manuscript (6th century AC). (Source: Wikipedia)	2
1.2	Computer and mobile keyboards. (Source: Wikipedia)	2
1.3	Natural scenes with handwritten and typographic text. (Source: OpenImages)	3
1.4	Example of personal data collection form containing both handwritten and printed text.	4
1.5	Transkribus interface example image	4
1.6	Image of an original document of the paper 'Digital computers applied to games' from The Alan Turing Digital Archive. The image is also an example of document that combines handwritten and typographic text	5
2.1	Example of offline (left) and online(right) handwriting data.	10
2.2	Example of layout analysis over a scientific paper	10
2.3	Example of scene text detection. Source: maps.google.com	11
2.4	Examples of handwriting (left) and typographic (right) text	12
2.5	Example of line and word segmentation performed on an image of the IAM database.	12
2.6	Calligraphy zones definition	13
2.7	Slant and slope in a handwritten word image	14
2.8	Example of image with noise take from Osborne database	14
2.9	Example of slope angle estimation used in this Thesis	15
2.10	Example of ROC curve and AUC area	21
3.1	Sample images of the NIST database	24
3.2	Sample images of the MNIST database	26
3.3	Sample images of the TICH database	27
3.4	Sample images of the COUT database	28
3.5	Sample image of a page of the IAM database.	29

3.6	Sample images of the IAM database.	30
3.7	Sample error images of the IAM database	31
3.8	Sample images of the RIMES database	32
3.9	Sample images of pages of the Osborne database	33
3.10	Sample images of the Osborne database	33
4.1	Example of slope correction	42
4.2	Example of typical HMM topology for the HTR problem	46
4.3	Feedforward neural network basic architecture	50
4.4	Neural network node detail.	50
4.5	RNN schema: a) compact, and b) unfold-over-time	52
4.6	Bidirectional RNN schema	54
4.7	Multi-layer RNN schemas: a) Stacked RNN. b) Bidirectional stacked RNN.	54
4.8	Long Short-Term Memory (LSTM) cell	55
4.9	Gated Recurrent Unit (GRU) cell	56
4.10	CNN schema for a kernel of 3×3 and one feature map	58
4.11	MDRNN 2 layer context example (adapted from [78])	60
4.12	MDRNN example of contexts. (a) 1-dimensional BRNN. (b) 2-dimensional MDRNN (adapted from [78]).	61
4.13	MDLSTM standard architecture (adapted from [216]).	62
4.14	Standard CNN-RNN-CTC general architecture.	64
4.15	CTC loss calculation: all possible paths to decode the output probabilities sequence are computed.	65
4.16	CTC Decoding path for one word	66
4.17	Basic schema of the Seq2seq architecture	69
4.18	Schema of a LSTM based decoder: (a) train(teacher forcing) and (b) eval-	71
4.19	Sample of text written with a typographic font that emulate the handwrit- ten text.	76
5.1	UML activity diagram of creation and transformation process of the char- acter database. Processes are presented as white boxes and outputs as gray boxes	82
5.2	Example of transformations chain applied to build a training sample	83
5.3	Proposed LeNet architecture schema for the MNIST database	85
5.4	VGG architecture model with the sizes of the layers and the sizes of the convolutional masks (in red). This model is instantiated for an input size of 64×64 and a final dense layer of 52 categories corresponding to the model applied on upper and lower case letters of the COUT database	86

5.5	ResNet architecture: a) schema of a ResNet module and b) schema of the complete architecture with 3 modules
5.6	Confusion matrix (only errors) of the VGG model over the test partition of MNIST database
5.7	Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST upper case letters
5.8	Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST lower case letters
5.9	Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST upper and lower case letters
5.10	Truncated confusion matrix (only errors) of the VGG model over the test partition of TICH database
5.11	Truncated confusion matrix (only errors) of the VGG model over the test partition of COUT lower case letters
5.12	Truncated confusion matrix (only errors) of the VGG model over the test partition of COUT database
6.1	Components of the proposed high-level model architecture for word recog- nition
6.2	Detailed model architecture
6.3	Word patch extraction: (up) original image word, and (down) extracted input model patches
6.4	Residual connection to convert the convolutional output into the feature vectors sequence
6.5	Decoder training diagram
6.6	Decoder score diagram
6.7	Example of attention mechanism outputs for two words and their respective original handwritten word images
6.8	Pre-processing algorithm steps
6.9	Line preprocessing example: (upper) original line and (lower) preprocessing results
6.10	Example of result of the enhanced contrast algorithm used in this Thesis applied on a Osborne database page
6.11	Example of slope angle estimation used in this Thesis
6.12	Slant correction samples over lines of IAM and Osborne databases 120
6.13	Example of normalization of the ascenders and descenders regions 121
6.14	Example of crop an resize word images: original at the left, and resized ones to fixed size at right
6.15	Examples of images generated by the data augmentation algorithm used in this Thesis.

LIST OF FIGURES

6.16	Variation percentage of WER and CER in validation and test partitions for different step sizes over the baseline
6.17	Variation percentage of WER and CER in validation and test partitions with different patch widths over the baseline
6.18	Variation percentage of WER and CER in validation and test partitions with different convolutional architectures over the baseline. $\dots \dots \dots$
6.19	Variation percentage of WER and CER in validation and test partitions with different encoder RNN sizes over the baseline
6.20	Variation percentage of WER and CER in validation and test partitions with different encoder layers over the baseline
6.21	Variation percentage of WER and CER in validation and test partitions with different encoder types over the baseline
6.22	Variation percentage of WER and CER in validation and test partitions with different input image resolutions over the baseline
6.23	Variation percentage of WER and CER in validation and test partitions with normalization and data augmentation over the baseline
6.24	Variation percentage of WER and CER in validation and test partitions with different training strategy over the baseline
6.25	Error analysis: WER in test vs word length
6.26	Error analysis: WER in test vs word length with the test lexicon 143

Algorithms

6.1	Slant angle identification algorithm proposed.	119
6.2	Train algorithm with data augmentation integrated	122
6.3	Data augmentation algorithm proposed.	123

Abbreviations

ANN	Artificial Neural Network
ASR	Automatic Speech Recognition
AUC	Area under the ROC curve
BPTT	Back-Propagation Through Time
BRNN	Bidirectional Recurrent Neural Networks
CE	Cross Entropy
CER	Character Error Rate
CNN	Convolutional Neural Network
CRNN	Convolutional Recurent Neural Network
CTC	Connectionist Temporal Classification
DAS	Document Analysis Systems
FCN	Fully Convolutional Network
FPR	False Positive Rate
GMM	Gaussian Mixture Models
GRCL	Gated Recurrent Convolutional Layers
GRU	Gated Recurrent Units
HMM	Hidden Markov Models
HOG	Histograms of Oriented Gradients
HTR	Handwritten Text Recognition
ICDAR	International Conference on Document Analysis and Recognition
ICFHR	International Conference on Frontiers in Handwriting Recognition
IJDAR	The International Journal of Document Analysis and Recognition
LDA	Linear Discriminat Analysis

LSTM	Long Short-Term Memory
MDLST	M Multidimensional Long Short Term Memory
MDRNN	Multidimensional Recurrent Neural Network
ML	Machine Learning
MLE	Maximum Likelihood Estimation
MLP	Multilayer Perceptron
MoChA	Monotonic Chunkwise Attention
NN	Neural Network
OOV	Out Of Vocabulary
PCA	Principal Component Analysis
PHOC	Pyramidal Histogram of Characters
PPHT	Progressive Probabilistic Hough Transform
ResNet	Residual Networks
RNN	Recurrent Neural Networks
ROC	Receive Operations Characteristics
seq2seq	sequence-to-sequence
STN	Spatial Transformer Network
SVM	Support Vector Machines
TPR	True Positive Rate
WER	Word Error Rate

Chapter 1

Introduction

This chapter describes the main motivations for the research carried out in this Thesis. It also details the objectives for this research and outlines the Thesis chapters.

1.1 Motivation

Literacy is the ability to read and write in, at least, one method of writing, and the first and principal writing method that is learned 'through' education is handwriting. The ability to write is one of the essential knowledge of human beings, recognized as a fundamental right in 1997 by the Hamburg Declaration¹ [211]. Until the invention of the printing press in the 15th century, handwriting text was the only method of transmitting and perpetuating knowledge, stories and beliefs over time. Writing has been present in all cultures and has taken many forms throughout the history, see Fig. 1.1. In fact, the appearance of writing is precisely one of the essential factors in determining the beginning of human history.

Handwriting is present in our daily lives, usually in notes, lists or other short texts in everyday life. But it is also used more systematically in other areas, such as taking notes in academic classes or in business meetings. Moreover, despite the advent of new technologies such as computers, tablets or smartphones, handwriting is still the preferred method for many people to capture their ideas or thoughts, at least initially. Handwriting can also be done virtually anytime and anywhere with a minimum of technology: with a pencil and a notebook, chalk on a wall or the hand in the sand.

In fact, the main disadvantage of the handwriting text is that it must be digitized to facilitate its preservation, arrangement, and dissemination. Now, we are in the digitization era. All the information is stored and indexed in digital formats, and all the business processes must be digital. With all data and knowledge stored in digital databases, multiple advantages are achieved in terms of accessibility and analysis of the information. In this context, the capacity to recognize and digitize the content of handwriting text is necessary to extract shareable knowledge from it.

It is very likely that if the handwritten text could be converted into digital text in a simple way, its use would increase significantly. However, at present, we are far from having a

 $^{^{1}}Literacy$, broadly conceived as the basic knowledge and skills needed by all in a rapidly changing world, is a fundamental human right



Figure 1.1: Historical writing examples. Maya (12th century AC), Egyptian hieroglyphs (30th century BC) and Georgian manuscript (6th century AC). (Source: Wikipedia)

simple and accurate way to convert handwritten text into a digital format. The current common way to generate digital text is through a keyboard like the ones in Fig. 1.2. The keyboard is not natural and it is effective only for experienced people. An alternative for generating digital text is the use of voice. Speech recognition has advanced significantly in accuracy in recent years and it works well with a continuous and linear stream of content [45]. But translating the thoughts into text is rarely a continuous and linear process. This unnatural operation is a drawback in popularizing speech as a digital text generation tool. So is the fact that it cannot be used in all contexts, such as for taking notes in an academic class or in work meetings.



Figure 1.2: Computer and mobile keyboards. (Source: Wikipedia)

Automatic handwritten Text Recognition (hereafter abbreviated as HTR) is a problem that has been active for a long time [165]. It is a relatively simple task for people but very difficult to model. It is currently considered in its general form as an unsolved problem yet and under active research. Several conferences and periodic publications are specialized in handwriting, as the International Conference on Frontiers in Handwriting Recognition (ICFHR) or DAS (Document Analysis Systems), and others dedicate specific tiers to the handwriting recognition problem such as the International Conference on Document Analysis and Recognition (ICDAR). There are also specialized journals in this field like The International Journal of Document Analysis and Recognition (IJDAR). The need to recognize text automatically appears in several contexts. The main problem is recognizing the text in digital images obtained by scanning paper pages from documents, books, letters, or forms. However, the problem also appears with images of real-world scenes that can include posters, labels, or nameplates among others. This need may also appear in retrieving text from videos. For example, in order to provide the autonomous driving systems the capacity to understand the names that appear in informative signs of the road. In the cases of managing real-world scene images, the problem is named Scene Text Recognition. and is an active research area [219] [107] [189]. Fig. 1.3 shows some examples of natural scenes that contains text.



Figure 1.3: Natural scenes with handwritten and typographic text. (Source: OpenImages)

The offline continuous HTR problem [79] consists of the automatic transcription of images that contain handwritten text. The problem has been extensively studied in the literature [180] [165] [79] [232]. This specific problem presents relevant applications such as bank cheque processing [103], automatic address reading [144] or content extraction from digitized databases of historical documents [177]. Despite recent advances, significant differences in the writing of individuals and in the imprecise nature of handwritten characters make this recognition task hard, and it remains as an open research problem [116].

Another relevant application of HTR is the automatic processing of forms such as the one in Fig. 1.4. In countless administrative tasks, forms are usually completed with handwritten text and often have to be digitized afterwards. This digitization is done manually. It is a repetitive and tedious task that consumes large amounts of work time. The automation of the task requires high precision HTR systems, because personal data or contact info like phone number or email usually requires precise transcription.

In the industry, there have been, for many years, solutions to solve the handwriting problem in some specific areas. For example, the French company $A2iA^2$ offers a solution to read bank check amounts. In these cases, where the recognition of handwritten text is

²https://www.a2ia.com/en

Nombre y apellidos:	NonBA	RYA	PACLI	Deg	P	AG NA	7 1
NIF / TR / PASAPORTE:	ME RAG 1						
Teléfono fijo: FAL E	=150 Mi	ivil: TAL	MEVIL	Correo electro	ónico:	LOVEN	renamen
Estado civil: Solte	ro/a	ado/a	Viudo/a	Divorciado.	/a	Pareja de he	echo
Régimen económico matrir	nonial: 🚺	Gananciales	Separaci	ón de bienes			<u> </u>
Tipo de vivienda en la que	reside: 🛛 Prop	ia 🗖 Alquila	ada 👌 Dom	icilio paterno	Dotra	as (especificar) 7	TROVIN OLDO
Nombre y apellidos cónyug	e: NCMDMI	L Y ARC	266100	J BA	L	Cer 40	IME
NIF / TR / PASAPORTE: N	IF anyou	a t					

Figure 1.4: Example of personal data collection form containing both handwritten and printed text.

performed on a limited domain such as numerical amounts, the accuracy of these industrial systems is sufficiently high to surpass manual management.

In recent years industrial products and services have appeared to ally in the general handwriting problem for localizing and read handwriting text in any type of documents. The most relevant are Azure Cognitive Services from Microsoft ³ and the Detect Handwriting in Images service of the Google Cloud Vision API ⁴. Another initiative coming from the academic world is Transkribus, [111] to recognize and index historical handwritten documents. This last one has led to the development of a popular software⁵ for automatic recognition of text in historical documents. Fig. 1.5 shows an example of the transcription interface in Transkribus desktop software.

	Transkribus v	1.3.5.2-SNA	SHOT	30-08-2017-09	506), Loaded doe: Copy of HS_115_TRAIN, ID: 6867, Page 3, Flo: HS_115_3_ebject_128271.jpg [Inoge Meta Info: (Resolution: 300.0, w4h: 2638 * 4362)] [current line: w4h: 1786 * 50] — =
		e e =	Search	conent document	д н 4 ≥ мох и е ⊟ то П
Server Overvie	w Layout Mela	adata Tools			OTR CALL CONTRACTOR CONT
× -	R - E	J			
Туре	Text	Structure	Readin	ID Coord	OR
Page					OW A A
* TextRegion		page-number	1	region_1 2219,	
▶ Line	1		1	ine_145, 2295,	An Jublico Joulicis
* TextRegion		heading	2	region_1 735,1	
+ Line	In Publico Polit		1	ine_145 889,2	A A A A A A A A A A A A A A A A A A A
* TextRegion		heading	3	region 1 182,3	· Intim Staten den 7 - Girline 1797.
* Line				line 145 223,4	or culune orgen for a sunno frya.
Baselin			-	228,4	and the sea of the sea
► Line	Vor Titl Herm		2	inc_145 239,6	101 - How litt Berny Vurdermerten von comico p:
► Line	In Beyseyn de		3	ane_145 227,7	1 how the
► Line	Herm Obertiec		4	ane_145 296.8	en of E & Pa from the tx outle
► Line	Herm Franz v I		5	ine_145 280.1	SIN Denvenne den Stalley that Calles.
+ Line	Herm Dr Andre		6	ine_145 276,1	the terms of terms o
▶ Line	/: Herr Dr Leop		7	ine_145 242,1	he Alendiai tana 4 De l'in China hana)
* TextRegion		heading	4	region_1 76,17	south our cullerant and sugarte
r Line	Titl Hr Bürgerr		1	ine_145 145,1	0 0 7 mm n 7 2 C
+ Line	No 768.		2	line 145 704,1	Berrn Frans & Manne und
* TextRegion		paragraph	5	region 1 107,2	
+ Line	Kreisamtämt. E		1	line 145 191,2	Rommer A. Stadas (Air and f.)
► Line	pro. 30th Junik		2	inc_145 151,2	and a compartie
► Line	ganz recht ges		3	ine_145 174,2	Che Con & product of the contraction of the contrac
► Line	von der durch		4	ane_145 171,2	1 Berr D. Cogold Jorelmann abreesend!
► Line	gükten Gemei		5	ane_145 157.2	
► Line	erstatteten Da		6	line_145 177.2	
+ Line	die von der Sta		7	line_145 167.2	Actum Bozen den 2 ^{ten} jūliūs 1792: ^e
► Line	haltene milde		8	ine_145 157.2	
* Line	dem hierortign		9	ine_145 187,2	Z vor Hti-Herrn-Burgermeister von Remich-etc.:
▶ Line	tentlich von de		10	ine_145 153,3	3 In-Bevsevn-der-Magistrats-Bäthe
► Line	bekannt gemai		11	ine_145 170.3	
► Line	werde daher d		12	ine_145 184,3	4 Herrn Oberlieutenant Karl v Kuglern.
+ Line	Anzeige zur gu		13	line 145 220.3	

Figure 1.5: Transkribus interface example image.

But these general-purpose systems are limited by the current state-of-the-art of the general HTR problem in terms of error. That is why they can only be applied when a certain significant margin of error can be considered acceptable. For example, in the digitization of documents for the application of statistical text analysis models. However, their use is

³https://westus.dev.cognitive.microsoft.com/docs/services

⁴https://cloud.google.com/vision/docs/handwriting

⁵https://readcoop.eu/transkribus

very complicated or not feasible in problems that require high accuracy, for example, the previous case of transcription of form fields.

Arthur Samuel [176] defines *Machine Learning* as the field of study that gives computers the ability to learn without being explicitly programmed. More specifically machine learning consists of the automatic development of computer algorithms capable of reproducing a complex task from previous experiences encoded in data [15]. For example, to teach a machine to play chess, make sure that it follows the game's rules. This can be easily encoded through a standard algorithm. But it is also necessary that the machine can play games reasonably well, and it is not obvious that standard algorithms can achieve this. Machine learning algorithms built by analyzing previous games encoded and stored in digital data are used in cases like this. Alan Turing well defined the different levels at which a machine can approach a complex problem like chess in [193]. An excerpt from the first page of the original typed document of this paper with handwritten notes by the author is included in Fig. 1.6.

"Faster then thought" For Eds. V. Sowder London LES3 Digital Computers applied to Games. AMT/6/7/1 When one is asked 'Could ome make a machine to play chess ?' there are several possible meanings which might be given to the words. Here are a few :-Could one make a machine which would obey the rules 1) of chess, i.e. one which would play random legal moves, or which could tell one whether a given move is a legal one ? Could one make a machine which would solve chess 11) problems, e.g. tell one whether, in a given position, white has a forced mate in three ? 111) Could one make a machine which would play a reasonably good game of chess, i.e. which, confronted with an ordinary (that is, not particularly unusual) chess position, would after two or three minutes of calculation, indicate a passably good legal move? Could one make a machine to play chess, and to iv) improve its play, game by game, profiting from its experience ? To these we may add two further questions, unconnected with chess, which are likey to be on the tip of the reader's tongue. Could one make a machine which would answer questions v) put to it, in such a way that it would not be possible to distinguish its answers from those of a man ? vi); Could one make a machine which would have feelings like you and I do ?

Figure 1.6: Image of an original document of the paper 'Digital computers applied to games' from The Alan Turing Digital Archive. The image is also an example of document that combines handwritten and typographic text.

For years Machine Learning has been successfully applied to structured data in the form of tables. Each table record corresponded to a learning example and each column was a numerical attribute with which to build the algorithm in the form of a mathematical formula. It has been applied for a long time to business problems such as credit scoring [137] or fraud detection [220]. However, machine learning algorithms failed to learn from unstructured data, such as images or text. This has changed drastically in recent years with the emergence of *Deep Learning*. Deep Learning are neural models in which the learning function is defined as the composition of a large number of simpler functions called layers [55]. Understanding function composition in the mathematical sense of $(f \circ g)(x) = f(g(x))$. These Deep Learning models are being successfully used in especially complex machine learning problems such as image classification [132], speech recognition [45] or machine translation [4].

The continuous HTR problem is a decoding problem [81]. An image containing a sequence of strokes is decoded into the sequence of characters that form those strokes. In this sense, it is similar to speech recognition problem, where an audio sequence is decoded into the sequence of the words of this audio. The *machine translation* problem, where a sequence of words in one language is decoded into its translation in another language, present also similarities with the HTR problem. Recent advances in the above problems through Deep Learning models specially oriented to sequence decoding, named as sequence-to-sequence (seq2seq) models [204], can be very relevant to solve the HTR problem.

1.2 Research objectives

This Thesis aims to propose a solution to general handwritten text recognition at the word level. This problem is defined as the development of algorithms capable of transcribing into text any image in a handwritten word, regardless of its content, author, or nature. Additionally, to achieve the above objective, it will be analyzed first the problem of isolated character recognition. To propose this approach, we have taken inspiration from how people learn to read, who first learn to identify each letter separately and then learn the words.

This general objective can be detailed into the following specific sub-objectives:

- Conduct a detailed historical study of the different approaches to the offline continuous HTR problem over the years. This study deepen into the relevant contributions of each approach and identifies the main limitations and criticisms. Having a deep understanding of how this problem has been proposed to be solved allows making the best decisions to address it with a new approach.
- Explore new normalization algorithms of handwritten text images to reduce the variability of the text. In this way, it is intended to facilitate the construction of word recognition models, with particular emphasis on the optimization of algorithms that correct the italic feature of the text.
- Contribute to research in character recognition through the construction of a new image database of handwritten characters. This database will be oriented to evaluate recognition models on the broadest possible set of characters present in handwritten text. It will thus complement the existing character databases.
- Study the application to the problem of character image classification of deep learning models that have worked well in other image classification problems. In partic-

ular, we will analyze convolutional network models, which are the ones that have obtained the best results in this type of problems in the last years [132].

- Study the word recognition problem under the perspective that it is a problem of transforming an input data sequence of word image column pixels into an output sequence of the ordered characters that form the word. Conduct experiments to determine whether this problem could be solved with algorithms used in other similar problems which have more active research, like language translation or speech recognition. These problems have in common with the HTR problem that, in all cases, it is a matter of decode an arbitrary length sequence of input data into another arbitrary length sequence of output data.
- Build sequence-to-sequence (seq2seq) network models applied to the HTR problem. This type of model has achieved essential breakthroughs in the problems mentioned above of language translation [204] or speech recognition [45]. Experiment with different deep learning architectures of such models to determine the optimal configurations when applied to the general HTR problem.

1.3 Thesis outline

This chapter has introduced the motivation of our work. We also propose the main objective and detail the specific objectives of this Thesis.

In Chapter 2 we provide an overview of the offline HTR problem putting it in the context of text recognition analysis in general. Next, we include the details about the experiment design and evaluation metrics.

In Chapter 3 we review the primary databases at character and word level used to evaluate the results of the experiments of the recognition models.

In Chapter 4 we include a detailed review of the works in offline HTR general, focused on the more active research lines of the last years.

Chapter 5 includes our proposed solutions to the offline handwritten recognition of isolated characters.

Chapter 6 details the solutions and seq2seq model architectures that we propose to solve the handwriting recognition problem at the word level. We also analyze the difficulties existing in the literature to identify comparable results between authors, proposing a framework of standardization of results to facilitate such comparisons.

Finally, in Chapter 7 we include our conclusions from this work and the future work is proposed.

Chapter 2

Analysis of the offline handwritten text recognition problem

This chapter analyzes the general offline HTR problem. This is defined and contextualized in the more general framework of text analysis and recognition. Section 2.1 introduces the problem, contextualizes and narrows it to define the framework of this Thesis. Next, all the elements necessary for the experiments are defined. In Section 2.2 image processing techniques applied as image preprocessing are introduced. Section 2.3 details the data augmentation strategies used. The different alternatives to decoding predictions are reviewed in Section 2.4. Finally, the evaluation metrics are detailed in Section 2.5.

2.1 Overview of the problem

The handwriting recognition problem (HTR) consists of developing models and algorithms able of transcribing the handwritten text into a digital format. The main difficulty of the problem lies in the significant variability of the handwritten text. Each person's handwriting is different from others (interpersonal variability), and even the same person writes the same word in many different ways (intrapersonal variability). Factors such as writing speed, typography size, type of paper and pen used, and even emotional state further increase this variability.

There are two main different lines of research in the HTR problem, depending on the nature of the source data. On the one hand, the problem of recognizing handwritten text on a paper page scanned into a digital image, called *offline handwriting recognition*. On the other hand, the so-called *online handwriting recognition*, which consists of recognizing the handwritten text from the continuous data of the (x, y) positions of the pen obtained while it is written [82]. These data are usually obtained by typing directly on a touchscreen or similar device. This Thesis focuses exclusively on the offline handwriting recognition problem. In Fig. 2.1 we include an example of offline versus online handwriting types of data.

The offline handwriting recognition problem keeps open several research lines that are far from being entirely solved. To recognize the handwritten text in a document is necessary that the algorithms identify which parts of the documents contain handwriting text, optionally segment lines, and words and recognize the words converting the images into



Figure 2.1: Example of offline (left) and online(right) handwriting data.

sequences of characters. The image of the text can appear into tables or forms or overlap with other elements of the page such as: pictures, schemes, or graphs. The text can be written using different languages with different character sets (i.e., scripts), such as Chinese, Arabic, or Japanese.

The first step to recognize the text consist in identifying which parts of the image contains a text. The problem is trivial in some cases; for example, in a well-centered scanned image of a textbook page. However, in other situations is complex, for example, in a commercial street photo with many posters or in the scanned image of a scientific paper that can include text, images, tables, graphs, or equations.

The identification of the different types of elements (e.g., handwritten text, printed text, graphics, etc.) in scanned images of documents is named *layout analysis* [18] (see example in Fig. 2.2). This type of analysis is usual in the automation of document management processes, such as forms processing or automated invoice management. It is also necessary to extract certain elements from the image that needs to be processed separately, such as handwritten signatures.



Figure 2.2: Example of layout analysis over a scientific paper.

The problem of detecting and cropping the text parts in natural scene images is usually analyzed under the general object detection [141] or image segmentation [234] frameworks. An example of scene text detection is provided in Fig. 2.3 This problem is particularly complex because the text in natural images is subject to extensive variability. For example, the use of a wide variety of typographies, the relative orientation and position of the text to the camera, the presence of multi-textured backgrounds or the frequent use of WordArt in posters and signs [219].



Figure 2.3: Example of scene text detection. Source: maps.google.com.

In all of the previous cases, the text detected could be of two types: typographic or handwritten. See differences in Fig. 2.4. As it can be seen, in many real problems, the extraction of the image of a handwritten text on which to apply transcription algorithms is not trivial. Depending on the source, the image of the handwritten text may: have more or less noise, be distorted, include artifacts coming from the elements surrounding the text, or present a partially cropped text. All this adds difficulties to the subsequent recognition steps and opens multiple lines of research regarding this problem [77].

Once the text has been extracted and trimmed, its recognition usually includes several steps [165]. In the most general case, the text image may contain a paragraph composed of multiple lines. In this case, it is most common to segment the text into lines [233]. However, the line segmentation is not trivial due to the slope of the lines, to the slant of characters, and to the fact that some characters in consecutive lines may overlap with others. Text lines can be applied directly to recognition algorithms, or they can be segmented into words, see Fig. 2.5, and recognition can be performed at the word level. Most of the current HTR algorithms can be applied either on lines or on words, which are still a particular case of short lines. In this Thesis, all the experiments performed have been done at the word level.

The training of handwritten text identification models using machine learning techniques requires handwritten text image databases properly annotated with a transcription of the text present in each image. Several such public databases exist; some of them have images of isolated characters, and others contain transcriptions of words, lines, and paragraphs. In any case, the volume of annotated data available to train the models is limited and makes it necessary to consider using data augmentation strategies for model training.

de pondria con ufano que hermosos recultados del nagocio caballer yo por il creado. - bada re observacion semejante à ma ida me parece de instable colness, cali direa de check si Pace to fuede porque infonzi à nozante de lus regocios / Luyos don el nino recien nacido. stopor, ote page Rafael avior de deviero à derlin adjunta una casteta para Broe LOREM IPSUN fectuados recresdos a Lodos repite tuy apris lerve? Y.M.C

Figure 2.4: Examples of handwriting (left) and typographic (right) text.

Servitor allen Ellender of Councard sparked of the apparitions by felling a klevinais andience Nevas a current Warhing lass that Weaver and had Cammun ham 1ha

Figure 2.5: Example of line and word segmentation performed on an image of the IAM database.

In the development of HTR models, the text to be recognized is not a random sequence of characters, but belongs to a given language and is therefore subject to the syntactic and grammatical rules of that language. The step that converts the direct output of an HTR visual model into the sequence of characters that make up the message in a given language is called *decoding*. It can be performed at different levels. A basic level can be used to make sure that the words in the text belong to a given dictionary or lexicon ; this is common in the word recognition problem. Advanced decoding models can also be built to learn the syntactic and grammatical rules of a language and are more common in line or paragraph level recognition models.

Finally, another element that must be defined in detail is the metrics chosen to determine the quality and accuracy of the models obtained, both at character and word recognition levels. The metrics must be selected to understand the model behavior and its recognition capacity, but the metrics used by other authors must also be considered to establish comparisons with the state-of-the-art techniques.

In the following sections, we analyze the considered elements for performing the differ-
ent character and word recognition modeling experiments in detail. The Subsection 2.2 describes some image processing techniques applied to reduce the inherent variability existent in handwriting images. Data augmentation techniques are summarized in Subsection 2.3. In the Subsection 2.4, we analyze the problem of decoding the predictions provided for the visual model to convert it into words. Finally, Subsection 2.5 details the common metrics used for evaluating the experimental results.

2.2 Handwritten text image preprocessing

As explained in the previous section, handwritten text exhibits a very high variability, which not only depends on the writer. Additionally, when the text is digitized, the quality and age of the paper and the digitization process itself can add significant noise to the image. That is why it is usual to pre-process the images of handwritten text [82] [170] before using them in models and recognition algorithms, which allows to eliminate noise and normalize the images to be analyzed as much as possible.

Some ablation ¹ studies of handwritten text recognition models, which include the analysis of the impact of image preprocessing on their results, support the importance of this step for the construction of a competitive recognition system (see for example the work of Dutta et al. in [68]).

To describe in detail the different preprocessing techniques applied to handwritten text, it is important to define the elements that characterize a calligraphy. As shown in Fig. 2.6 a handwritten line or word has three defined zones [163], separated by two horizontal lines called baseline and upperline or header. The central area, called the core-region, is where the characters that do not have ascending or descending strokes are located, such as vowels or some consonants like: 'c', 'm', 'n', 'r', 's', 'v', and 'x'. The height of this zone is called x-height. The upper area, called the ascenders area, includes the upper strokes of characters such as: 'b', 'd', 'f', 'h', and 'l'. The lower area, called the descenders area, contains the lower strokes of characters such as: 'f', 'g', 'j', 'p', and 'q'.



Figure 2.6: Calligraphy zones definition.

Although different authors [82] [23] [126] perform this task with different processes, there are, in most cases, a series of common steps. These are:

- Remove the noise of the image, usually by contrast enhancement techniques.
- Correct of the slope and slant of the text. Fig. 2.7 illustrates with an example the slant and slope definitions.

¹The ablation study aims to measure the performance of an AI model by removing components from it, in order to understand their contribution to the overall performance of the model. The concept was introduced by Newell et al. in [155] p. 43

- Normalize the height of the ascenders and descenders areas.
- Resize and frame the text. For example, to a fixed height or by fitting the text to the edges of the image or by adding fixed size empty borders.



Figure 2.7: Slant and slope in a handwritten word image.

Next we describe each of the above four processings in detail in the context of the HTR problem.

2.2.1 Remove the noise

The digitized image of a handwritten text is subject to multiple sources of noise that make it difficult to recognize, even for a human. The paper may contain marks, may not be completely white, or are degraded. If it is thin, it is also possible to see what is written on its back (i.e., bleed-through effect). The digitization process can introduce artifacts in the image caused, for example, by dirt on the scanner. The final result can be noised images like the examples shown in Fig. 2.8.



Figure 2.8: Example of image with noise take from Osborne database.

Image denoising is the first step in image processing and standardization. The objective is to have a grayscale image in which the text has sharp, well-defined black strokes and the background is white with no elements that could affect the model.

2.2.2 Slope correction

The slope or also called skew is the inclination of the lines of text with respect to a completely horizontal baseline. It usually appears when the text is written on a blank page without a pre-printed pattern of lines or boxes. The deviation of the line can have

a positive angle if the writer tends to raise the line when he/she writes it, or it can also have a negative angle with respect to the horizontal if the line tends to descend. Fig. 2.9 shows an example line slope with a positive angle corresponding to the RIMES database [84]. Slope correction can be performed at the page level, as well as at the line level or even at the word level.

of Radicalism, & the god knows what may happen Horizonta Baseline Horizontal

Figure 2.9: Example of slope angle estimation used in this Thesis.

Likewise, the slope correction is sometimes carried out simultaneously with the slant correction since they are related, because a slope angle correction modifies the slant angle of the words by the same amount (see Fig. 2.7).

2.2.3 Slant correction

The slant (see Fig. 2.7) is present in multiple handwritten texts, especially in the ascending and descending strokes. Usually, the text presents an inclination to the right with the strokes forming an angle of less than 90 degrees with the horizontal, but left-handed writers tend to write with a slant to the left, that is, with an angle greater than 90 degrees with respect to the horizontal.

The slant is a common feature of cursive-type calligraphy used in the learning process of writing. That is why it is very widespread and is one of the main sources of variability in the handwritten text that makes it difficult to recognize text using automatic systems.

The slant identification and correction is a critical aspect in HTR since many recognition algorithms that use handwritten text images as inputs usually have an approach that manages the image column by column. For example, those that use dense neural networktype models such as multilayer perceptron or recurrent neural networks that handle the image as a sequence of columns. Models using convolutional networks do not have this problem.

In any case, the slant correction provides a variability reduction of the handwritten text that facilitates its recognition regardless of the type of model used. And it is, along with denoising, the most common preprocessing technique when tackling the problem of HTR.

Slant analysis is especially relevant when analyzing databases of historical handwritten documents since many calligraphies from past centuries have a very pronounced slant.

The slant correction is usually done in two steps [170]: firstly, identifying the inclination angle of the lines and then applying a transformation to correct that angle.

2.2.4 Normalize ascenders and descenders regions

Fig. 2.6 shows how the writing area is divided into three horizontal regions: the core region that contains most of the strokes, the ascending region that includes the strokes of

the upper characters and the descending region that includes the lower strokes of certain characters. A very important source of variability is precisely the relative size of the ascending and descending regions, depending on the stroke width of each calligraphy.

To standardize the ascending and descending regions sizes it is necessary to identify and delimit the ascenders and descenders zones and scale them in some way.

2.2.5 Crop and resize input images

Handwritten text recognition models based on direct image modeling (i.e., segmentation free), which are practically all used in recent years, usually need that input images to be fixed size, specifically fixed height. For this reason, a series of transformations are applied to a previously standardized line or word images to fit them into this size. These transformations usually include: centering the text in the image, adding borders around the text and resizing the image, keeping the aspect ratio (for which the image is usually completed with columns of white pixels on the right).

The centering of the text in the image is done by removing the empty borders that do not contain any image strokes. Sometimes, a frame of blank rows and columns of fixed size is added around the text image to be centered. The image is then rescaled to a fixed height.

Usually, the selection of the rescaling height, as in the previous reference, reduces the size of the original image by lowering its resolution. This is done because sometimes the scanning resolution of the text image is high, and it is possible to reduce it without affecting the human ability to read the text. It is also done because, in the case of using deep models, the size of the input image to the model has a significant impact on the memory requirements and training times of the model, especially if it is a deep model with multiple layers.

If the model used supports input images of variable width, no additional steps are necessary. If, on the other hand, the model only supports inputs of a fixed size, the last step is to set the width of the image to a fixed value without changing the aspect ratio of it. To perform this stage, maximum image width is selected and all images are adjusted to that width by completing each one of them with empty columns on the right.

2.3 Data augmentation strategies

Data augmentation is a technique in which a training set is expanded with class-preserving transformations.[58]. In the case of handwritten text recognition, it consists of slightly modifying the image of the text to be recognized so that the text remains legible and has the same transcription as the original. For this purpose, some image transformations such as translations, slight rotations and, in general, other affine transformations that meet the above requirement are used. These transformations are usually applied in combination (e.g., translate 5 pixels with a positive rotation of 3° of the image text).

There are two data augmentation strategies: data augmentation during training and data augmentation in the inference or evaluation process called test-time data augmentation. Test-time data augmentation is a practice that allows the improvement of the accuracy

metrics of models by evaluating multiple versions of test data. In general, it can be applied in a standard way on any predictive model.

2.4 Decoding predictions

This Thesis aims to develop models for the recognition of general-purpose handwritten text. This implies that such models must be able to recognize any word that appears written in an image. For this reason, it is not possible to employ strategies in which the system's output is a closed set of words from a vocabulary. Therefore, the systems proposed here are capable of generating any sequence of characters as output.

However, not all character sequences are valid from a grammatical point of view and not all word sequences are valid from a semantic point of view. Therefore, the character sequence resulting from applying a visual recognition model to the text image is processed to be considered a valid word. This is a common step in text recognition systems as described in Subsection 2.1

This decoding can be performed at several levels. The simplest one is at the vocabulary level, for which a lexicon is used as a closed set of words that define the vocabulary where it must be the resulting character sequence of the visual model. If the visual model result is in the lexicon, it is determined that the word is identified in the image. If, on the contrary, it is not, the nearest lexicon word is assigned using a word proximity metric. The most common metric in this case is the Levenshtein distance [138] which is described in detail in the Subsection 2.5.

There are some contexts with a limited vocabulary where this lexicon-based approach is the most appropriate for validating the visual model output. For example, in the case of recognizing form fields with a closed list of valid options such as addresses and postal codes [194], in recognition of telephone numbers [119] or in the processing of bank cheques [63]. In some more general application areas, the main drawback of this method is that it does not consider the word context to refine its recognition. On the other hand, this is an easy-to-reproduce and straightforward method.

When applying this decoding method, the primary consideration is to define which words will be part of the lexicon. In the case of closed output recognition systems, words that are not included in the lexicon, called Out-of-Vocabulary words (OOV), can never be correctly identified. On the other hand, if an extensive lexicon is used when decoding the output character sequence of the visual model, the error rates increases.

The direct search for the nearest word in the vocabulary has a complexity that grows linearly with the vocabulary size. More efficient strategies have been proposed because many words share characters in the same positions. A review of different techniques of this type can be consulted in [124]. Currently, there are very efficient search implementations using Levenshtein distance; for example, the one provided by the open-source software Apache Lucene [16].

In addition to the fact that the word to be decoded belongs to a vocabulary, the context of that word (i.e., the words that are before and after it in the sentence) imposes grammatical and semantic restrictions on the word to be identified, and it can be used to make a more accurate identification of it. This fact introduces a second level at which decoding can be carried out. The most common way of introducing this context into the decoding

process, in the area of handwriting recognition, is the statistical *n*-gram language models [14], [126].

An *n*-gram language model is a probabilistic model that predicts the next word in a sequence as a function of the previous *n*-1 words: $P(x_i \mid x_{i-(n-1)}, \ldots, x_{i-1})$, regardless of the position of the word in the sequence. The calculation of the maximum plausible estimator is made in frequency terms as indicated in the equation 2.1, by taking into account that the probability of a word conditioned to the previous word sequence follows a categorical distribution also known as *multinomial distribution*.

$$p(x_i|x_{i-1},...,x_{i-n+1}) = \frac{Count(x_i,x_{i-1},...,x_{x-i+1})}{\sum_x Count(x,x_{i-1},...,x_{i-n+1})}$$
(2.1)

These models are generally constructed from large corpora that allow sufficient repetition of the most common word sequences in a language. However, as n increases, the probability of an n-gram not appearing in the corpus also grows. It results in many sequences having an estimated probability of zero. To solve this problem, the most common way is to add a small smoothing value δ to all frequency counts as indicated in the equation 2.2.

$$p(x_i|x_{i-1},...,x_{i-n+1}) = \frac{Count(x_i,x_{i-1},...,x_{x-i+1}) + \delta}{\sum_x (Count(x,x_{i-1},...,x_{i-n+1}) + \delta)}$$
(2.2)

A comparative review of different ways of applying smoothing to n-gram based language models can be found in [39].

2.5 Evaluation metrics

In order to evaluate the quality of the models built, metrics must be defined to assess the usefulness and applicability of the models and their limitations. Likewise, the usual metrics in the literature should be provided so the results obtained can be compared precisely with those from other authors.

In the case of character recognition models, the most common metric used is *accuracy* (i.e., the percentage of cases correctly identified). However, for word or line recognition, this metric cannot distinguish between an error produced by a single character from another error in which all the characters of the word are incorrect. To solve this inconvenience, the *Levenshtein edit distance* metric [138] is used. This metric counts the minimum number of operations required to transform a character sequence into another one by considering substitutions, deletions, and insertions of characters.

2.5.1 Metrics to evaluate character recognition models

Although the metric almost exclusively used in the evaluation of this type of multicategory classifier is accuracy or its complement, the *error rate* (percentage of cases incorrectly identified), these metrics have their limitations and drawbacks. Being N the total number of cases and c_{ij} the number of cases of true category *i* assigned to the predicted category *j*, the accuracy is defined according to the equation 2.3.

$$accuracy = \frac{1}{N} \sum_{i=j} c_{ij} \tag{2.3}$$

Accuracy is a metric that should always be provided since it is the one that allows comparing the results with other authors. However, accuracy alone does not provide details about the limitations and types of errors that each model has, which provide clues for its optimization and improvement. Additionally, accuracy is influenced by the number of different target values and the distribution of these values' frequencies. For example, the accuracy of a random model based on the target distribution for a binary target with an unbalanced distribution of 9 to 1 for the most frequent category would be 90%, and for a target of 10 balanced categories, it would be 10%.

In unbalanced samples, the less frequent categories may obtain a low allocation rate by the model even if high overall accuracy is obtained. To monitor this, *recall* metric is used to measure how many of the actual cases in each category are correctly predicted by the model. This metric is calculated individually for each of the target categories. To follow the usual notation used in the case of binary targets, given a reference category i, we will adopt the following notation:

- True Positive (TP) represent the cases of the category i correctly classified
- False Positive (FP) represent cases from other categories assigned to category i
- False Negative (FN) represent cases in category i assigned to other categories
- True Negative (TN) represent cases from other categories assigned to other than the i

Following this, the *recall* metric is defined according to the equation 2.4

$$recall = \frac{TP}{TP + FN} \tag{2.4}$$

In the case of unbalanced samples also the most frequent category or categories are usually assigned cases of other minority categories that could be similar. To analyze this effect, the category *precision* metric is calculated. It estimates the percentage of cases correctly assigned by the model to that category (see the equation 2.5);

$$precision = \frac{TP}{TP + FP} \tag{2.5}$$

Accuracy and recall metrics provide good information about the model behavior in each category. To obtain a single unified measure of the model goodness in each category, the metric called $F1 \ score[168]$ is usually used, which is mathematically defined as the harmonic mean of precision and recall:

$$F1score = 2 * \frac{precision * recall}{precision + recall}$$
(2.6)

The metrics above quantify the volume of model error but do not provide information on what specific allocation errors the model produces (i.e., what categories the model is confusing). The confusion matrix is used to evaluate the marginal distributions of the error cases in each category. This square matrix, of size equal to the number of categories in the target, is constructed by including in each cell (i, j) the number of cases in category *i* that the model has assigned to category *j*, c_{ij} . Therefore, the main diagonal of this matrix includes the number of cases correctly assigned to each target category, and the cells outside this diagonal include all the cases erroneously assigned to different categories.

With the different metrics previously indicated, it is possible to analyze the results of a classification model in detail. However, it must be remembered that all of them are influenced by the target distribution and that the assignment is made to the category for which the model provides a higher probability conditioned on the input variables. If one wants to obtain a metric that measures the quality of the predictions in relative terms and not from their absolute values, independent of the target distribution and the classification thresholds of each category, one must consider the *Area Under the ROC curve* (AUC).

The *Receiver Operations Characteristics* (ROC) curve [32], is a line graph showing the performance of a classification model for each of the possible classification thresholds of a given category. The graph is calculated for each category of the target and the curve represents the relationship between two parameters:

- The so-called *true positive rate* (TPR), calculated as the number of cases in the category correctly classified (TP) divided by the total number of cases in the category, which includes those correctly classified plus those incorrectly classified in other categories (TP + FN). This is the recall of the category.
- The so-called *false positive rate* (FPR), calculated as the number of cases in the other categories misclassified in the category (FP) divided by the total number of cases in the other categories, which includes those misclassified in the category plus those classified in the other categories (FP + TN)

$$TPR = \frac{TP}{TP + FN} \tag{2.7}$$

$$FPR = \frac{FP}{FP + TN} \tag{2.8}$$

The ROC curve represents TPR versus FPR for each of the different possible classification thresholds for the analyzed category. It is starting on the threshold zero, which does not assign any case to the category and therefore provides values of 0 to FPR and TPR, and ends in the threshold one that assigns all cases to the category and provides a value of 1 to both TPR and FPR. An example of such a curve is shown in Fig. 2.10.

The ROC curve provides detailed information on the classification behavior in a category as a function of the classification thresholds. This information is usually summarized in the area metric under the ROC curve (AUC) [32] which measures all the two-dimensional area under the curve between the points (0.0) and (1.1) as shown in Fig. 2.10

A perfect model, without errors, would provide a TPR of 1 and an FPR of 0 and therefore an AUC=1. A purely randomized model that assigns cases to the category at random



Figure 2.10: Example of ROC curve and AUC area.

following the proportion of that category in the target, assigns values of TPR = FPR for each possible classification threshold, producing a ROC curve equal to the bisector of the first quadrant and an AUC of 0.5 that is used as baseline reference of a randomized model.

2.5.2 Metrics to evaluate word recognition models

The two main metrics commonly used to evaluate word and line-level handwriting recognition models are the *Character Error Rate (CER)* and the *Word Error Rate (WER)* [72]. The CER measures the Levenshtein distance [138] between the predicted and the real character sequence of the word. The Levenshtein distance, also sometimes called *edit distance*, is a metric to measure the difference between two sequences. Informally, the Levenshtein distance between two words (the model's prediction and the real one) is the minimum number of insertions, deletions, or substitutions needed to transform the prediction into the real word divided by the length of the real word, as indicated in the equation 2.9. In this Thesis, the CER results provided have been calculated with the Python library editdistance that implements the algorithm efficiently by means of dynamic programming as indicated in [101].

$$CER(prediction, real) = \frac{substitutions + insertions + deletes}{len(real)}$$
(2.9)

The *word error rate* is defined in a similar way to CER by calculating the minimum number of insertions, substitutions and deletions of words needed to go from a text string predicted by the model to the real text string. In the case that the recognition is done at the level of individual words, the WER represents the percentage of words correctly identified by the model and therefore matches the accuracy of the model.

The two previous metrics can be directly calculated from the estimation of the visual model or on the result of applying a subsequent decoding process by searching lexicons or language models. This Thesis provides evaluation results of the proposed models using these two previous metrics, both applied to the direct estimation of the visual model and then using a decoding process by searching in standard lexicons for reference datasets.

It is important to highlight that the above metrics are strongly influenced by the set of characters you decide to select to encode the texts of the reference corpora. That is, the different corpora used as benchmarks in handwriting provide an exact transcription of the text images included in them, differentiating capitalization, lower case and punctuation. However, some authors, when training and evaluating their models, eliminate punctuation marks or do not distinguish between upper and lower case letters. Without discussing which approach may be the most appropriate, it is clear that these are different problems not comparable to each other in terms of the above metrics. This is one of the first sources of problems when it comes to comparing results from different lines of research in the area of handwriting, since simplifying the coding masks all errors in recognition of punctuation marks or apostrophes and all the confusion that may exist between upper and lower case forms of the same letter.

In order to determine that there are statistically significant differences between different methods, the comparison must consider not only the estimated value of the error metric obtained from an experiment, but also a measure of the estimation error of that metric. Confidence intervals are usually used to determine whether differences between two values of a metric can be considered statistically significant or not. The most common way to construct these confidence intervals is to assume that the metric under consideration follows a known statistical distribution. For a rate-of-error type metric, the usual assumption is that it follows a binomial or normal distribution. An alternative, if this assumption about the distribution of error cannot be guaranteed, is to use a non-parametric estimation method such as non-parametric boostrapping [3]. In the case of this Thesis, the confidence intervals shown have been calculated using this non-parametric technique.

In the case of handwritten text recognition, few authors provide confidence intervals for the error measures of the proposed models (for example, [170] or [19]).

Finally, one must also consider that handwritten text recognition models are becoming more computationally complex since they are built with increasingly complex deep learning architectures. In addition to the cost of the visual recognition model, we must add the computational cost of the decoding process, which includes context information in the results. If this is done by searching in lexicons, the cost increases significantly with the size of the lexicon as indicated in the Subsection 2.4, and if a language model does it, these models, especially those based on neural networks, also have a very high computational cost. Reporting these computational time and cost metrics is also important to evaluate the feasibility of the practical application of the proposed algorithms when using deep learning architectures. It is also relevant to provide these efficiency metrics if CPU-based or GPU-based hardware is used because the computation time and cost of deep architectures in these two types of hardware platforms are very different. Some authors like [142] provide these metrics, although this is not common in most publications.

Chapter 3

Handwriting Databases

In this chapter, we describe the databases used for the experiments carried out in this Thesis. First, the handwriting character databases in Section 3.1 and, then, the general databases at word and line level in the Section 3.2. Finally, Section 3.3 present a summary of the analyzed databases.

3.1 Handwriting characters databases

In this section we provide a brief description of the research handwriting isolated character databases used in this Thesis. For a complete list of databases available and a more detailed description, Hussain et al. [100] provides a comprehensive survey about handwritten databases.

The following four databases have been selected for the isolated character recognition experiments carried out in this Thesis:

- The NIST database is the largest database of isolated characters but the images are binary.
- The MNIST digits database is the most widely used database but only contains digits.
- The TICH database provides grayscale images of letters and digits.
- The new COUT database includes special characters and punctuation marks.

In the following sections, we describe these databases, as well as example images and the distributions of the available data by partition and available characters.

3.1.1 NIST database

We provide some details about the second edition of the NIST Special Database 19, Hand printed Forms and Characters [85]. This database provides isolated characters segmented from 3,669 handwriting sample forms written each one by a different person.

The isolated character database includes a total of 814,225 images of letters and digits characters corresponding to '0'- '9', 'A'- 'Z' and 'a'- 'z'. Al the images are saved in PNG binary format with a resolution of 128×128 pixels per image. The characters are well centered on the image with large margins. We show some examples of the character images in Fig. 3.1.

The database has eight partitions: the partitions 0, 1, 2, 3, 6, 7, and 8 were obtained from Census Bureau employees in Maryland and partition 4, proposed as test partition, was completed by the Bethesda high school students.



Figure 3.1: Sample images of the NIST database.

The distribution of the available characters in the train partition is biased with high digit frequency and low word frequency. This is because the template form used to get the data includes 28 boxes for numbers but only 2 for alphabet characters and 1 for a 53 words paragraph. For digits, the distribution is similar in the train and test partitions. In Table 3.1 we provide the frequencies of each digit in the partitions.

Char	#Train	%Train	#Test	%Test
0	$34,\!803$	10.11	5,560	9.48
1	38,049	11.05	$6,\!655$	11.35
2	$34,\!184$	9.93	$5,\!888$	10.04
3	$35,\!293$	10.25	$5,\!819$	9.92
4	$33,\!432$	9.71	5,722	9.76
5	$31,\!067$	9.02	$5,\!539$	9.44
6	34,079	9.90	$5,\!858$	9.99
7	35,796	10.40	$6,\!097$	10.40
8	$33,\!884$	9.84	$5,\!695$	9.71
9	33,720	9.79	$5,\!813$	9.91

Table 3.1: Digits frequency in the NIST train and test partitions.

For letters, the distribution is clearly different in the test partition, where the available examples are evenly distributed. We provide the frequencies of each letter in the train and test partitions of the database in Table 3.2.

Char	#Train	%Train	# Test	%Test	Char	#Train	%Train	#Test	%Test
А	7,010	1.81	459	1.92	a	$11,\!196$	2.89	481	2.01
В	4,091	1.06	435	1.82	b	5,551	1.43	461	1.93
\mathbf{C}	11,315	2.92	518	2.16	c	2,792	0.72	494	2.06
D	4,945	1.28	396	1.65	d	$11,\!421$	2.95	439	1.83
Ε	$5,\!420$	1.40	365	1.52	e	28,299	7.31	424	1.77
F	10,203	2.63	419	1.75	f	$2,\!493$	0.64	468	1.95
G	2,575	0.66	389	1.62	g	3,839	0.99	437	1.83
Η	$3,\!271$	0.84	402	1.68	h	9,713	2.51	504	2.11
Ι	$13,\!179$	3.40	815	3.40	i	2,788	0.72	364	1.52
J	3,962	1.02	426	1.78	j	1,920	0.50	293	1.22
Κ	$2,\!473$	0.64	377	1.57	k	2,562	0.66	395	1.65
\mathbf{L}	$5,\!390$	1.39	496	2.07	1	$16,\!937$	4.37	916	3.83
Μ	10,027	2.59	460	1.92	m	$2,\!634$	0.68	475	1.98
Ν	$9,\!149$	2.36	439	1.83	n	$12,\!856$	3.32	460	1.92
Ο	$28,\!680$	7.40	459	1.92	0	2,761	0.71	454	1.90
Р	9,277	2.39	467	1.95	p	2,401	0.62	415	1.73
\mathbf{Q}	2,566	0.66	452	1.89	q	$3,\!115$	0.80	384	1.60
\mathbf{R}	$5,\!436$	1.40	446	1.86	r	$15,\!934$	4.11	491	2.05
\mathbf{S}	$23,\!827$	6.15	445	1.86	s	$2,\!698$	0.70	438	1.83
Т	10,927	2.82	469	1.96	t	20,793	5.37	434	1.81
U	$14,\!146$	3.65	458	1.91	u	$2,\!837$	0.73	475	1.98
V	4,951	1.28	482	2.01	v	$2,\!854$	0.74	524	2.19
W	5,026	1.30	475	1.98	w	$2,\!699$	0.70	465	1.94
Х	2,731	0.71	472	1.97	x	2,820	0.73	472	1.97
Υ	5,088	1.31	453	1.89	у	$2,\!359$	0.61	387	1.62
Ζ	$2,\!698$	0.70	467	1.95	z	2,726	0.70	450	1.88

Table 3.2: Letters frequency in the NIST train and test partitions.

3.1.2 MNIST database

The MNIST¹ database [135] is a subset of the previously discussed NIST database. It includes only images of digits characters, where the digits have been size-normalized and centered in a fixed-size image. This dataset is extensively used in the literature to compare deep learning models.

The MNIST character database contains a balanced sample of 70,000 digits characters from '0' to '9' selected from the previous NIST database. The database is available in several ways. It is included in different Python packages as scikit-learn, tensorflow or pytorch. Each digit image has a resolution of 28×28 pixels in grayscale format of 128 bits.

To build it, the original binary images from the NIST database were cropped and resized to 20×20 pixels while preserving the aspect ratio. The grey levels appear as a result of the anti-aliasing technique used. Finally, the images were centered over a 28×28 image, including margins of 4 pixels.

The database is distributed with a standard train test partition of 60,000 images to train and 10,000 to test. In each partition, the distribution of characters is balanced, including 6,000 samples of each digit in the training partition and 1,000 samples of each digit in the test partition. This partition is different from the NIST original one, but it is guaranteed that the partitions contain different writers who produced the digits.

¹http://yann.lecun.com/exdb/mnist/



Figure 3.2: Sample images of the MNIST database.

3.1.3 TICH database

The Tilburg Characters data set (TICH) contains labeled characters, cut out from the Firemaker collection [185] using a semi-automatic approach and curated them by a human labeler. The result includes a total of 40,141 images of isolated characters. The database includes only uppercase characters and digits and is described extensively by van der Maaten in [143].

The database is available to download in https://lvdmaaten.github.io/publications/misc/characters.zip² in Matlab format.

The distributed database includes the set of original images cropped from the Firemaker collection scanned in grayscale at 300 dpi with 8 bits per pixel. In addition, it provides two versions of normalized images. In the first one, the original images are normalized by resizing them to fit into a box of 50×50 pixels keeping the aspect ratio and then 3 pixels width are added to pad the borders resulting in a 56×56 pixels images. The second normalization centered and normalized the characters in a box of 90×90 pixels without borders. In this Thesis all the experiments and results were developed with this second normalization. Some samples of images of the TICH database are shown in Fig. 3.3

This database also includes a list of character labels, a list of writer labels, a default partition in train and test subsets, and a baseline benchmark for the isolated character recognition problem.

The distribution of available characters is biased with the high frequency of English most common characters and they are detailed in Table 3.3 for the train partition.

3.1.4 New COUT database

The new COUT (Characters Offline from Unipen Trajectories) database contains images of isolated handwriting characters and is built using the UNIPEN online handwriting database [90].

 $^{^{2}}$ last access May 2021



Figure 3.3: Sample images of the TICH database.

Char	Ν	Freq.	Char	Ν	Freq.	Char	Ν	Freq.
0	375	1.21%	C	432	1.39%	0	2,177	7.02%
1	171	0.55%	D	1,704	5.49%	Р	317	1.02%
2	177	0.57%	E	$2,\!428$	7.83%	Q	171	0.55%
3	185	0.60%	F	304	0.98%	R	$2,\!279$	7.35%
4	168	0.54%	G	454	1.46%	S	$1,\!286$	4.15%
5	124	0.40%	Н	591	1.91%	Т	1,267	4.08%
6	156	0.50%	I	668	2.15%	U	2,049	6.61%
7	190	0.61%	J	104	0.34%	V	1,525	4.92%
8	135	0.44%	K	696	2.24%	W	785	2.53%
9	152	0.49%	L	903	2.91%	X	0	0.00%
А	2,707	8.73%	M	1,163	3.75%	Y	364	1.17%
В	610	1.97%	N	$3,\!469$	11.18%	Z	733	2.36%

Table 3.3: Character frequency in the train partition of TICH database.

To generate the COUT database, the point trajectories obtained from the original UNIPEN online handwriting database were used. Variable-thickness strokes were used depending on the original resolution of characters to ensure that all final characters have a similar thickness. The images from different available categories (i.e., different types of alphabet characters) were generated: uppercase, lowercase, digits and punctuation marks. These generated images are resized to 64×64 pixels without changing their aspect ratio. Finally, the generated images were curated manually, one by one, to ensure that they are assigned to their correct category and that they are human-legible. Some samples of the database images can be viewed in Fig. 3.4.

This database contains 93 categories and a total of 62,382 image characters in grayscale organized in 93 folders, one for each different character. The folder name is assigned using the ASCII numeric code of the character. The included characters are all between the ASCII code 33 (character: !) and the ASCII code 126 (character: ~), both included. As for as we know, it is the handwriting character database that includes the largest number of different characters. The frequency of each character is detailed in Table 3.4.

The database is free available to download for research purposes from: https://github.com/sueiras/handwritting_characters_database



Figure 3.4: Sample images of the COUT database.

Char	Ν	Freq.	Char	Ν	Freq.	Char	Ν	Freq.	Char	Ν	Freq.
А	923	1.49	a	2,219	3.58	0	$1,\!103$	1.78	;	223	0.36
В	505	0.81	b	624	1.01	1	1,036	1.67	<	168	0.27
С	644	1.04	с	880	1.42	2	1,056	1.70	=	254	0.41
D	553	0.89	d	1,074	1.73	3	1,047	1.69	>	162	0.26
Ε	1180	1.90	е	2,962	4.78	4	1,028	1.66	?	194	0.31
\mathbf{F}	486	0.78	f	608	0.98	5	957	1.54	0	83	0.13
G	453	0.73	g	760	1.23	6	943	1.52	[111	0.18
Η	557	0.90	h	990	1.60	7	1,031	1.66	j	104	0.17
Ι	898	1.45	i	2,035	3.28	8	1,055	1.70	^	88	0.14
J	476	0.77	j	427	0.69	9	1,049	1.69	_	80	0.13
Κ	508	0.82	k	557	0.90	!	207	0.33		42	0.07
L	703	1.13	1	$1,\!415$	2.28	"	267	0.43	{	73	0.12
Μ	496	0.80	m	879	1.42	#	152	0.25		91	0.15
Ν	689	1.11	n	$1,\!906$	3.08	\$	192	0.31	}	77	0.12
0	$1,\!126$	1.82	0	2,048	3.30	%	190	0.31	~	59	0.10
Р	614	0.99	р	786	1.27	&	104	0.17			
\mathbf{Q}	413	0.67	q	427	0.69	,	276	0.45			
R	816	1.32	r	1,708	2.76	(346	0.56			
\mathbf{S}	799	1.29	s	$1,\!557$	2.51)	359	0.58			
Т	738	1.19	t	1,781	2.87	*	128	0.21			
U	461	0.74	u	1,319	2.13	+	146	0.24			
V	407	0.66	v	555	0.90	,	320	0.52			
W	528	0.85	w	680	1.10	-	447	0.72			
Х	368	0.59	x	463	0.75		486	0.78			
Υ	457	0.74	У	680	1.10	/	259	0.42			
Z	459	0.74	z	505	0.81	:	287	0.46			

Table 3.4: Character frequency in train partition of COUT database.

3.2 General handwritten text databases

This subsection describes in detail the databases used at word level. These datasets do not include any information that would allow these words to be segmented into their component characters. The databases considered provide segmented images of words as well as their transcriptions. In some cases, these databases have images and transcriptions at line or page level. In general, the datasets considered include text written by multiple authors with a high variability regarding writing styles and they cover three of the main languages with Latin script characters. In particular:

• IAM database corresponds to English text,

- RIMES database corresponds to French text, and
- The Osborne database is written in Spanish

3.2.1 IAM Database

IAM database was introduced by Marti et al. [146] and it consists of handwritten English sentences extracted from the Lancaster-Oslo/Bergen (LOB) corpus [110]. This database is widely used in the scientific literature; see for example Doetsch et al. [64], Bluche et al. [21], or Graves et al. [79]. This is due to its large diversity of styles contained and its size (as it includes more than one hundred thousand annotated words). IAM database also does not impose any restrictions on the annotated character set, including not only capital and lowercase but also the most common punctuation marks. This database provided images of forms, lines and words, with a resolution of 300 DPI that were saved as PNG with 256 gray levels. Fig. 3.5 show an example of one IAM form.

Sentence Database	A01-000
A MOVE to stop Mr. Gaitskell from nominating any more Labour be made at a meeting of Labour M Ps tomorrow. Mr. Michael Foot a resolution on the subject and he is to be backed by Mr. Will Grif Manchester Exchange.	life Peers is to has put down ffiths, M P for
MOVE to stop Mr. Gaitskell	from
nominating any more Labour iil	Fe Peors
is to be made at a meeting	s af Labour
MPs domorrow. Mr. Michael	Foot has
put down a resolution on the	subject
and he is to be backed by	hr. Win
siffiths, MP for Manchede	Exchange

Figure 3.5: Sample image of a page of the IAM database.

The complete database can be downloaded from https://fki.tic.heia-fr.ch/databases/ iam-handwriting-database ³ previous registration. A total of 657 writers contributed with samples of their handwriting and the database includes a total of 1,539 text pages, 13,353 isolated and labeled text lines and 115,320 isolated and labeled words. In this Thesis we use the isolated words images and their corresponding annotations. Fig. 3.6 includes some examples of several isolated words.

 $^{^{3}}$ Last access May 2021



Figure 3.6: Sample images of the IAM database.

The authors also provided a standard partition per writer in a training set, two validation sets (val1 and val2) and one test set. In the problem of handwritten text recognition, both at word and line level, the train, val2 and test partitions are used, and the val1 partition is ignored. This selection of partitions contains a total of 8,962 lines from 454 different authors distributed as shown in Table 3.5

Partition	#Authors	#Pages	#Lines	#Words
train	283 43	747 115	6,161	53,841 8 566
test	43 128	232	1,861	17,616

Table 3.5: Distributions of authors, pages, lines and words in the IAM official partition.

This official partition was used by many different authors (see references [23], [169], [68], [46], or [99] for the development of models both at line and at word levels).

Additionally, there exist a second unofficial partition, usually named Aachen or RWTH split, widely used by most authors (see references [126], [26], [64], [216] or, [201]) who used the database for handwritten text recognition. To our best knowledge, this partition was first used in [126]. This partition included a total of 747 pages to train, 116 for validation and 336 for test. The lines and words are distributed as shown in Table 3.6, The partition is available in http://www.openslr.org/56/⁴.

Partition	#Pages	#Lines	#Words
train	$747 \\ 116 \\ 336$	6,482	55,081
validation		976	8.895
test		2,915	25,920

Table 3.6: Distributions of pages, lines and words in the IAM Aachen partition.

The database includes a metadata file with the details of each isolated word segmented, including the necessary label transcription of each word. The file also includes a column that indicates if the word was correctly segmented or not. These segmentation errors are marked at the line level, even though only one word in that line may have segmentation errors. Since it is not possible to identify in the data which words of each line are incorrect, a valid option is to remove the wrong marked lines from the different partitions. This is because the IAM database was originally annotated only at line level. In a later work of

⁴Last access May 2021

the same authors [241], a bounding box and a word level transcript were provided, based on a segmentation algorithm of the identification of related components, when it was later validated manually at line level to identify erroneous cases. Altogether, there was 2,009 error marked lines out of a total of 13,353 available lines. In addition, it has been identified that the words annotated with the character '#', although they are identified as correct segmentation, correspond to crossed-out words that must also be discarded (there are a total of 33 words in this situation). Some examples of images of these words are included in Fig. 3.7. In summary, after filtering the words that are marked as correctly segmented and with different annotations for the character '#', were obtained a final 47,952, 7,558 and 20,306 words for the training, validation and test sets, respectively, over the Aachen partition. These are the final datasets that we have originally used in the publication [201] which is discussed in detail in Chapter 6.



Figure 3.7: Sample error images of the IAM database.

The set of characters that appears in IAM database texts are composed of a total of 77 different symbols, including all the English lowercases and capital characters, and also the punctuation marks included in Table 3.7. In addition, it also includes the symbol '#' to represent garbage (basically crossed-out words) and also the 33 occurrences of the blank character (the ASCII code 32), mainly used to separate capital letters from acronyms and abbreviations.

Char	Ν	Freq.	Char	Ν	Freq.	Char	Ν	Freq.f
•	5,005	1.244%	,	4,418	1.098%	,	$1,\!379$	0.343%
-	$1,\!184$	0.294%	"	1,080	0.268%	!	185	0.046%
?	162	0.040%	;	159	0.040%	:	137	0.034%
(103	0.025%)	96	0.024%	&	47	0.012%
/	10	0.002%	*	5	0.001%	+	5	0.001%

Table 3.7: Frequency of punctuation symbols that appears as a segmented words in the IAM database.

3.2.2 RIMES database

RIMES (Reconnaissance et Indexation de données Manuscrites et de fac similÉS / Recognition and Indexing of handwritten documents and faxes) database [84] consists of images of handwritten letters in exchange of gift vouchers. The database is maintained by the company A2IA and the complete version contains 12,723 pages corresponding to three types: handwritten letters, forms and fax covers. This database was designed to cover several tasks such as layout analysis, handwriting recognition, writer identification, logo identification and information extraction. The database is available at http://www.a2ialab.com/doku.php?id=rimes_database:start⁵.

RIMES database has been used for several handwriting competitions (such as ICFHR 2008, ICDAR 2009, ICDAR 2011) with different tasks and datasets. In this Thesis we have used the ICDAR 2011 competition on word recognition. This version of the database includes 51,739 words images to train, 7,464 for validation and 7,776 to test, respectively. Fig. 3.8 includes some examples of several isolated words at RIMES.



Figure 3.8: Sample images of the RIMES database.

The set of characters in words of the RIMES database include all the lowercase and capital characters contained in standard ASCII, the ten digits and some punctuation marks and specific French characters detailed in Table 3.8

Char	Ν	Char	Ν	Char	Ν	Char	Ν
é	5,155	,	2,646	è	762	-	360
à	178	ê	158	ç	94	ô	69
î	67	/	58	û	52	â	23
ù	15	0	13	ë	3	ï	2
%	1	2	1	É	1		

Table 3.8: Frequency of accented vowels and punctuation symbols in the train partition of the RIMES database.

3.2.3 Osborne database

Founded in 1772, Bodegas Osborne is one of the world's oldest companies in active. Along their history they provide wines to several illustrious clients in Europe like the British Royal Family or the Vatican. This has caused that the company had have to exchange mail with relevant historical figures from the 18th century onwards. The Osborne historical archive counts with several hundreds of documents including personal and commercial letters or counting sheets. Fig. 3.9 contains two documents examples of this database. All the available document images and transcriptions can be accessed at https://www.fundacionosborne.org/es/simple-search⁶.

The dataset used in this Thesis includes a subset of images corresponding to isolated words extracted from a total of 138 pages from 58 documents digitized from this archive. A total of 6 documents are from the year 1877, other 7 ones are from 1882 and 45 were dated in the

⁵Last access May 2021

⁶Last access May 2021

Part I'mary . her 2- 182 Sevilla 4 Mayo 1883 My dear his El neg I retain you by the bearer 1.1.1.2.3. Hist new Rega . Grand Cabeza de Vaca the . Simul as las Inon ralat r Vide de Cerenshi Velles marches los mi true Vinducen de Calders Ar. Caro que despu a solor il hate ant: . lipon 1 the Paris here Robber here has du ke articles which totad as In an henselsuping iv no estare' en esa by see tall in the cause 7 a obrado para la facin - Juido and when when s s.

Figure 3.9: Sample images of pages of the Osborne database.

year 1883. These documents were scanned using RGB 400 DPI resolution. The annotation process was developed by a specialized documentation team and the isolated word images are obtained by applying an automatic process that includes line segmentation and word segmentation. The result is a human curated dataset to ensure that the cropping and translation are correct. The final images were converted to grayscale with 8-bits pixels depth and saved in PNG format. Fig. 3.10 includes some image examples from Osborne database.



Figure 3.10: Sample images of the Osborne database.

The dataset includes a default partition in train, validation and test which was used in this Thesis. The partition has been built randomly by selecting all pages of complete scanned documents and all the words of each page in one partition. The train partition has a total of 7,149 word images extracted from 129 pages of 54 different documents. The validation partition has 288 words images extracted from 5 pages of 2 documents, and the test partition covers 194 word images extracted from 4 pages of 2 different documents.

The target annotations are included in the name of each image file. In this name it is registered: the year of the document, the document identifier, the page of the document, the word position in the page and the transcription of the word encoded in hexadecimal format. The number of different words in the database is 2,893 and the most frequent word is 'de' with a total of 358 occurrences.

Because the segmentation into words has been done by an autonomous process, in some cases, the segmented image corresponds to a part of a word or to two or more words whose strokes are close and have not been identified as different words. In these cases, the blank character ' ' may be part of the transcription of the image text. It has been manually verified that all the transcriptions are correct.

char	Ν	Freq.	char	Ν	Freq.	char	Ν	Freq.
	468	1.42	F	9	0.03	j	112	0.34
!	3	0.01	G	6	0.02	k	14	0.04
&	6	0.02	Н	26	0.08	1	1,549	4.69
,	52	0.16	Ι	15	0.05	m	898	2.72
(16	0.05	J	24	0.07	n	2,002	6.06
)	13	0.04	L	52	0.16	0	2,919	8.83
,	264	0.80	Μ	78	0.24	р	752	2.28
-	284	0.86	Ν	25	0.08	q	445	1.35
	111	0.34	0	20	0.06	r	2,022	6.12
0	19	0.06	Р	44	0.13	s	2,216	6.71
1	58	0.18	Q	10	0.03	t	1,318	3.99
2	49	0.15	R	18	0.05	u	$1,\!240$	3.75
3	36	0.11	S	98	0.30	v	364	1.10
4	18	0.05	Т	43	0.13	w	22	0.07
5	7	0.02	U	77	0.23	x	20	0.06
6	7	0.02	V	23	0.07	У	373	1.13
7	14	0.04	W	1	0.00	Z	80	0.24
8	67	0.20	Y	17	0.05	i	7	0.02
9	7	0.02	a	3843	11.63	i	2	0.01
:	5	0.02	b	392	1.19	á	55	0.17
;	11	0.03	c	1163	3.52	é	50	0.15
?	7	0.02	d	1588	4.81	í	13	0.04
А	83	0.25	e	4319	13.07	ñ	54	0.16
В	32	0.10	f	200	0.61	ó	17	0.05
\mathbf{C}	76	0.23	g	387	1.17	ú	7	0.02
D	89	0.27	h	340	1.03			
Ε	87	0.26	i	1785	5.40			

Table 3.9 details the frequency of characters that appears in the database.

Table 3.9: Character frequency in the Osborne database.

3.3 Databases summary

This subsection includes two summary tables of the databases studied. For the character databases we include the character set and the cardinality of the target, The number of images, the number of writers (if it is knows) and the resolution of the images in the Table 3.10.

For the continuous text databases, data on the number of different writers, number of pages, number of lines, and number of words are included for each partition in the Table 3.11. The RIMES data are calculated from the RIMES words database of the ICDAR

Database	Char set (N)	Writers	Resolution	Size
MNIST	Digits(10)		28×28	70,000
TICH	Digits-upper(36)	250	90×90	40,141
NIST	Digits-upp-low(52)	$3,\!669$	128×128	$814,\!225$
COUT	Digits-upp-low-symbols(92)		64×64	$62,\!382$

Table 3.10: Summary for the isolated character databases.

2011 competition 7

Database	Partition	#writers	#pages	#lines	#words
	Train		747	6,482	55,081
IAM (Aachen)	Validation		116	976	8.895
,	Test		336	$2,\!915$	$25,\!920$
	Train	283	747	6,161	53,841
IAM (Official)	Validation	43	115	940	8.566
	Test	128	232	$1,\!861$	$17,\!616$
	Train		1,344	8,807	51,739
RIMES (2011)	Validation		196	1,268	7,464
	Test		179	1.250	7,776
	Train		112		6.340
Osborne	Validation		13		646
	Test		13		645

Table 3.11: Summary for the continuous handwriting text databases.

⁷http://www.a2ialab.com/doku.php?id=rimes_database:data:icdar2011:word: icdar2011competitionword

Chapter 4

Review of Handwriting recognition approaches

This chapter reviews the different approaches to solving the offline handwritten text recognition problem over time. In the last twenty years under review, the problem has been actively investigated, and several approaches to the problem have emerged, which have prevailed at different stages and have generally replaced the approaches of the previous stages. For a review of previous years, see Plamondon and Srihari's survey paper [165].

In the first Section 4.1, the different stages that have been identified over time are summarized. The next Section 4.2 reviews the various input image preprocessing and normalization algorithms used. Section 4.3 describes data augmentation techniques considered. This is followed by details of the HMM-based algorithms that have been predominant up to 2013 (Section 4.4). Subsequently, the solutions using neural networks, predominant from 2013 onwards, are analyzed in Section 4.5 and the different types of networks used. The next Section 4.6 reviews the use of Connectionist Temporal Classification (CTC) in the offline handwriting recognition problem, usually employed with previous architectures based on neural networks. This is followed by a review of k-NN-based solutions in Section 4.7. The next Section 4.8 details the approaches employing seq2seq architectures. Section 4.9 reviews the works that have employed synthetic data and transfer domain strategies which are mostly concentrated in the last three years. Finally, in Section 4.10 we explain the historical applications of language models in the HTR problem.

4.1 Introduction

The problem of automatic Handwritten Text Recognition (HTR) persists since document digitization started. Text recognition is a simple task for humans, but it has been proved to be complex for automatic systems. In fact, it is considered an unsolved problem and under active research [116], [166]. The high variability between writers and the cursive nature of the handwriting text are the main difficulties presented by this problem. These difficulties have meant that historically, the practical applications of offline handwriting recognition technologies have been quite limited [41].

The first Offline HTR models were aimed at recognizing isolated characters [134]. The MNIST database [135], described in Subsection 3.1.2, consisted of fixed-size images of

handwritten digits from 0 to 9, is well known. This database, created in 1998, is still widely used today as a problem benchmark for nonlinear models.

Subsequently, the analysis was focused on the recognition of certain handwritten content restricted to some specific application domains. For example, recognize numerals [202], characters [102], bank checks [103] or postal addresses [33], among others. This Thesis focuses on the general HTR problem, and the goal is to develop systems capable of recognizing any image in a handwritten text, regardless of its content, author, or nature. The publication of the IAM database in 2002 [146] settles a turning point to detect an increased number of publications focused on this specific problem.

In this chapter, the main approaches to the general HTR problem are reviewed, including a description of their evolution along time as well as the main contribution of each one. In certain cases, the relationships of each model with the experiments described in this Thesis and their application are also indicated. The review of these methods and their evolution over time provide adequate context to understand the suggested models and the carried out experiments.

The different general HTR models are typically composed of three steps and different approaches are considered for each one. First, the strategy to extract features from the handwriting image. Second, the model to convert these features into an output signal. Third, the way to decode the output signal to predict the text that appears in the input image. It is useful to provide a brief review of the evolution of the developed approaches in each step to understand well the evolution of the different approximations to the HTR problem.

Initially, the strategies to extract image features were rather complex and elaborated because the recognition models were not expressive enough to analyze the image pixel values directly. For example, Doetsch et al. [64] and Kozielski et al. [126] employed *Principal Component Analysis* (PCA) to extract components over fixed-size frames of pixels. Bideault et al. [17] also extracted features, in this case using *Histograms of Oriented Gradients* (HOG). Graves et al. [79] considered features extracted of each column pixel, such as mean and other moments, center-of-gravity, transitions and other aggregations.

The change arose when more complex and expressive models, with deep *Neural Networks* (NN) architectures, composed of *Recurrent Neural Networks* (RNN) and *Convolutional Neural Networks* (CNN), appeared. The use of RNNs, and specially CNNs, make the extraction of features unnecessary and pixel features are used directly as input of the handwriting model. The works by Graves et al. [79] and Bluche et al. [21] are two of the first examples of this strategy. The direct use of the pixels from the image as an input to the model had become a standard over the past few years, and this is the strategy selected in this Thesis.

The second component, the recognition model, is the core of each solution to the HTR problem. These models are the most developed and researched ones, and the evolution (and the main contributions) of these models are detailed in the next sections. This is linked to the important development of neural models in recent years and the emergence of deep architectures. The progress in the architectures for the HTR problem has frequently relied on the advances in Computer Vision, Speech Recognition or Language Modeling [80] [5].

These models were initially based on *Hidden Markov Models* (HMM), first considered separately and later combined with a neural network, which acted as a feature extractor

for the HMM. These networks are usually *Multilayer Perceptrons* (MLP) or *Recurrent Neural Networks* (RNN).

As an alternative to HMM, the first models were based on the *Multidimensional Recurrent Neural Networks* (MDRNN). They were introduced in 2007 by Graves et al. [78], and its architecture exploits the bi-dimensional nature of the handwritten images. In all cases, the authors propose a deep architecture with several *Multidimensional Long Short Term Memory* (MDLSTM) networks. Some examples are included in the works [79], [216] and [21].

Later on, the MDLSTM models were progressively outdated due to their high computational cost, whereas more efficient architectures offered similar or even better results. At that time, an architecture based on one sequence of CNNs and RNNs became popular and it continues to be used nowadays.

Next, the models named as *sequence-to-sequence* (seq2seq), described in Section 4.8, were proposed. These are also the ones used in this Thesis. These models become popular after the papers of Sueiras et al. [201], and Kang et al. [113], both published in 2018 and they persist as an active research line for the HTR problem.

In the third step of output decoding, the handwriting prediction output is transformed into the sequence of text characters that appear in the image. When the model was a HMM, this step is developed with an alignment algorithms like the *Viterbi* one [171] (see for example [23] or [64]). This approach based on HMM with a subsequent alignment was complex because it required several steps as well as quite manual tunning.

A second option for decoding was provided by the *Connectionist Temporal Classification* (CTC), introduced in 2006 by Graves et al. [80]. This layer, specialized in the alignment of sequences of different lengths, allowed the training end-to-end of an HTR model without the need for several steps (as it happened with HMM). The first application to the HTR problem was carried out by Graves et al. [79] in 2009. Since 2013 its use became popular (see for example references [21] and [216]), and it continues to be so actually.

Finally, with the emergence of the seq2seq models, described in the Section 4.8 of this chapter, the third type of decoding algorithm arises. In these models, the decoding is performed by a decoder component of the model, usually composed of a Recurrent Neural Network. This RNN generates the sequential decoding, character by character, of the text contained in the input image.

4.2 Data normalization background

The main difficulty of the HTR problem lies in the large variability of handwritten text. Each individual has his/her own particularities when handwriting. Even the same person writes differently on different situations, for example writing faster or slower or using different sizes of letters. Additionally, different types of paper or pens incorporate additional variations. This is why a great deal of effort has long been devoted to the development of text normalization algorithms to reduce this variability.

In this section we introduce the main concepts of handwritten text image normalization, such as the baseline and upperline detection, and the slant or slope corrections. Next, we review the main contributions of the literature in this area.

Data normalization has been of great importance in HMM-based models. This is due to the limited ability of such models to describe all the variability existing in unnormalized text. This was also the case with other NN-based approaches where the size of the model was limited by the existing computational capabilities of the time when they were proposed. It is at that time that the normalization techniques described below were most developed.

In recent years, the size of NN models that can be trained in reasonable times has increased significantly. Therefore, publications such as [114] or [99] in which no prior normalization of the input images is applied to the model, except for a rescaling and adjustment to a fixed size, have appeared. However, some authors have demonstrated their effectiveness [68] also in this type of larger models. In this Thesis, data normalization techniques are used and their impact in terms of accuracy are analyzed in Chapter 6.

The main normalizations carried out on handwritten images are the following ones:

- Removal of background noise to improve the image contrast.
- Text slope correction.
- Text slant or cursive correction.
- Normalization in size of the ascenders and descenders zones in characters.
- Image resizing keeping its aspect ratio.

The application of the above techniques in the offline HTR literature is quite variable, although three main groups can be distinguished:

- There are authors who perform a 'complete' image normalization and use all of the above normalizations with minor variations, such as: [14], [81] o [26].
- Other authors perform a basic normalization, focusing on the two aspects identified as most relevant, which are contrast normalization and slant correction. In this case they are: [64], [23] or [126]
- Some authors do not apply any standardization, except a resize to fit the data into the input dimension of the model, such as: [66], [114] or [99].

The main contributions of the literature for each of the above normalizations are summarized below. It is important to note that the data normalization procedures described in this section are applied to databases of text written in Latin alphabets. The analysis of normalization techniques in non-Latin languages, such as Arabic or Chinese, is beyond the scope of this Thesis.

4.2.1 Noise removal

The objective of the denoising process is to obtain an image with a white background and well defined characters. For this, the usual thing is to use contrast enhancement techniques. The simplest way to increase the contrast is to binarize the image. That is, a grayscale image with pixel values in range [0,255] is converted into a binary image. For this, a threshold is selected so that if the pixel value is lower than the threshold, the pixel is encoded as black (pixel value 0) and if the value is higher or equal, it is encoded as white (pixel value 255). The most common way to select this threshold is to use the algorithm proposed by Otsu in 1979 [157], which assumes the existence of a bimodal distribution in the pixel image histogram and selects a threshold so that the quotient of the variance between the two segments in distribution and the variance within each segment is maximized.

Some authors use two thresholds, for example, [163] and [19] start from the hypothesis that text strokes pixel values are lower than the background pixels and establish a threshold that maps 5% of the darkest pixels to black and another threshold to map 70% of lighter pixels to white. Intermediate pixels are linearly mapped to the range of values (0, 255) that represent the entire gray scale.

Selecting a single threshold for the entire image can be ineffective, especially if the noise level of the image is different in separate sections of the image. In these cases the use of adaptive thresholds methods such as [179], which calculate neighborhood-based thresholds for each pixel, provide better results.

In any case, the conversion of the handwritten text image into a binarized image is not the best option for its subsequent recognition by an automatic method, because the binarization to two black and white values produces irregularities and artifacts on the edges of lines that distort them.

Hybrid methods which combines the advantages of local binarization and preservation of the gray scale provided better results [214]. The authors modify the Saviola's algorithm [179], so that instead of binarizing each pixel based on its neighborhood, it converts the pixel into a gray value through a linear transformation.

4.2.2 Slope correction

One of the most popular slope correction techniques is the *Hough transform* [67] due to its robustness and simplicity. However, this technique is computationally expensive. Therefore, several authors have proposed variants that reduce the size of the Hough space [158] [29], although the computational cost remains high. Boudra et al. [28] proposed a more efficient method, called *Progressive Probabilistic Hough Transform* (PPHT), which computes the global slope angle estimation of a document with a more adjusted computational cost.

Other widely used method, both to identify the slope and the slant, are based on projection profiles, such as [117] or [160]. The horizontal projection of the text pixels is used to identify the slope, and the vertical projection is used to identify the slant angle. These methods are quite sensitive to image noise, and for applying them it is necessary first to use contrast enhancement techniques such as those described in previous section.

Alternatively, other authors identify the slope angle by estimating the line that best fits a selection of pixels in the image. For example, Papandreou and Gatos [159] use the pixels of the so-called core-region which is the area between the upper and lower baselines of a line or word image. Fig. 4.1 shows these lines that separate the central region of the

text from its ascenders and descenders. The estimation can be done in multiple ways, for example, Kumar et al. [12] propose the adjustment of an ellipse on the image to the word that allows the simultaneous estimation of the slope and the slant from the axes of the ellipse, Okun et al. [156] estimate the best fit line using the maximum eigenvectors of the covariance matrix and Gupta and Chanda [89] use linear regression of the x, y positions of pixels in the core region.



Figure 4.1: Example of slope correction.

4.2.3 Slant correction

As indicated in previous section, methods based on projection profiles are commonly used to identify the slant angle (see for example [117] and [160]). In these cases, the algorithm performs an angle systematic search in a given range by applying the correction transformation for each possible value and obtaining a measure of verticality to be maximized. In the case of work by Kavallieratou et al. [117], the maximum intensity of the *Wigner-Ville Distribution* [51] of the document image horizontal histogram is used as criterion for its skew angle estimation. Pastor et al. [160] propose a method based on the hypothesis that the distribution of the vertical projection profile histogram presents a maximum variation for the deslanted text.

In the work by Gupta and Chanda [59], an algorithm for identifying the slant angle is proposed, which is based on the ability of the *Gabor filter* to detect directional textures. To do this, it applies a Gabor filter of a size equal to the height of text core-region and rotated at a certain angle in a given search range. The angle for which the absolute sum of all pixel values is maximum, is the overall slant angle of the text image.

Another example can be found in Vinciarelli and Luettin [215] that propose a deslanting technique based on the hypothesis that the word image is deslanted when the number of columns that contains continuous strokes is maximum. In the case of this last algorithm, Pastor et al. [161] have shown that it is very sensitive to the usual sources of noise such as salt and pepper, so these authors proposed a variant that softens the requirement of analyzing vertical strictly strokes, extending it to quasi-continuous strokes.

Alternatively, other authors have proposed a direct estimation of the slant angle without the need for a systematic search [145] [228]. These methods usually have a lower computational cost than the previous ones. Gupta and Chanda [59] proposed a second slant angle estimation method in addition to the one described above, which uses the *Fourier* transformation to convert the image of the binary word into the Fourier spectrum. The repetition of points along a given direction activates the frequency space in its perpendicular direction, and this direction corresponds to the skew angle. To identify the most relevant line orientation in the frequency space, the Hough transformation was used.

Other methods of direct estimation of the slant angle are based on the analysis of 8directional chain code of the thinned image, such as [228] or [121]. In the case of work [145], the characters contours were used instead of the thinned image.

4.2.4 Normalize ascenders and descenders

To standardize the ascending and descending regions sizes, it is necessary to identify the baseline and the upperline to delimit the ascenders and descenders zones and scale it in some way. This is actually done in the slope correction step described above. In particular, by means of projection profiles methods [117] [160] or by estimating the baseline and upperline lines [209].

In most cases, the algorithms used for the identification of ascender and descender zones are more or less standard and delimit the zones by lines. However, some authors propose more advanced alternatives. For example, Pastor-Pellicer et al. [162] train a CNN to classify the pixels of the scanned text line into the main body area and into the ascenders and descenders areas.

Finally, it is worth mentioning the work by Wigington et al. [222] that introduces a normalization technique to compensate differences in word size between writers by adjusting the sizes according to the distances between the upper and lower baselines. The published ablation study indicates that this technique provided significant improvements in the model. Although the use of test-time data augmentation and the use of the test lexicon in the decoding (and several different character sets) make it difficult to interpret the results and compare them with other authors.

4.3 Data augmentation

An additional problem to the high variability of HTR is the fact that existing databases for model development are limited in size. Data augmentation is a technique that tries to mitigate this problem by generating additional transformed data from existing ones. In particular, transformations invariant to the text content present in the image are applied to the input image. This strategy is regularly used in other Computer Vision problems such as image classification or object detection.

There are two data augmentation strategies: data augmentation during training and data augmentation in the inference (or evaluation) process, also called test-time data augmentation.

In the first case, for algorithms based on neural networks trained iteratively by means of backpropagation, it is a common practice to introduce the data augmentation process in each iteration. This integration makes possible that, in each epoch of the model training algorithm, a set of input data different from those of the previous epoch is obtained. This allows to perform the training with a theoretically infinite sample of data. This is the usual way of incorporating data augmentation into the HTR problem; for example, see references [68] or [150].

Train data augmentation contributes to improving the generalization capacity and accuracy of models. For example, Coquenet et al. [54] report an improvement of between 0.94% and 1.3% in Character Error Rate (CER) using IAM and RIMES dataset. Word recognition using data augmentation consisting of: contrast modification, sign flipping, long and short scale modifications, and width and height dilations.

Although data augmentation is usually introduced in the training as a first transformation step prefixed at the beginning of each iteration, in some cases the data augmentation is

configured as another layer of the model with trainable parameters. For example, Krishnan et al. [129] introduced an initial *Spatial Transformer Network* (STN) [107] layer configured to perform a data-driven augmentation. So that the parameters of this layer are trained together with the network parameters to adapt the data augmentation parameterization to the training data epoch by epoch, like the rest of the model components.

Another relevant contribution is presented by Wigington et al. in [222]. These authors introduce a new data augmentation technique by applying grid-based local distortions. This technique is based on one previously used by Simard et al. [191] to construct random elastic distortion for single character images. It is performed by applying random distortion on a grid of control points defined for each image and aligned on the baseline of the text. In [222], this technique is used both dynamically in the training process and in the evaluation process (test-time data augmentation).

Finally, it is worth noting that another technique improving performance is *sign flipping* [56], motivated by the fact that the text content is invariant with color or texture. Text and background color inversion is a simple transformation, which works [231] and [56] have pointed out that improve generalization, even if all the training and test data have the same text and background colors.

In the case of test-time data augmentation, the process of generating new modified images occurs at inference time, so in order to evaluate a handwritten text image, multiple variations of the image are evaluated and the result is combined to obtain the final evaluation. For example, Poznanski et al. [169] propose rotations and shear transformations in predefined angles to be used in both train and test time. Wigington et al. [222] use a random distortion on a predefined grid that apply local distortions over the original image and it is applied both in train and test time. Finally, Dutta et al. [68] applied affine transformations, elastic distorsions and multi-scale transformations.

Test-time data augmentation is not a common practice in HTR research and, as discussed in Section 6.4, it generates many difficulties when comparing results from different authors.

4.4 Classical approaches based on HMM

Continuous HTR problem, where no information is available to segment the text into its constituent characters, has been studied for quite some time [165]. Initially, the predominant approach to the problem was the use of *Hidden Markov Models* (HMM) as a technique that allowed the decoding of variable-length transcribed text. This same technique had previously been used successfully in the speech recognition problem [74], which has significant similarities with the continuous HTR problem.

In recent years, the use of HMM for decoding has fallen into disuse in the face of other alternatives based on Neural Networks. However, understanding how it works and how it has been applied to the HTR problem, and especially how it has been combined with models based on Neural Networks, provides a necessary perspective to understand many relevant works on HTR.

In the next subsection, the main concepts related to HMMs are introduced, and in the Subsection 4.4.2 their application to continuous HTR systems is detailed.

4.4.1 Overview of Hidden Markov Models (HMMs)

HMMs are a technique that has been widely used over time, especially on the speech recognition problem [210]. It was also used extensively on the continuous HTR problem until the years 2012-2014 [14] [66] [64]. The technique is well developed and explained in several authors, particularly in Theodore Bluche's Thesis [19] is detailed, both in general and in its particularization for the HTR problem. Therefore, only a general summary of it is included here.

An HMM represents a stochastic process that models the behavior of a sequence called *observed values* and modeled by a random variable denoted by X. The X values are produced by a sequence of *states* modeled by a second random variable denoted by S. The sequence of states remains hidden and acts in the background on the X sequence of output observed values [172].

The sequence $S = s_1, ..., s_n$ represents a set of states connected to each other by transitions subject to a probability distribution P that models the transitions between states: $P(s_{t+1}|s_t)$. This function is called the *transition function*.

The values of each element of the sequence observed values $X = x_1, ..., x_n$ could be emitted with a given likelihood based on a probability mass or density function of the observed value based on the associate state $p(x_t|s_t)$. This function is called the *emission function*.

Additionally, an HMM must fulfill two requirements:

• The sequence of states S must satisfy the condition that it is a first-order Markov process, in which the probability of a future state depends only on the current state.

$$P(s_1, ..., s_n) = P(s_1) \prod_{t=1}^n P(s_{t+1}|s_t)$$
(4.1)

• The value of observed values sequence X at time t depends only on the value of the sequence of states S at the same time t.

$$P(x_1, ..., x_n | s_1, ..., s_n) = \prod_{t=1}^n P(x_t | s_t)$$
(4.2)

To train an HMM we traditionally use the *Maximum Likelihood Estimation* (MLE) [36] criterion applied over the observed values of the train data. In other words, it is identified the set of parameters that maximizes the probability that the feature vectors generated by the training data produce the reference target sequences. The set of parameters to be optimized includes both the parameters of the transmission function and the emission function.

Since the above optimization problem it is not possible to solve it exactly for complex probabilistic models with hidden variables, a numerical method of finding a local optimum is used. The method used is the *Expectation-Maximization* algorithm [60]. More particularly, in the case of HMM, we use the variant called *Baum-Welch algorithm* [6] [7]. The most popular HMM toolkits implement these algorithms or variations, such as *Viterbi's algorithm* [229] [167].

The following subsection describes the application of HMMs to the HTR problem. These models were very popular for solving the HTR problem because they provided many relevant advantages. In particular, they produced good results for low-volume data samples, and it was not necessary to segment words into characters to model word recognition [64]. It was also easy to integrate HMMs with language models for application during decoding as a prior probability.

4.4.2 Hidden Markov Models in handwriting recognition

Historically, HMMs have been used assiduously in the Automatic Speech Recognition (ASR) problem [210], which is a problem with many similarities with the continous HTR problem. In both cases it is a matter of obtaining the transcription of a coded text message, either as an audio or as an image. This, together with the fact that efficient training and decoding algorithms are available for their application, has made them very popular in handwriting as well [13] [196] [98] [145] [183] [227] [14] [64].

In HTR, an output sequence of characters or words with variable length which includes the text transcription in the input image must be obtained. It has to be done from a sequence of input signals extracted from that image that has a length usually different from the length of the output sequence, so it is necessary to perform a decoding step to align the two sequences and build the transcript. HMMs come to solve this alignment and decoding problem.

When an HMM is applied to the HTR problem, some structural decisions are made to reduce the number of model parameters. Usually, a sequential topology with unidirectional left-to-right connections is adopted. In this way, each emitting state has two transitions, the first is a transition to the next stage, and the second is a self-transition over the same state. In Fig. 4.2 we include an example of this topology.



Figure 4.2: Example of typical HMM topology for the HTR problem.

Most of the work on continuous HTR models each character employing an individual HMM. The number of states can be fixed for each character or variable since there are shorter or longer characters requiring more or less states to be modeled correctly [14] [240]. Regarding the emitting states, Gaussian Mixture Models (GMM) with diagonal covariance matrices as density function to model $P(x_t/s_t)$ are used. A complete analysis of different HMM topology optimization methods applied to the HTR problem is provided by Günter et al. in [88].

The text image has to be previously transformed into a sequence of feature vectors to apply the HMM model. The most common technique is to construct handcrafted geometrical features using a left-to-right sliding window over the handwriting image [14]. Sometimes the Principal Component Analysis (PCA) [205] technique is applied to reduce the feature vector dimension, as in [65].

It is also possible to construct the sequence of feature vectors from a visual model applied directly to the input image. The most recent works on the application of HMM to the HTR problem employs visual neural network type models, especially RNNs [26] [235], but also CNNs [23]. There are two approaches to combine a discriminative visual model with a generative HMM model, the *hybrid approach* and the *tandem approach*.

The *hybrid approach* was introduced by Bourlard and Morgan in [31] and it tries to model the dependence of a directly constructed HMM using the visual model output, usually a neural network. It is done in two steps: first, the original features provided by the visual model are used to obtain an initial alignment; second, the posterior estimates are used as scaled emission probabilities in the HMM, which is already trained in the previous step.

The tandem approach was introduced by Hermansky et al. in [93] with the idea of using a neural network visual model output in HMMs. In this case, the output of the visual model are used as a new vector of features, and a completely new HMM is trained using these features. Again, the process is performed in two steps. In the first step, the original features obtained directly from the image are used to build an initial alignment, allowing the training of the visual model. In a second step, the estimations of the visual model are used as a feature stream to train a new *GMM-based* system used in the HMM model.

HMMs were for a long time the best alternative for modeling the continuous HTR problem because of their ability to segment and recognize continuous text simultaneously. However, they had several well-known drawbacks. The main one was that it assumed that the probability of each observation depended only on the current state, which made contextual factors challenging to model. Another problem was that HMMs are a generative model, whereas discriminative models generally perform better in supervised classification problems [81].

In 2009, Graves et al. published the paper [81], where a fully discriminative model was introduced to solve the continuous HTR problem using the CTC algorithm to perform the decoding. This opened the door to full-discriminative models based on neural networks to solve the HTR problem. These models, which are currently used, are described in the following sections.

4.5 Neural networks and deep learning architectures

Until the years 2012-2013, the approaches to the problem of recognizing handwritten words were based on building text features from the images that usually required domain expertise, and after that modeling them through Hidden Markov Models (HMM), as it is described in the Section 4.4. One improvement over the previous approach was the introduction of a Neural Network (NN) [148] that improves the HMM capacity to decode the correct sequence of characters. Neural networks are used to extract relevant characteristics of the image or to transform them as a previous step to the use of HMM to align and decode the transcriptions. This usually improves the HMM capacity to decode the correct sequence of characters.

More recently, deep neural networks [132], which learn a multilayer hierarchy of features, have been the type of model massively selected to analyze unstructured data such as images or texts. These models achieved significant improvements in several complex problems such as image recognition [132] or speech recognition [45]. The case of handwriting recognition is not an exception and several deep architectures has been proposed for the word and line recognition problems with significant improvements in the results, as shown in [79], [19] or [170].

The deep neural networks became more popular due to their success in the problems of Computer Vision [132] and fueled by the increased computing capacity of the computers. Their high popularity is also linked to the appearance of software frameworks which allow the efficient and simple implementation of the neural models with a high number of parameters and multiple layers.

Since then, neural networks have become the most commonly used methods to address the problem of offline handwriting recognition. Different authors are proposed multiple architectures and use different types of networks to handle this problem.

In the next sections, we describe in detail the use of this different neural networks architectures to solve the problem of handwriting recognition. But first we include two subsections to introduce the historical background and introduce the most relevant concepts of neural networks applied to the HTR problem.

4.5.1 Historical background

This subsection includes an introduction to neural networks, starting with a brief summary of its history. Further details about this history can be found in [118]

The first appearance of neural networks as a mathematical model, which mimics the functioning of neurons in the brain, was presented by McCulloch and Pitts [148] in 1947. These authors introduced a model of simplified neurons as a circuit that can perform computational tasks. Neural networks begin to gain interest until 1969 when Minsky and Papert, in their book Perceptrons [152], expose several problems and limitations of such models.

It was not until the 1980s, with the application of the *backpropagation* technique for their training, that neural networks again gained the attention of the scientific community. The backpropagation technique was proposed by several authors in the 1960s and in 1974. Paul Werbos proposed its use in neural networks in his Thesis [221]. But it did not become popular until the publication in 1986 by David Rumelhart, Geoffrey Hinton and Ronald Williams of two papers: "Learning representations by back-propagating errors" [175] that clearly details its practical implementation as it is known today, and "Learning internal representations by error propagation" [174] that addresses the problems evidenced by Minsky almost 20 years earlier and shows how neural networks can be successfully trained to solve complex problems. Only a few years later, and by applying the techniques proposed in these two papers, one of the first practical applications of neural networks appeared, precisely for the problem of reading handwritten text [134].

Neural networks were once again a topic of interest for the scientific community thanks to the application of backpropagation for their training and to the publication in 1989 of the paper "Multilayer feedforward networks are universal approximators" [97]. In this work,
it was mathematically demonstrated that a multilayer neural network can theoretically implement any function, which eliminated the main problem that Minsky had raised to them twenty years earlier.

But soon arises a problem, precisely related to backpropagation, that makes impossible to train multilayer neural networks with several layers. In short, the fact that when the backpropagation algorithm distributes the responsability for the error among the layers of the network and it assigns gradients to try to reduce the error in the next iteration. These gradients either decay to zero exponentially with the number of layers or they explode. This is known as the 'vanishing or exploding gradient problem' [11]. Additionally, in 1992, a new algorithm called *Support Vector Machines* (SVM) [27] appeared. In a very basic way, it can be described as a mathematically optimal way of training an equivalent of a two-layer neural network. The SVM begins to show itself as superior to the difficulty of working with neural networks, as indicated by the first comparisons precisely in the handwritten text recognition problem [225].

The interest in neural networks declined again, until the publication in 2006 of the paper: 'A fast learning algorithm for deep belief nets' [94]. It shows that it is possible to correctly train the multilayer neural networks provided when their weights are initialized intelligently rather than randomly and proposes an initialization procedure by means of unsupervised layer-by-layer learning. This paper marks a turning point in the interest in neural networks, which is now being rekindled. From this moment until today, there is an escalation in the application of neural networks to a multitude of problems, as well as in the number of layers and in their accuracy. All this fed by the increase in both the size of the datasets and the computational capacity of computers.

Nowadays, neural networks completely dominate a multitude of research areas and in both academic and business environments it is one of the most important topics of intense research [184] [55].

4.5.2 Multilayer perceptrons

An Artificial Neural Network (ANN) is a computing system based on a collection of connected units or nodes organized in layers. The connections between nodes are called edges, and they are directed and have an associated parameter called weight, which modulates the signal as it passes through each edge. There are multiple network architectures, oriented both to the construction of supervised and unsupervised models. In this section we focus on the so-called feedforward neural networks or Multilayer Perceptrons (MLP). For these cases, the network receives an input signal x through a set of initial nodes called input layer and it produces an output signal y through other nodes called output layer. The intermediate layers, through which the signal flows and is transformed, are called hidden layers h. Fig. 4.3 shows a basic neural network scheme with the above components. These models are called feedforward because the input information x follows a single forward path until it is transformed into the output information y. When feedback connections are added to such a network in which the output is fed back into itself then a particular type of network called *Recurrent Neural Networks* (RNN) is obtained.

Fig.4.4 shows a detail of the operation of a particular node of the network. Each node receives multiple input signals x_i through its incoming edges, modulated by their respective weights w_i . It merges them by means of a combination linear function, which is



Figure 4.3: Feedforward neural network basic architecture.

usually the sum, and generates an output signal y which is a real number. This output is modulated by an non-linear activation function before being sent to the next nodes in the network. This activation is a monotonic function that usually limits the range of variation of the output signal.



Figure 4.4: Neural network node detail.

Formally, the input signals $x_1, ..., x_n$ coming from the previous layer are multiplied by their respective weights $w_1, ..., w_n$ and combined at the node, in this case with the sum function. To the previous result a bias b is added and to this the activation function f is applied as indicated in the equation 4.3.

$$y = f(b + \sum_{i=1}^{n} w_i x_i)$$
(4.3)

The combination function is usually standard, calculated as given in the equation 4.3 as a linear combination of the inputs x multiplied by the weights w. The activation function must be monotonic. Traditionally the most common choices were the hyperbolic tangent function, which rescaled the output in the interval (-1, 1) or the logistic sigmoid which rescaled it in (0, 1). But with the advent of deep architectures, the preferred activation function is the Rectified Linear Unit (ReLU) [108] calculated as: f(x) = max(0, x).

The goal of a neural network is to model the relationship between the input data x and the output data y by means of some function f that approximates it. This function

depends on the structure of the network that has been defined and a set of parameters *theta* that characterize it, which are basically the weights w of the edges of the network. To make the adjustment, the neural network defines a mapping $y = f(x, \theta)$ and through a process called *training*, it learns the values of the θ parameters that best approximate the function f to the data. To do this, a learning algorithm is employed that uses the experience provided by the known data (x, y) to iteratively search for the value of the θ parameters that optimize the fit of f to these data.

The common learning algorithm used is *backpropagation*. This algorithm, based on gradient descent, allows to optimize the parameters of a network through an iterative process that seeks to minimize the error of the network by representing a set of known data (x, y), which are called *training data*. More formally, the backpropagation algorithm is defined by three elements:

- A neural network $\hat{y} = f(x, \theta)$, where θ represents the set of parameters of the neural network.
- A data set $X = (x_1, y_1), ..., (x_n, y_n)$ where x_i is the input and y_i is the desired output of the network with x_i input.
- An error function $E(y, \hat{y})$ that measures the error between the desired output y and the calculated output \hat{y} for a given value of the parameters θ .

At each iteration t, the algorithm performs two steps:

- A first step called *forward* in which, given the value of the current parameters θ^t , the estimated value of \hat{y} is calculated as $\hat{y} = f(x, \theta^t)$. And then the error $E(y, \hat{y})$ is calculated as a function of the difference between the calculated values \hat{y} and the expected values y.
- In a second step called *backward*, the θ parameters are updated as a function of the gradient of the error function E with respect to the θ parameters as shown in the next equation:

$$\theta^{t+1} = \theta^t - \lambda \frac{\partial E(X, \theta^t)}{\partial \theta}$$
(4.4)

where *lambda* is a training parameter called *learning rate* that determines the step size at each iteration of the algorithm and allows to control how quickly the model adapts to the training data.

In the HTR problem, the MLP networks have been mostly used as part of the solutions based on the Hidden Markov Models. In certain cases, they have been used to predict the HMM states within a hybrid approach like [69]. In other cases, they have been applied to extract features which can be used later in models based on GMM and HMM, following the tandem approach like the one included in [66].

In the hybrid approach [31], a NN replace the traditional Gaussian Mixture Models (GMM) as emission models for the HMMs. This approach has been implemented with

diverse neural networks. For example, [66] and [69] use MLP, while [10] and [23] use convolutional networks. In the tandem approach [93] the NN and the HMM are combined using the NN as a features extractor for the GMM-HMM system. Similarly to the previous case, the network is also trained to predict the HMM states or characters, although the probabilities are not further directly used in the visual model, the network works as a feature extractor. For example, the work [127] uses the activations of a *Long Short Term Memory* (LSTM) network, described in the Subsection 4.5.3, as the features to build a GMM-HMM model. Considering that the activations of a NN are highly correlated to themselves, a transformation is usually applied to eliminate such correlation, frequently Principal Component Analysis (PCA) [205] or Linear Discriminant Analysis (LDA) [153].

The wide variety of alternatives to use the outputs of the NNs is a great advantage of the tandem approach. However, the main inconvenient lays on the fact that it requires a complete training of the HMM after the outputs of the NN are extracted. Besides, the non-correlation and the reduction of the characteristics introduce additional parameters which have to be manually tuned. Nonetheless, the comparative studies indicate that this strategy is superior to the hybrid approach [66].

In the next sections, it will be shown that the construction of HTR systems exclusively based on the NN will not use MLP networks. They are based on the *Recurrent Neural Networks* as well as on *Convolutional Neural Networks*.

4.5.3 Recurrent Neural Networks

Recurrent Neural Networks [175] are one type of networks aimed at sequential data processing $x_1, ..., x_n$. These networks can model the sequential relationships in data when they are relevant for the goal of the training. For this, they rely on the idea of sharing parameters between the different nodes of the recurrent layer while the input data sequence is processed. Thus, in a RNN each output element h_t is function of the previous output element h_{t-1} and, based on this recurrent definition, is a function of the input elements $x_1, ..., x_t$. Fig. 4.5 shows the scheme of a basic RNN.



Figure 4.5: RNN schema: a) compact, and b) unfold-over-time.

According to the notation of Fig. 4.5 the equations of a basic RNN are defined as follows:

$$h_t = \tanh(W_{hh}h_{t-1} + W_{xh}x_t)$$

$$y_t = W_{hy}h_t$$
(4.5)

These equations describe a basic strategy in order to apply the recurrence and the activation *tanh* in a recurrent connection. This model is the most common one, although the recurrent networks accept multiple variations, and some of them are introduced in the next sections.

The training of a RNN can be done with a modified version of the backpropagation algorithm, named *Back-Propagation Through Time* (BPTT), which is an optimization of the backpropagation applied on a RNN unfolded. The BPTT includes a truncation strategy in the calculation of the gradients which ensures the numeric stability and, for efficient computation, intermediate values are cached.

The basic recurrent network described previously has a causal structure because in time t only information from the past $x_1, ..., x_{t-1}$ and the current time x_t are used. However, sometimes the prediction y_t has to depend on the whole context, both previous and after the moment t. For example, in the HTR offline problem, the identification of a specific character in the center of the text can benefit from analyzing both the previous and subsequent characters.

In order to include the whole text in the model *Bidirectional Recurrent Neural Networks* (BRNN) [186] are used. They have been successfully applied to the HTR problem during a long time [82] [79]. These networks blend one RNN executed in an increased direction of t. From the beginning of the sequence, with other RNN executed in a decreasing direction of t, from the end of the sequence towards its beginning. Fig. 4.6 shows a typical scheme of a BRNN where h_t is the RNN state that moves forwards through time and g_t is the RNN state that moves backwards through time. The output sequences of both RNN are combined at each step using an aggregation function, being concatenation the most common one, although the sum or the mean applied elementwise are also occasionally used.

This idea can be naturally extended to two dimensions or more, as it happens for images, generating the *Multidimensional Recurrent Neural Network* (MDRNN) architecture, which are discussed in Subsection 4.5.5 and which have been frequently use for the HTR problem, as mentioned in [79] or [164].

Similarly to what it happens with the MLP networks, where multiple dense layers are connected to create a multi-layer architecture, it is also common to connect multiple recurrent layers so the layer output sequence is passed as an input sequence to the next layer in the stacked RNN architecture, shown in the left side of Fig. 4.7.

The sequential combination of multiple Bidirectional RNN interconnected, is also common and it generates a Bidirectional stacked RNN architecture as the one shown in the right side of Fig. 4.7. In Chapter 6 of this Thesis we explore the application of these previous architectures to the HTR problem.

In this background chapter, basic RNNs with a vector h as a recurrent unit have been explained. However, previous studies [11] confirm that these types of networks have a vanishing gradient problem and gradients propagated through many layers tend to vanish. This makes it impossible for the network to model long dependencies along time. The most common solution to this problem is the use of gated RNNs as these networks are



Figure 4.6: Bidirectional RNN schema.



Figure 4.7: Multi-layer RNN schemas: a) Stacked RNN. b) Bidirectional stacked RNN.

based on creating paths through time which avoid the problem of vanishing gradient. The most commonly used ones are the *Long Short-Term Memory* (LSTM) networks and the *Gated Recurrent Units* (GRU) networks. Several experiments with these two types of gated RNN to the HTR problem were performed in Chapter 6 for specific seq2seq models.

LSTM

Long Short-Term Memory (LSTM) networks are a special type of Recurrent Neural Networks (RNN) which are able to learn long-term dependencies. LSTM networks were introduced by Hochreiter and Schmidhuber [96] and include two ways to translate the previous information across the net: the output vector h and the state vector c, that combined using three gates, are explicitly designed to store and propagate long-term dependencies. Fig. 4.8, includes a LSTM cell schema.



Figure 4.8: Long Short-Term Memory (LSTM) cell.

The gate i, named the input gate, learns which values will be updated in the state vector. The gate f, named the forget gate, learns the information from the previous state that can be thrown away. With the output of these two layers the network creates a vector of new candidate state values c. Finally, the gate o, named the output gate, learns what information will go to the output h. The following equations model in detail the LSTM gate.

$$f_{t} = \sigma(W_{f}[h_{t-1}, x_{t}] + b_{f})$$

$$i_{t} = \sigma(W_{i}[h_{t-1}, x_{t}] + b_{i})$$

$$o_{t} = \sigma(W_{o}[h_{t-1}, x_{t}] + b_{o})$$

$$\hat{C}_{t} = \tanh(W_{c}[h_{t-1}, x_{t}] + b_{c})$$

$$C_{t} = f_{t} \odot C_{t-1} + i_{t} \odot \hat{C}_{t}$$

$$h_{t} = o_{t} \odot \tanh(C_{t})$$

$$(4.6)$$

where W and b are trainable parameters, σ is the sigmoid function, x is the input sequence, i is the input gate, f is the forget gate, o is the output gate, C is the state vector and h is the output vector and \odot denotes the element-wise product.

GRU

The *Gated Recurrent Units* (GRU) [43] is an alternative gated RNN to the LSTM architecture with less parameters. There is no clear consensus to determine the best type of architecture. Depending on the problem, sometimes the performance of the LSTM can be superior, whereas in other scenarios the GRU offers better results [83]. However, in general, it is necessary to experiment both for each specific problem. As exceptions, some authors as [48] and [86], indicate that GRUs exhibit better performance that LSTM networks on smaller datasets. In reference [43] it is pointed out that the GRU have been demonstrated to be more efficient that LSTM networks for manage long-distance temporal correlations in different applications.

The main difference between a GRU network and a LSTM network is that in the GRU only one gated simultaneously controls the forgetting factor and the decision to update the state unit. Fig. 4.9 shows a scheme of one GRU network which uses two gates. The reset gate r which controls the parts of the current state will be used to calculate the new state and the update gate u which controls what information will be moved to the output h.



Figure 4.9: Gated Recurrent Unit (GRU) cell.

For the notation of Fig. 4.9, the equations for the forward step are the following ones:

$$u_{t} = \sigma(W_{u}[h_{t-1}, x_{t}] + b_{u})$$

$$r_{t} = \sigma(W_{r}[h_{t-1}, x_{t}] + b_{r})$$

$$\hat{h}_{t} = \tanh(W_{h}[r_{t} \odot h_{t-1}, x_{t}] + b_{h})$$

$$h_{t} = (1 - u_{t}) \odot h_{t-1} + u_{t} \odot \hat{h}_{t}$$
(4.7)

where W and b are trainable parameters, σ is the sigmoid function, x is the input sequence, r is the reset gate, u is the update gate and h is the output vector and \odot denotes the element-wise product.

RNN in handwriting

The initial use of the neuronal networks in the HTR problem focused on combining them with HMM in the hybrid approach [37]. In the specific case of the RNN, multiple combined models with HMM have been proposed, for example [198], [187] or [182].

The use of models with only RNN was restricted to the recognition of characters [30], mostly because the target function needs one training signal for each element of the entrance sequence, and it implies that data has to be pre-segmented. To our best knowledge, the work by Graves et al. [81] is the first paper where one full RNN architecture was proposed for the HTR problem. In order to solve the previous limitation, these authors used one *Connectionist Temporal Classification* (CTC) layer as an decoder. This layer performs the alignment between the output of the model and the target sequence of characters. CTC only imposes that the input sequence must have a larger or equal length to the output sequence. The proposed model is a one-layer Bidirectional LSTM with 100 memory cells. From that work, these RNN are going to be present in one way or another in most of the publications related to the HTR problem.

The popularity of RNN approaches to the HTR increased with the help of the CTC. Initially, the most common approach used was Multidimensional RNN, especially the MDLSTM models (which are discussed in detail in Subsection 4.5.5). MDLSTM seemed better suited to the bi-dimensional nature of the input data, that is the images of the handwritten texts.

In 2017, several publications such as [22] and [170] criticized the high computational cost of the MDLSTM models. These publications proposed alternative models based on the CNN-RNN-CTC architecture which are discussed in depth in Subsection 4.5.6. In these models, the RNN component is usually composed by several bidirectional layers stacked (like in [170] or [68]).

RNNs are also the core of current approaches to the HTR problem, based on sequenceto-sequence (seq2seq) models. In these models, both the encoder and the decoder include several types of RNNs. For example, Bluche et al. [21] used MDLSTM, Sueiras et al. [201] used Bidirectional LSTM and Kang et al. [113] employed GRU layers.

In recent years new approaches without using RNNs in their architectures have emerged. For example, Kang et al. [115] used transformer layers [213] to build the HTR model. Coquenet et al. [52] have proposed a model which uses only Convolutional Neural Networks (CNN) and the CTC as a decoding layer. Convolutional Neural Networks (CNN) and their detailed application to the HTR problem are discussed in the next section.

4.5.4 Convolutional Neural Networks

Convolutional Neural Networks (CNN) [134] are a type of neural network specialized in modelling the information with a grid-like structure, such as images considered a twodimensional pixel grid. These networks use the convolutional mathematical operation (*) instead of the standard matrix multiplication to calculate the network output.

One convolutional layer transforms an input array X into an output y commonly known as feature map, by convolutioning the input with a array of parameters K known as kernel. This operation is performed locally so that the output $y_{i,j}$ is calculated based on the neighborhood of $x_{i,j}$ with the kernel convolutional operation. Fig. 4.10 illustrates this operation with a two dimensional input and a 3×3 dimension kernel which generates an output with a single feature map.



Figure 4.10: CNN schema for a kernel of 3×3 and one feature map.

On the notation of Fig. 4.10, the equation corresponding to the forward step for this network is as follows:

$$y(i,j) = (K * X)(i,j) = \sum_{n} \sum_{m} X(i-n,j-m)K(m,n)$$
(4.8)

where y is the the output feature map, K is the kernel of trainable parameters and X is a 2-D input array.

Convolutional networks allow to identify local characteristics of the input. For this purpose, small kernels are used, generally with a dimension much smaller than the input dimension. These small kernels make the CNN very computationally efficient and robust to input transformations such as scale changes, translations or rotations.

A standard convolutional architecture usually contains three elements. First, a convolutional layer as described above. Secondly, a nonlinear activation function, usually a *Rectified Linear Unit* (ReLU) function, which controls the range of variation of the output. The third stage consists of applying a *pooling operation* on the output of the activation function.

The pooling operation replaces the network output at a given position (i, j) by a statistic summary of the output at the nearby positions. The most common pooling function is called *max pooling* [239] which assigns to the position (i, j) a value calculated as the maximum of the outputs within a rectangular neighborhood. Pooling helps the convolutional architecture to be invariant to small translations of the input image. This is a very relevant property for the case of handwritten text images, for which a translation of the image does not modify the text written on it that is intended to be recognized.

Convolutional architectures play a fundamental role in all models working with data of spatial nature, especially 2-D images, but also 1-D time series or 3-D videos. Over the years, multiple convolutional architectures have been proposed for different problems related to image analysis. In general, these architectures include at least the three elements of convolution, activation and pooling indicated above. A complete review of these architectures is out of scope of this Thesis. For a complete reference of the different convolutional architectures proposed in recent years see Khan et al. [120]. In this Thesis, we have worked with three types of convolutional architectures, which are: the LeNet [225], VGG [192], and ResNet [91]. These architectures have been used in both the character recognition experiments in Chapter 5 and the word recognition models in Chapter 6.

LeNet architecture was the first proposed one for handwritten digit recognition in 1995. It is characterized by using 5×5 kernels and having two layers: the first with 20 feature maps and the second with 50 feature maps. VGG architecture is characterized by proposing a convolution-convolution-pooling layer configuration with 3×3 kernels. ResNet architecture is characterized by proposing direct connections (residuals) between the input and output of each convolutional component. Additionally, it employs the regularization strategy called *batch normalization* [105]. In the following chapters, more information of these three architectures are provided, including a detailed definition of the architectures and diagrams of specific configurations used in the experiments of this Thesis.

CNN in handwriting

In the HTR problem domain, CNNs have been initially used for handwriting digits recognition [134]. Later, these networks were applied to handwritten text recognition in specific domains such as online handwriting [9] or bank checks transcription [133] [135].

One of the first applications of CNNs on the general handwriting recognition problem was proposed in 2013 by Bluche and collaborators [23]. In this work, a CNN-HMM model was implemented using the tandem approach. The convolutional architecture selected for feature extraction was inspired by LeNet-5 model. The results with CNN were superior than using handcrafted features and similar to those obtained with LSTM networks.

In 2016, several related publications appeared [128] [131] [169], which applied the nearest neighbor approach to the HTR problem. In these works, a CNN-type model is used to extract the features employed to identify the nearest neighbors and perform text recognition. In the same year, a full convolutional model was proposed for the identification of n-grams of rank 1, 2 and 3 in images of handwritten words that are then used to recognize the text by means of *Canonical Correlation Analysis* [169].

In 2017, CNNs take on a central role in an HTR recognition model [22]. In particular, convolutional gates where used to enable context-sensitive and hierarchical feature extraction. A novel strategy of pre-training the convolutional part with multiple corpora of different languages was also proposed. Subsequently, the rest of the network is trained with each corpus separately.

Starting in 2017 with the publications of Puigcerver [170], and Bluche and Messina [22] popularizing the CNN-RNN-CTC architecture, CNNs are included in almost every approach to the HTR problem to nowadays. With the emergence of the seq-2seq-type models (discussed in Section 4.8), CNNs remain an essential part of the solutions. In these cases, they are also always incorporated as the first component of the network, with the goal of extracting visual attributes from handwritten text images.

Finally, it should be noted that recently, some authors have proposed a full convolutional architecture with CTC to avoid the costs of running LSTM network architectures. Yousef et al. [231] proposes a Fully Convolutional Network (FCN) architecture for the line recognition problem. The model is based on the concatenation of several blocks that use attention gates to control the information that is transferred between them. In this way,

signals that are not relevant, such as background noise, are filtered out. The attention gates used are based on the highway network architecture proposed by Srivastava et al. [195]. The same author presents, in a later publication [230], a fully convolutional architecture for HTR in pages with multiple lines. The paper proposes a strategy for adapting any fully convolutional model (at the line level) for text recognition (at the paragraph level). This adaptation is performed without needing the transcription to be segmented into lines, which is an advantage over other page-level or paragraph-level approaches, such as [21] or [223].

4.5.5 Multidimensional Recurrent neural netwoks

In this subsection, we first introduce the *Multidimensional Recurrent Neural Networks* (MDRNN), describing their main characteristics, and then detailing their application to the offline handwritten recognition problem. Next, we review the usage of MDRNN in the HTR problem.

MDRNN network architectures were introduced in 2007 by Graves et al. in [78], applying them to the image segmentation problem. MDRNNs generalize standard RNNs by defining recurrent connections along multiple dimensions, as opposed to RNNs that provide recurrence in a single dimension. This allows these types of networks to model spatial recurrences such as those that appear in handwriting images. Additionally, they are more robust to local distortions than the standard RNN along the different input dimensions. Distorsions include rotations or shears, which are very common in handwritten text.

To illustrate how MDRNNs take into account the context in multiple dimensions, Fig. 4.11 shows how the calculation of the hidden layer at position (i, j) use the values of the hidden layer at positions (i - 1, j) and (i, j - 1), respectively.



Figure 4.11: MDRNN 2 layer context example (adapted from [78]).

One-dimensional RNNs can define the context of a given point in the sequence in two different ways depending on whether the sequence is followed in the forward or backward direction. When the sequence is followed in both directions, a Bidirectional Recurrent Neural Networks (BRNN) [186] are obtained. In the same way, MDRNNs can define the context of a given point in multiple ways depending on how the different dimensions are followed.

Fig. 4.12 shows the two contexts associated to a one dimensional RNN and comparatively some possible contexts associated to a two dimension MDRNNN, which incorporates four different networks starting each one of them in a corner of the analysis surface and with a different directionality. The exact definition of the contexts in the multidimensional case can be quite varied, depending on the computation order strategy of the input elements. For example, in the case of a 2-D input, some possible orders can be: by rows first, by columns first or starting diagonally from each of the origin points of the corners.



Figure 4.12: MDRNN example of contexts. (a) 1-dimensional BRNN. (b) 2-dimensional MDRNN (adapted from [78]).

MDRNNs are usually built with LSTM networks, and abbreviated as MDLSTM. LSTM networks avoid the vanishing gradient problem and ensure that the range of the context used in a given step is large enough, since they are able to model relationships across the entire context.

MDRNN in handwriting

The first use of MDRNNs in the offline handwriting recognition problem was proposed in 2009 by Alex Graves and Jürgen Schmidhuber [79], the same authors who two years earlier had introduced MDRNNs in [78]. MDRNNs were proposed as a unified alternative to address a problem such as offline handwriting recognition, which requires both Computer Vision and Sequence Learning. Previously, the proposed systems addressed these two requirements separately, the former one with complex preprocessing and feature extraction techniques and the latter with sequence learning through models, such as HMMs that generated the transcriptions. This paper proposed the specific use of MDLSTM networks built with multidirectionality from the four corners of the text image to be processed.

The architecture proposed in [79] included several MDLSTM type layers with interleaved convolutional and feedforward layers and a final CTC layer [80], which is a type of decoding layer also introduced by Graves et al. three years before. Fig. 4.13 shows a schematic representation of the proposed architecture, which was later adopted by other authors with few variations. This architecture is fully trainable from raw pixel to text decoding, it does not require any alphabet-specific preprocessing like HMMs and it is directly applicable on any character set of different languages.

Even if the work by Graves and Schmidhuber [79] showed a significant improvement in recognition results, it was not until 2012 when other authors [149] start embracing



Figure 4.13: MDLSTM standard architecture (adapted from [216]).

MDLSTM networks in general HTR problem, almost always with a few changes about the initial architecture proposed which is shown in Fig. 4.13.

In 2012, the French company A2IA in the paper [149] explained their approach to the RIMES-ICDAR2011 competition. It stated the use of a MDLSTM architecture with CTC as part of an ensemble consisting of two models based on HMM and the MDLSTM model. To our best knowledge this is the second reference of MDLSTM use applied to HTR after the original 2009 reference. The paper does not detail the architecture used but indicates the results obtained by each model in the ensemble. The results of the MDLSTM model were clearly superior to the other two HMM-based models: 9.2% of WER of the MDLSTM model task on RIMES database. With the MDLSTM model alone, this paper already obtained a clearly better result in the competition than the rest of the participants.

In 2013, the same company presented the same architecture in the competition of the Arabic Handwriting recognition OpenHaRT [24], also with good results. From here on, during the years from 2014 to 2017, the architecture based on MDLSTM and schematized in Fig. 4.13, appears in several papers about offline handwritten text recognition, with very competitive results (see for example the works of [164], [197], [216], [21], and [40]).

Pham et al. [164] proposed to improve the standard MDLSTM-based architecture with the application of the *dropout regularization strategy* [95]. This technique randomly removes a fixed percentage of hidden units from a neural network at each iteration of the training phase of the network, so that in the backpropagation error step, only some of the weights are updated at each training step. The dropout has a reducing overfitting effect and accelerates the convergence of the training process in deep models with multiple layers. Then, in the evaluation phase, all hidden units and weights are used. This paper [164] proposes the application of dropout only in the feedforward layers of recurrent networks, without including it in the recurrent connections, to avoid affecting the ability of RNNs to model sequences. A detailed analysis of different dimensioning of the recurrent layers and, especially, on which layers to apply dropout is performed. The authors found that the optimal strategy for this architecture is the application of dropout on all recurrent layers as opposed to the usual practice at that time that was using it only on the last layers of the model.

Until 2016, the application of MDLSTM networks had the drawback that the existing implementations of MDLSTM networks were CPU-based and computationally expensive,

which meant that training a network could take weeks [197]. The paper by Voigtlaender et al. [216] made publicly available an efficient implementation of GPU-based MDLSTM networks that allows the training of these architectures with more layers and with more hidden units in each layer than before. The paper proposed an implementation strategy for MDLSTM contexts with a parallelizable diagonal pattern instead of the usual hierarchical row and column implementation. Finally, it was shown that the increase in number and size of MDLSTM layers and the adoption of a different initialization strategy (proposed by Glorot et al. in [76]) provided substantial improvements in both the convergence speed of the train step and the accuracy in the evaluation step.

The MDLSTM architecture was proposed by Bluche et al. in [21] and [20], as the basis for the implementation of a handwritten paragraph-level recognition systems, without the need of line segmentation. These are the first cases where an end-to-end handwritten paragraph recognition system is proposed. The solution is based on the combination of an MDLSTM-like network with the usual architecture, augmented with an attention mechanism [4].

In 2017, a variant of the MLSTM network called *Separable MDLSTM* (SepMDLSTM) is proposed in the paper [40] by Chen et al., where the rows and columns of the input image to the recurrent layer are traversed sequentially in two successive layers instead of in parallel layers as previously done. This new architecture is applied in a multi-task problem for simultaneous script identification with a classifier and handwriting recognition with a CTC layer.

Finally, in 2017 two publications by Bluche and Messina [22] and Puigcerver [170], respectively, criticized the high computational cost of MDLSTM networks and proposed the same alternative architecture. This is based on first employing a set of CNN-type layers, that capture visual attributes of the text image, followed by LSTM-type layers that capture the recurrent nature of the handwritten text. In both cases, the CTC is used as the final decoding layer and yields results that significantly improve those obtained with MDLSTM architectures.

After the above publications, MDLSTM-based architectures are no longer used in the handwritten recognition problem and CNN-LSTM-CTC based architectures become the new standard. These architectures are discussed in the next section.

4.5.6 The CNN-LSTM-CTC Architecture

As described in the previous subsections, CNNs and RNNs had been frequently used in the HTR problem although usually separately. In 2016, a paper by Suryani et al. [203] proposed an architecture for HTR based on CNNs followed by BILSTM-type of RNNs, using HMM for final transcription decoding. The model was applied to Chinese handwriting recognition with discrete results compared to those obtained at that time with MDLSTM type architectures. In any case, it is, to the best of our knowledge, the first reference of a CNN-RNN architecture applied to the HTR problem. In the same year, Shi et al. [189] published, and also proposed a CNN-RNN architecture applied to the recognition of printed text in natural images. But it is in the publications by Bluche et al. [22] and Puigcerver [170] where the CNN-LSTM-CTC based architectures become the reference to address the handwritten recognition problem. The general architecture proposed by the different authors is very similar, although each author introduces some variations that are described below. A general outline of the architecture is shown in Fig. 4.14.

These architectures have proven to achieve good results in another text recognition tasks. For example, Breuel [34] shows how this architecture improves the obtained results with BLSTM-CTC architecture in optical characters recognition (OCR) task. Li et al. [139] used it to obtain the state-of-the-art in the license plate recognition problem. Shi et al. [189] also used this same architecture for scene text recognition in natural images.



Figure 4.14: Standard CNN-RNN-CTC general architecture.

Bluche and Messina [22] proposed a standard CNNs alternative, called gated-CNN. Basically, the idea is that the convolutional filter was followed by a sigmoig activation that controls the propagation or not of the feature to the next layer, depending on the values of the analyzed position and the neighboring positions. The paper employs a large multilingual corpus of 132,983 lines in a total of 7 languages with Latin characters. The convolutional layers are trained with this joint corpus and subsequently a fine tuning per language of the LSTM layers is performed.

In Puigcerver [170] the same previous architecture is proposed: CNN layers followed by LSTM layers with a CTC decoding. The author uses batch normalization [105] and dynamic train data augmentation (as it will be described in Section 6.4.5), which are techniques that have been successfully employed in image classification problems with convolutional architectures at that time.

Dutta et al. [68] also proposes the CNN+LSTM+CTC architecture but introducing at the beginning a spatial transformer network (STN) layer [107]. It is a trainable layer that produces geometrical transformations on the input similar to those produced in handwriting due to random hand movements. In the published ablation study the authors indicate an improvement of the Word Error Rate (WER) of 9% for IAM database at word level by the introduction of the STN layer in the model. Unfortunately, comparisons with results from other authors are difficult since the paper uses a charset reduced to lowercase characters, test lexicon decoding and test-time data augmentation.

Just as MDLSTM-based models have been used as a basis for building page-level HTR systems as in [21], CNN-LSTM-CTC architectures have also been used for the same purpose. In 2018, Wigington et al. [223] propose an end-to-end full-page handwriting recognition system composed with three elements. First, a Region Proposal Network to identify the start position of each text line in the page. Second, a novel line follower network that pre-process and convert text line into a normalized line image. Third, a CNN-LSTM-CTC type network that transcribes the normalized line.

Recently, some publications question the efficiency of CNN-RNN architectures for modeling complex spatial and temporal relationships like the ones in handwriting text. In particular, it is questioned the fact that both relationships could be extracted optimally with two sequentially connected components but not integrated. As alternative, variations of recurrent networks to introduce the spatial component [99] or variations of convolutional networks to introduce the temporal or sequential component [104] are proposed.

Huang et al. [99], propose a new neuronal architecture that combines convolution, pooling and recurrence in a unified framework named *Convolutional Recurent Neural Network* (CRNN). The CRNN can be seen as an improved version of RNN where the direct connection between input and hidden layer is replaced by a local connection and shared weights like a convolutional kernel.

In Ingle et al. [104], it is discussed the model scalability with a RNN component because its execution cannot be parallelized like with a CNN. As alternative to LSTM, it is proposed the use of *Gated Recurrent Convolutional layers* (GRCL) [218]. In summary, these layers are conventional convolutional layers with added recurrent connections in the time direction. Additionally, they include a gatting mechanism that allows to control the information propagated by these recurrent connections like in GRU layers.

4.6 Connectionist Temporal Classification (CTC)

The Connectionist Temporal Classification (CTC) is an output layer defined to align sequences. The layer can directly transform an input sequence S_i into an output sequence S_o provided that the size of the input sequence is larger or equal than the size of the output one, i.e. $|S_i| \ge |S_o|$. This layer performs the output decoding of the visual model by converting it into the sequence of characters that transcribe the text contained in the image. This allows training HTR models in a single step, since it does not need an additional decoding process like the one needed by HMM-based models.

The CTC interprets the network output as a probability distribution over all possible character sequences that can happen, conditioned to an input sequence. From this probability distribution, it is possible to obtain a differentiable objective function that maximizes the probability of the correct output sequence from the input one, that can be trained by backpropagation in conjunction with the other layers of the network. Fig. 4.15 shows all possible decoding paths in a particular example simplified to a 2-character dictionary. For this particular case, the probability of the correct sequence "to" is: 0.072 (-to) + 0.432(t-o) + 0.288(tto) = 0.792.



Figure 4.15: CTC loss calculation: all possible paths to decode the output probabilities sequence are computed.

To better handle transitions between characters, CTC introduces a special character called "blank", usually represented by the symbol "-", which should not be confused with the whitespace character for hyphenation. This "-" character allows to properly model transitions between characters where there may be a certain separation, as well as possible spaces at the beginning or end of a handwritten text image. It also facilitates learning with images of the same text that may have different alignment and spacing between characters.

For the prediction of new images in the evaluation phase, the CTC selects the sequence or decoding path with the highest marginal probability given the input and applies a operator (B) that firstly removes duplicate characters and secondly removes the special character 'blank' (-). An example of decoding a handwritten word can be seen in Fig. 4.16.



Figure 4.16: CTC Decoding path for one word.

The white character can optionally appear between the other characters and is only required to appear between two identical characters that are repeated in the input image.

Thus, for example, the decoding of the following sequences is as follows

- B(-aa-b) = ab
- B(aa-bb) = ab
- B(aa-b-bb-) = abb

The first two examples illustrate a case of decoding the same text due to differences in the alignment of the characters in the input image.

The use of the white character is optional. In Bluche et al. [25], an analysis on the inclusion or not of the white character is provided. This analysis conclude that its inclusion favors the use of CTC in handwriting recognition.

There are different variants of the decoding process that allow to restrict the decodings to a prefixed dictionary or to introduce a language model [81]. A detailed efficiency analysis of the CTC applied to the handwritten text recognition problem, compared to the classical HMM-based approach, can be found in [25]. It also compares the performance of CTC with inputs from RNN and MLP, concluding that RNN provides better results because it can better handle dependencies in an arbitrary length context.

CTC in handwriting

The CTC model was introduced in 2006 by Graves et al in [80] as a general method for temporal classification with RNNs applied to the phonetic labeling problem. Later, in

2009, the same authors applied it to the handwritten word recognition problem in [81] where an architecture based on bidirectional LSTM (BLSTM) with a final CTC layer for decoding was proposed. This model is applied on a set of nine graphical features obtained from the normalized image by means of a one-pixel sliding window. In addition, the initial definition of CTC is modified to allow the integration of CTC with a dictionary and a language model in a new algorithm called *CTC token passing*. This allows to restrict and orient the decoding process and to obtain results directly comparable to those obtained in HMM-based architectures, where the language model, usually based on n-grams, supports the decoding process.

In the same year, Graves and Schmidhuber published a novel architecture for the handwritten recognition problem [79] that, starting directly from the image pixels, applied a Multidimensional LSTM (MDLSTM) architecture in several layers with a final CTC layer. To the best of our knowledge this is one of the first times that a unified probabilistic model is proposed for the unconstrained handwriting recognition problem. It trains character sequence recognition directly from the pixels of the text image. This MDL-STM+CTC architecture applied directly on the image and has become the reference for the handwriting problem between 2013 and 2017.

Although the first use of CTC in handwriting dates from 2009 in [81], it is not until 2014 that the CTC has become popular in the field of handwritten text recognition. Since them it is a standard used in the vast majority of HTR publications between 2014 and 2018 [26] [164] [25] [216] [20] [21] [40] [22] [170] [223] [46] [68].

Recently Scheid et al. in [181], have reviewed and compared different decoding strategies for the CTC algorithm. A new decoding algorithm called *word beam search decoding* was proposed. It allows the integration of a dictionary and a language model, like the *CTC token passing algorithm* proposed in [81], but reducing its computational complexity and allowing the decoding of special characters between words in the dictionary, such as numbers or punctuation marks.

In recent years, it has become increasingly common to decode the final sequence of characters using a decoder layer of the seq2seq architecture, which is also described in this chapter. In any case, CTC is still used sometimes with encoder-decoder architectures like seq2seq. For example, in Bluche et al. [21] the CTC is used as an auxiliary method to perform a pre-training at word or line level and these authors apply an architecture with attention and decoder to perform a prediction at paragraph level. Another use case of CTC in an encoder-decoder type architecture can be found in [53], where this architecture with hybrid attention for paragraph or page level handwriting recognition using CTC for the final decoding of the text, is proposed. Similarly, Yousef et al. [230] describe a neural network that can convert any fully convolutional handwritten line recognition model trained with the CTC into a handwritten page recognition model without needing for the image and its transcription to be segmented by lines.

4.7 Solutions based on Nearest Neighbors

Between 2014 and 2016 some authors have proposed an alternative to HMM or CTCbased approaches common at the time, that consists in representing the images and transcriptions of handwritten words in a shared n-dimensional embedding space and then applying a *k*-Nearest Neighbors (k-NN) algorithm to associate transcriptions and images. This approach was primarily aimed at solving the problem of word spotting, which consists of identifying whether a given word is present in a text image or not (usually for the purpose of indexing databases of historical manuscript text images). By extension, the application of k-NN to single word images is also a way of transcribing such images,

In 2014, Almazan et al. [2] address the problems of word spotting and word recognition with the above approach. They introduce a technique called *Pyramidal Histogram* of *Characters* (PHOC) to encode transcripts in a way that considers whether a specific character appears at a particular position in the text string. Then, the word images are projected into another space, where each dimension encodes whether a word image contains a given character in a particular region. Finally, the method trains a projection of the previous spaces into a lower dimensional common subspace where it uses k-NN for both word spotting and word recognition problems. Later, some other authors, like [128] and [169], employed a similar procedure, including the PHOC technique for transcription coding, but using attributes trained with a CNN for the generation of embedding associated to the word images. In reference [128], a proprietary architecture called *HWNet* is used, which is described in detail in [131], and in the case of reference [169], a VGG [192] style network is used.

Due to the search solution approach in a shared *n*-dimensional embedding space, these methods require a default lexicon to perform text recognition, which is a drawback because they cannot identify *out of vocabulary* (OOV) words.

4.8 seq2seq architectures

In recent years a new type of architectures applied to the offline HTR problem has appeared, the so-called *sequence-to-sequence (seq2seq)* architectures that follow the encoderdecoder framework [204], and that are oriented to model relationships between unaligned sequences, as it happens in the case of offline HTR.

This section introduces the seq2seq architectures and details their use in the HTR problem in recent years. These architectures are the ones used in Chapter 6 of this Thesis about deep architectures to continuous offline handwriting recognition.

4.8.1 Overview of seq2seq architectures

The seq2seq architectures were introduced in 2014 by Sutskever et al in [204], and Bandanau et al. [4] proposed a key modification of it by including an *attention mechanism* that allowed a good management of long sequences. These new kind of neural architectures are oriented to model sequence transduction problems, where it is necessary to decode an information sequence into another sequence not necessary aligned with the first one. The most common cases and these architectures are applied are: speech recognition [45] [5], machine translation [4] [204] and image captioning [226]. Their success in the above problems has also led to their use in other problems, such as text recognition in natural images [136] [190] or even speech recognition from lip movement in videos [47].

In Fig. 4.17 is outlined the different principal components of seq2seq architecture, composed of the following elements:

- An *encoder* that allows coding variable length input sequence in a fixed subspace. It uses to be composed by recurrent layers and, in the case that the input is a image or a video, it usually also includes convolutional layers.
- A *decoder* that, from the decoder output, builds the output sequence element by element. It is usually composed by a unidirectional RNN.
- An *attention mechanism*, that allows each decoder step to access to the most relevant encoder information for that step.



Figure 4.17: Basic schema of the Seq2seq architecture.

4.8.2 seq2seq Encoder

In seq2seq architecture the *encoder* is in charge of coding each input sequence, that could have variable length, into a fixed length sequence that codifies such input in a feature subspace. In that way, given a variable length input elements sequence $\mathbf{x} = (x_1, ..., x_{n_x})$ it is transformed in a fixed length sequence $\mathbf{h} = (h_1, ..., h_T)$ by the encoder, usually through a input sequence previous transformation that depends of its nature and, after that, by a RNN f in such a way that:

$$(s_1, ..., s_T) = g(x_1, ..., x_{n_x})$$

$$h_t = f(s_t, h_{t-1})$$
(4.9)

where g can be simply padding over the input sequence for length alignment, or a complex convolutional architecture in the case that one or several images (of one or more channels) being the input, or other options depending of the problem nature and the coding data. Additionally, $h_t \in \mathbb{R}^n$ is a hidden state at time t.

Finally, the output sequence $\mathbf{h} = (h_1, ..., h_T)$ is transformed in a context vector \mathbf{c} that will be sent to the decoder. For example, in the original seq2seq model by Sutskever et al. [204], the transformation is $\mathbf{c} = q(h_1, ..., h_T) = h_T$ and the context vector \mathbf{c} , that is the encoder sequence last element, it is moved to first decoder element. In the case that the model include an attention mechanism, it will allow to each decoder step, access to the complete encoder output sequence $\mathbf{h} = (h_1, ..., h_T)$.

The encoder architecture in a seq2seq model is very variable and dependent on the problem nature and on the input data. In fact, most of the variability in the different approaches

that use seq2seq models for HTR problem is in this component and, in occasions, in the attention component used.

Although the encoder architecture can be very diverse, the last layer is usually RNN type, in particular those kind that can capture well dependent relationships in long sequences, like LSTM or GRU. The decoder usually uses the same kind of RNN network with the same cell number in these cases, because it is usual that the final state of the encoder \mathbf{c} will be the initial state of the decoder. If a bidirectional RNN is used as encoder, it is possible to select as decoder initialization the final state of the forward layer or the same of the backward layer or a combination of both.

Quite complex cases can be codified by the encoder with multiple input sequences of different length. For example, in reference [47] the aim is to recognize an audio sequence that came along with a video sequence with the speaker's facial movements to complement the audio signal.

In handwriting recognition problem, where a 1-channel grayscale image is usual, the encoder typical architecture is composed by convolutional and pooling layers followed by recurrent layers. The convolutional and pooling layer sequence is oriented to extract the image visual characteristics of the handwritten text. The final recurrent layer sequence learns the characteristics of the sequential nature of the text. It is the same underlying idea as in CNN-RNN-CTC model described in Subsection 4.5.6. Different encoder architectures used in HTR literature will be described in detail in Subsection 4.8.5. In Chapter 6 multiple experiments are detailed with different alternatives related to the encoder module of a seq2seq model applied to HTR in the main reference databases.

4.8.3 seq2seq Decoder

In the seq2seq architecture, the *decoder* is in charge of generating the output sequence of the model, which is usually of different length than the input sequence. Unlike the CTC case, the output sequence may be longer than the input sequence.

The decoder predicts each of the elements of the output sequence y_t as a function of the previous elements in the sequence $y = (y_1, y_{t-1})$ and of a context c provided by the attention mechanism. Thereby, the decoder defines a probability on the output sequence y as the product of the ordered conditional probabilities of each previous element of the sequence, as given in the equation:

$$p(y) = \prod_{t=1}^{T} p(y_t | \{y_1, ..., y_{t-1}\}, c)$$
(4.10)

The decoder is usually implemented as a unidirectional RNN [201] [113], generally speaking of LSTM or GRU types. The RNN output is additionally transformed by a dense layer d to fit its dimension to the cardinal of possible set of values of y_t . It is finally applied an activation softmax function [35] to get the probability distribution estimation $p(y_t)$.

The initial state of the RNN decoder h_0 is normally initialized with the value of the final state of the encoder h_{last}^e . This is shown in Fig. 4.17 by the direct connection defined between the encoder and the decoder. Although it can also be randomly initialized [113].

The decoder RNN input during training phase is the target sequence of characters \mathbf{y} but initiated with the special element $\langle GO \rangle$. The output sequence finalization is indicated when the special element $\langle END \rangle$ is decoded, like is illustrated in the left of Fig. 4.18. This figure includes a decoder detail with all previous elements. As it can be observed, in the training stage the input is compound by the original values of the output sequence \mathbf{y} (left sub-figure in Fig. 4.18). In the evaluation stage the input at step t is the step t-1 output estimation, denoted as \hat{y}_{t-1} .



Figure 4.18: Schema of a LSTM based decoder: (a) train(teacher forcing) and (b) evaluation.

The training of the seq2seq model by passing as input to the decoder the original values of the output sequence, but shifted to t - 1 is called *teacher forcing* [224]. This training strategy can lead to problems in the evaluation phase because an error in one of the generation steps of the output sequence is propagated through the conditioned probabilities. This causes a decrease in performance because the predicted sequence diverges from the real one from the point of error.

Teacher forcing also has a negative effect: the train overfit. This is because at each step t of the decoding process the previous real value of the character sequence y_{t-1} is used. To mitigate this, several alternatives have been proposed such as the approach of combining the actual value and the estimate of the previous step as input to the next step [8]. It is also common to train with the same criterion used in the evaluation step, i.e., to use as input to step t the estimate generated in the previous step \hat{y}_{t-1} instead of the actual value y_{t-1} .

The values y_t , usually discrete, have to be coded so they can be used in the decoder input. They can be transformed by one-hot codification or a dense layer can be used that codifies the values of y_t in low dimensionality embeddings.

In addition, Fig. 4.18 also shows that, in case that an attention mechanism is included in the seq2seq model (in the t estimation step) the attention vector c_t is introduced into the decoder by concatenating it with the RNN input. This concatenation is done directly with the previous element of the output sequence y_{t-1} in the training step with the teacher forcing strategy. In the evaluation step, it is done with the estimation obtained in the previous step \hat{y}_{t-1} (see sub-figure right in Fig. 4.18).

The following equations describe the detailed calculation carried out by the RNN decoder:

$$\hat{y}_{t} = argmax(p(y_{t}))$$

$$p(y_{t}) = softmax(dense(h_{t}))$$

$$h_{t} = LSTM(concat(e_{t}, a_{t}), h_{t-1})$$

$$e_{t} = embbeding(y_{t-1})$$

$$h_{0} = h^{e}_{last}$$

$$e_{0} = embedding(< GO >)$$

$$(4.11)$$

The estimated output sequence $\hat{\mathbf{y}} = (y_1, ..., y_k)$ can be easily built by selecting the maximum probability values in $p(y_t)$ by an *argmax* function for each t, as it is indicated by previous equations. But, it is also possible to use more complex decoding strategies like bean search decoding or even addressed to a language model [44].

There are several variations on the above formulation. For example, one can try to obtain a more robust decoder that could explore another decoding sequences by selecting a random element based in *softmax* probability distribution instead of selecting the higher probability element. This general technique is called *multinomial decoding* [43].

Another common strategy to improve the stability in multi-category predictive models is the *label smoothing* [206], which consists of replacing the identificators 0, 1 of the selected category in each step of the decoder by a regularized version $\frac{\epsilon}{k}$ and $1 - \epsilon \frac{k-1}{k}$. This procedure usually improves the generalization capacity of the model, but spending longer training periods. In the case of reference [113] this technique has a significant positive impact when the model seq2seq is applied to the HTR problem.

Finally, although the decoder is usually an unidirectional RNN, sometimes it uses another different architecture. For example, in [112] a combination of RNN and de-convolutions is used.

4.8.4 seq2seq Attention Mechanism

This component, introduced by Bandanau et al. [4], confronts the original seq2seq problem as it connected the encoder and the decoder only with the state vector c, limiting the use of seq2seq models with long longitude sequences because the performance of the original model deteriorated rapidly when the length of the sequences increased [4].

The attention mechanism solves the previous problem by connecting each step of the decoder with all the elements of the encoder sequence that may be relevant for the decoding of that step. The model then predicts each element of the output sequence based on the predictions of the previous elements and this context provided by the attention mechanism.

More formally, and following the notation of the previous sections, to calculate the attention vector c_t at each output time t over a encoder output length N_e :

$$e_{t,i} = w^T \tanh(Wh_i^e + Vh_{t-1}^d + b)$$

$$a_t = softmax(e_t)$$

$$c_t = \sum_{i=1}^{N_e} a_{t,i}h_i$$
(4.12)

where w and b are trainable weight and bias vectors, W and V are trainable matrices, h_i^e is the hidden state of the encoder at time $i \in [0, 1, ..., N_e]$ and h_{t-1}^d is the hidden state of the decoder at time t-1. The attention mechanism is implemented as a neural network layer that combines the states of the encoder and the previous states of the decoder with the hyperbolic tangent as activation function. The vector e_t contains in the position i a score of how much attention need to put the current decoder step t on the i-th encoder hidden state h_i . Then, we apply the softmax function to normalize it and create the attention masks a_t over all the encoder states. The final attention vector c_t are concatenated with the input of the decoder layer as we show in Fig. 4.18.

Location Based Attention

The previous attention mechanism, also called *content-based attention*, is one of the most common and is the one used in this Thesis. This algorithm has been seen to have scalability problems with very long input sequences [45] (e.g. of the order of thousands, as for example those that had the problem of speech recognition). This is motivated by the fact that, in this type of sequences, the previous model is not capable of distinguishing multiple representations of the same input fragment in different positions. In this Thesis, where we work at word images level, the length of the maximum input sequence corresponds to the width of the word image and it does not exceed a few hundred pixels, and the attention mechanism does not seem to face the those problems. However, in a line, or even paragraph or page level of the handwritten recognition approach, it can be relevant.

The modification of the previous attention mechanism which has been proposed to solve this problem [45] is called *location-based attention*. It is based on the fact that the previous attention mechanism explicitly takes into account, both the identification of the position considering the previous step, and the characteristics of the input sequence. This is achieved by adding auxiliary convolutional characteristics as inputs to the attention mechanism l_t . They are extracted by convolving the attention weights from the previous step a_{t-1} with trained filters F. In this way, the calculation of $e_{t,i}$ is replaced in the equation 4.12 by:

$$l_{t} = F * a_{t-1} e_{t,i} = w^{T} \tanh(Wh_{i}^{e} + Vh_{t-1}^{d} + Ul_{t,i} + b)$$
(4.13)

Attention smoothing

Another problem that has the classical attention mechanism detailed in the equations 4.12 is that the included *softmax normalization*, calculated as:

$$a_{t,j} = \frac{exp(e_{t,j})}{\sum_{i=1}^{L} exp(e_{t,i})}$$
(4.14)

is likely to generate noisy information when the input sequence h is long. This makes it difficult for the calculation equation of attention to focus on the relevant input elements h_i for the decoding of each of the elements of the output sequence y_j . To solve this, Chorowski et al. [45] proposed to replace the normalization with softmax, with another one using the bounded logistic sigmoid σ such that:

$$a_{t,j} = \frac{\sigma(e_{t,j})}{\sum_{i=1}^{L} \sigma(e_{t,i})}$$
(4.15)

which softens the focus found by the attention mechanism.

This smoothing technique has been used in the case of HTR (for example, in [113]) with worse results than those obtained with the normalization. In this case, it has worked at the word level, so the length of the input sequences is relatively short and smoothing does not seem to be an advantage, but at line level it could be useful.

Transformers

In recent years, the attention mechanisms have gained great relevance, since they are the basis of the networks called *transformers* [213]. These have produced important advances in modeling both textual information [62] as well as images [38]. The transformers are an encoder-decoder type architecture similar to the seq2seq architecture with the difference that the encoder and the decoder do not use CNN or RNN but are basically designed on attention mechanisms. They were first introduced by Vaswani et al. [213] in 2017, and they were applied to machine translation and English constituency parsing.

Transformers have two advantages over RNN-based encoder-decoder models. The first is that their execution is highly parallelizable, and even though they generally have a much higher volume of parameters, their computational cost and training times are lower than using RNN architectures. The second is that they handle long-range dependencies better, since the length of the paths that forward or backward signals must travel to cross the network are constant in a transformer. On the other hand, in an RNN-based seq2seq architecture, the length of paths are of the order of O(n), where n is the length of the input sequence. The shorter these paths, the easier it is to learn long-range dependencies [125].

4.8.5 seq2seq in handwriting

From the success of the seq2seq models in other sequence transduction problems, its application to the offline HTR problem became popular. Additionally, these models tried to overcome certain limitations inherent to CTC-based architectures, such as the fact that the alignment between the input and the output had to be strictly monotonous and that the maximum length of the output sequence was conditioned to the length of the input sequence.

An additional advantage of seq2seq architecture is that the decoder module can potentially learn a language model from input character sequences. This model is limited because of the small size of the available handwriting text databases compared to the volume of text typically used in the construction of general-purpose language models (i.e. usually larger).

The first reference to such seq2seq architecture applied to HTR is probably Bluche et al. [21] in 2016. The model is applied to full paragraph recognition, using a line level pre-trained MDLSTM-CTC auxiliary model as a base for the encoder.

Later on, in early 2018, Sueiras et al. [201] proposes a standard seq2seq architecture to the problem of recognition handwritten words images. This architecture is analyzed in depth in Chapter 6.

In the paper by Kang et al. [113], the seq2seq architecture is also proposed with a CNN-GRU type encoder, an attention mechanism and a GRU decoder. The GRU components of the encoder and decoder are configured as 2 layers of size 512 after comparative analysis. As CNN component, a VGG-19-BN architecture [192] started with pre-trained weights from ImageNet is selected. Additionally, it is experimented with both content-based and location-based attentions, as well as with attention smoothing techniques, multi-nominal decoding [43] and label smoothing [206]. In conclusion, the selection of a location attention with the application of the label smoothing technique provided the best results. This same conclusion is obtained by the same authors in [116].

Most of the works that apply seq2seq to the HTR problem perform the decoding with the standard greedy decoding algorithm that decodes in each step the class with the highest probability provided by the softmax transformation. As an improvement, Chowdhury and Vig [46] propose an alternative decoding algorithm called *beam search* that maximizes the joint probability of the entire output sequence. This is the same algorithm as the one proposed in [181] for the CTC. According to the experiments published in [46], by applying the beam search algorithm to the decoding provides significant performance improvements to the seq2seq model of more than 20 % in CER and WER on the same model using a greedy decoding.

Another contribution to the decoding process is provided by Kang et al. [116], who proposed a new way of integrating a language model in the decoder. This technique, called *candidate fusion*. It is an alternative to other language model integration methods such as *shallow fusion* and *deep fusion* [87].

In reference [46], two additional improvements are proposed over the standard model. The first one is the use of an alternative loss function to the standard cross entropy (CE) called *focal-loss* [140] which solves the problems that the CE loss has with unbalanced samples (where the CE makes that the most frequent and easily classifiable cases tend to dominate the gradients). The second one is the application of batch normalization [105] in convolutional layers, which has also been used by other authors [113]. In both cases, the modifications represent a significant improvement of the WER and the CER in the experiments carried out with the IAM and RIMES databases.

An extensive analysis of different alternatives for the attention model in the application of the seq2seq to HTR can be found in Michael et al. [150]. In particular, the following variants were reviewed: *penalized attention* [178] that penalizes feature vectors that have received high attention scores in the past; *monotonic attention* [173] that processes the attention in a left-to-right manner: *Monotonic Chunkwise Attention* (MoChA) [42], which relaxes the hard monotonicity condition that splits the input sequence into chunks within which a content based attention is applied. The paper concludes that monotonous and chunkwise attention mechanisms tend to provide better results.

Finally, in 2020, Kang et al. [115] proposed for the HTR problem an encoder-decoder type model based on transformers. As indicated in the previous section, the transformers emerge as an alternative to RNN-based seq2seq models to speed up the processing and better handle long dependencies on the input sequences. These models have obtained very good results in several areas, and for the HTR, they also achieve superior results

than those obtained with seq2seq architectures.

The *transformer-type* models are currently a very active research area and, in our opinion, it is one of the most promising work lines of in HTR [115]. Another very promising line is the one described in the next section, which it is based on the use of synthetic data and transfer domain strategies. This is intended to compensate the problem caused by the limited size of existing annotated handwritten text image databases.

4.9 Synthetic data and domain adaptation

The use in HTR of increasingly complex and layered neural architectures is conditioned by the limited volume of existing annotated data. The most extensive database available in English is the IAM database [146] with only 13,353 lines annotated. This limits the complexity of the models that can be built, which quickly begin to overfit.

There are several known strategies to alleviate these limitations. The most common is data augmentation (discussed in the Section 4.2). Briefly, this technique consists of constructing distorted versions of the input images that keep the strict invariant text in them, for example scale changes, rotations, slant or skew transformations or elastic distortions. The technique is simple to implement and quite effective, so it is very popular. The new transformed images are usually built dynamically in each epoch of the training which generates a theoretically infinite sample. This technique has the disadvantage that it cannot generate new texts, only modify existing ones in the train database.

A second more recent strategy to compensate for the scarcity of annotated data is the use of synthetically generated images, concretely using computer fonts that imitate the handwriting style. In 2014, in the paper by Jaderberg et al. [106], the use of standard typographic fonts is proposed to build a database of synthetic images used in the natural scene text recognition problem. This approach provides very good results in terms of error reduction. Inspired by the above previous papers, Krishnan and Jawahar [130] proposed the use of typographic fonts that emulate the handwritten text to generate synthetic images that can be used in the HTR problem. To do this, they generated a database of 90,000 word images from 750 handwritten-style fonts with a limited character set. This database has been used in paper [68] to pre-train a CNN-LSTM-CTC architecture with the result of a relative improvement of more than 10 % in CER and WER by introducing pre-training with synthetic data over a not pre-trained baseline of the same characteristics. Fig. 4.19 provide an example of text written with a typographic font that emulate the handwritten text.

Example of text created with a computer font based on the author's wife handwriting.

Figure 4.19: Sample of text written with a typographic font that emulate the handwritten text.

The same pre-training strategy with synthetic data generated from freely available electronic sources that imitate cursive handwriting has been followed in [115] at line level, and in [116] at word level. In the case of [115] relative improvements close to 40% are

cited in CER and WER at the line level for the IAM database produced by pre-training with synthetic data.

The use of synthetic images generated from handwritten fonts provides improvements in the cases where it has been used [238] [115]. However, the resulting images have not the enormous variability that the handwritten text has. It also has an artificial regularity in the spatial distribution of the characters that makes a significant domain difference to the actual handwritten text. This generates a shift domain problem, whose solution is based on methodologies to adapt a model trained in an original domain (in this case of synthetic words) to a new domain such as a database of real handwritten text.

The *shift domain problem* is well studied in images [75], although usually the proposed solutions are focused on classification problems. Most use convolutional architectures to project the source and the target domains in a shared space in which the two domains are aligned. The function to be optimized is generally a combination of the error of the original problem and the error in the classification of each sample in its correct domain. This is relatively simple to implement for the case of a multinomial target classification problem, but it cannot be directly applicable to the HTR problem of transcribing an image into a sequence of multiple characters of variable length (precisely due to this characteristic variable dimension of the target).

To solve the above, Zhang et al. [238] proposed a seq2seq type architecture that included a module called *Gated Attention Similarity*. This module allows aligning the distributions of the source and target domains in a common feature space. On this space, a loss function is defined at the character level that is used to make the model invariant to the domain transformations.

Another work in the same line is that of Kang et al. [114], where a seq2seq type architecture is also proposed to perform domain adaptation in HTR. To do this, it incorporates a discriminator module of the input image domain into the model. This module includes a temporary pooling step that allows handle sequences of variable length. This paper shows the efficacy of the proposed model to the problem of unsupervised writer adaptation, training the recognition model exclusively with synthetic images generated from typographic sources and applying the domain adaptation module on the destination domain of unlabeled handwriting databases. In this way, the efficiency of the algorithm is shown in cases in which there is no annotated information in the destination domain.

4.10 Language models background

As described in the Section 2.4, the *language models* (LM) allow to introduce the context into the decoding process. The most commonly used types of LM in the area of handwriting recognition are the statistical *n*-gram language models [145] [14] [64].

An *n*-gram language model is a probabilistic model that predicts the next word w_n in a sequence as a function of the previous $w_0, ..., w_{n-1}$ words regardless of the word position in the sequence. These models are generally built from a large corpus that allows sufficient repetition of the most common word sequences in a language. In addition, they include a smoothing technique to handle sequences of words not present in the training corpus, the most common is the *Kneser-Ney* technique [123]. The details of how these models are built are given in the Section 2.4.

These LM models played an important role in the HMM-based approaches because it was the usual way to obtain the a priori probability value of each word to decode P(w) required in the decoding process with HMM. The type of *n*-gram LMs used is varied, for example in the case of [14] a 2-gram statistical language model is used and, in [126] and [26], both 3-gram and 4-gram LMs are employed. The corpora used for training the *n*-gram LMs are fairly standard and dependent on the handwriting database used.

The *n*-gram LMs implemented by the different authors are at word level. Therefore, they only decode those words that are in the LM training corpus. In order to recognize also those words that are out of vocabulary (OOV), some authors like [126] add a second LM at character level that complements the word level LM. Also, in some cases, LMs other than the *n*-gram type have been used. In particular, in [235] a language model based on neural networks instead of an *n*-gram based LM is proposed.

In the approaches based on CTC, LMs are not strictly necessary. Therefore, many authors do not use them directly, providing direct results from the visual model or those obtained after performing a search into lexicons. This eliminates one of the sources of variability when comparing results from different publications, using different LMs by each author. In any case, LMs can be included as part of CTC-based models; for example, a modification of the CTC algorithm called *CTC token passing*, which includes a bi-gram LM in the decoding process, is proposed in [81].

In the case of approaches to the HTR problem based on the seq2seq architecture, LM is generally not used with some exceptions. For example, Kang et al. propose in [116] a method called *candidate fusion* to introduce an LM in an integrated way in the decoder of seq2seq type models. The LM integration results in a WER improvement of 14% for RIMES database, however for IAM database, the improvement is less than 1%. Subsequently, in [115] the same authors propose a model for continuous HTR based on transformers and point out that the results with or without LM are almost the same, concluding that the transformer already learns a particular type of LM.

Chapter 5

Deep architectures for isolated offline character handwriting recognition

This chapter describes the models and experiments performed on the problem of recognizing isolated handwritten characters present in images. The main objective is to analyze the performance of different convolutional architectures in the simplest version of the HTR problem: recognizing isolated characters. The aim is to identify the more efficient character recognition models, general enough to be used as part of a word recognition system without segmenting the words into their component characters. The results of this chapter will be used in the next chapter, where a word recognition model is proposed, and the character-level models developed here are applied over a sliding window that runs along with the words to provide evidence of the occurrence of each character sequentially.

The first section introduces the handwritten character recognition problem. Then, in Section 5.2 the creation of a new handwritten character database named COUT is detailed. For this, firstly, in Subsection 5.2.1 we describe the process of the database creation. Next, in Subsection 5.2.2 the curation algorithm proposed to emulate a continuous handwriting context is described. Once the new COUT database has been introduced, the following sections describe the proposed models and the experimental results. Section 5.3 introduces the three convolutional model architectures with which the experiments will be performed. In Section 5.4, all the experimental setup and achieved results are detailed. This section is divided into the following subsections: in Subsection 5.4.1 the baseline results on the MNIST reference database are provided; then, in 5.4.2, for NIST and 5.4.3 for TICH datasets, respectively results are provided to make a comparison of the proposed architectures with the results of other authors; finally, in Subsection 5.4.4 the detailed results over the new COUT database are presented. This chapter concludes with Section 5.5 in which all the obtained results are discussed.

5.1 Introduction

The main goal of this Thesis is to provide a general solution to the continuous offline HTR problem, particularly the word recognition problem. For this aim, we propose an architecture with a first component based on convolutional networks, which performs a first step of extraction of the most relevant characteristics of the text images related to the characters which they contain. This first step is considered essential to obtain a solution

to the continuous problem. To identify the most suitable convolutional architecture, an analysis of different types of network models for the most basic problem of recognizing isolated characters is carried out.

Building a character-based recognition model that is effective for characters included in a word without segmenting it is complicated since most handwritten character databases contain only isolated characters [61] [90]. Additionally, existing character databases are limited to upper and lower case characters and digits. They do not include punctuation marks or special characters such as parentheses or brackets present in general handwritten text databases at the word or line level. These isolated character image databases have also been obtained from templates in which the characters have been written isolately. These characters may contain different strokes or stroke shapes than appear when they are written as part of a word. In these cases, ligatures are added, and the characters appear surrounded by artifacts from the strokes of the preceding and the following characters.

To avoid this, we have created a new offline handwritten character database using the UNIPEN online database [90] that includes all visible ASCII characters, from ASCII 33 (corresponding to the symbol '!') to ASCII 126 (corresponding to the symbol '~'). In addition, an algorithm has been defined to include, in a synthetic way, artifacts for each character image. We aim to approximate the effect of the character shape when it is placed inside a word, basically by moving and truncating it. We also have added some artifacts to the left and right of each character. We named this new database as COUT (Characters Offline from Unipen Trajectories).

In order to build our model, we tested new architectures based on CNNs due to the recent impressive results produced by these networks in image classification problems, both in general [132] and also in handwriting digit recognition [61], and character classification problems [237] [232].

For the experiments, we have selected three architectures that have shown good results in the handwritten character recognition problem in the past [61] [49], and that have been proposed as the first convolutional component in general HTR models [170] [113]. These architectures are LeNet [225], VGG [192] and ResNet [91], respectively. These have also been selected because they have an increasing level of complexity in terms of layers and number of parameters that allow a varied range of experiments to be conducted.

Different handwritten character databases were used for the problem of isolated character handwriting recognition by several authors. For example, Deng [61] uses the MNIST database, Ciresan and collaborators [49] built ensembles of CNNs over the NIST database, and Van der Maaten [143] used neural networks from 2 to 5 layers over the TICH database. In order to evaluate the proposed architectures against the results of other authors, experiments have been performed on the same public databases MNIST, TICH, and NIST, respectively.

Finally, systematic experiments have also been performed on the proposed COUT database, including models on different character sets. Extensive error analysis has also been carried out to understand well the limitations of the selected models.

5.2 COUT character database generation

A new character recognition model is built, and its generalization capacity using different character sets is validated. Our objective is to use it to develop a new approach to handwritten word identification that avoids the character segmentation problem. To develop it, we need a handwriting character database where images of characters resemble as closest to the corresponding images obtained with a sliding window over a word image (and that is why the current databases are not helpful at all). Thus, we decided to create a new offline handwritten character database to overcome these limitations.

For the construction of this new image database of handwritten characters, we used UNIPEN online handwriting database. This online database contains the information of the (x, y) coordinates of the handwritten text points defining the strokes captured with a digitizing device.

Several reasons to create this new database are enumerated next. First, the UNIPEN database includes a character set extended to symbols and punctuation marks that are not included in other character databases but do appear in continuous handwritten text databases. Second, the UNIPEN database is an online handwritten database of words, lines, and paragraphs segmented at character level. It means that the characters were written as a part of a word. In other character database, we can control the resolution and the stroke size to get character images as closest as possible to the word-level databases.

In this section, we describe the steps followed to create the new dataset of handwritten characters. It is specially oriented to test model architectures that work with characters but also as a component of a more general system to automatically recognize handwritten words.

The COUT database was built in two steps:

- First, we extracted isolated characters from the UNIPEN online handwritten database.
- Next, we perform several transformations on the initial character images to simulate how these characters can appear inside real words.

The UML activity diagram shown in Fig. 5.1 illustrates the stages involved for producing the new character database. The next two subsections describe in detail these stages.

5.2.1 Extraction of characters from UNIPEN online

Using this online database to extract the "base" characters, we avoid the character segmentation problem in offline handwritten text, and we also obtain perfect isolated characters. The isolated annotated characters are first identified. Next, these characters are hand-retouched so that the x and y coordinates of pixels are connected in each stroke which shapes the character. Variable-thickness strokes are used depending on the original resolution of the characters to ensure that all final characters have a similar thickness.

The images from each available category are generated: upper case, lower case, digits, and punctuation marks, respectively. These generated images are resized to 64×64 pixels without changing their aspect ratio or relative size in paragraphs. Finally, the generated images



Figure 5.1: UML activity diagram of creation and transformation process of the character database. Processes are presented as white boxes and outputs as gray boxes.

are curated manually, one by one, to make sure that they are assigned to their correct category and that are human-legible. Our database contains 93 categories and 68,382 image characters, where the average number of samples per category is 735. Table 5.1 illustrates the types of characters present in our dataset. Our database of curated characters can be downloaded from https://github.com/sueiras/handwritting_characters_database to facilitate the reproducibility of the experiments. We also provide here a Python code that implements the transformations described in the next subsection.

!	"	#	\$	%	&	,	()	*	+	,	-		/	0	1	2	3	4
5	6	7	8	9	:	;	<	=	>	?	0	А	В	\mathbf{C}	D	Е	F	G	Η
Ι	J	Κ	L	Μ	Ν	Ο	Р	\mathbf{Q}	R	\mathbf{S}	Т	U	V	W	Х	Υ	Ζ	[]
^	_	4	a	b	с	d	e	f	g	\mathbf{h}	i	j	k	l	m	n	0	р	q
r	\mathbf{S}	\mathbf{t}	u	v	W	х	у	\mathbf{Z}	{		}	~							

Table 5.1: Characters contained in the COUT database.

A detailed description of this new database, including sample images and character frequency, was included in Chapter 3.

5.2.2 Transformations applied to original images

A set of transformations were applied to the original character images with two goals: augmenting the training set size in experiments and ensuring good results with the models when they are applied on parts of word images (i.e., at the word level where the characters are not isolated). Fig. 5.2 illustrates some examples of transformations applied to Unipen characters to build the training sample.

The applied transformations are the following ones. First, we resize characters and translate them up or down. Specifically, we reduce the characters by 25% of their original size, and the new image is moved up or down for each character. The main idea is that some

CHAPTER 5. DEEP ARCHITECTURES FOR ISOLATED OFFLINE CHARACTER HANDWRITING RECOGNITION



Figure 5.2: Example of transformations chain applied to build a training sample.

characters only can appear up or centered in a word, while others can only appear down and centered, and finally, other ones can appear in any place. This location depends on whether the characters usually cover the ascenders or descenders areas. In this way, the relative vertical positions of the characters match with those when extracted from a continuous line of text. The characters included in each case are:

- The letters: 'a', 'c', 'e', 'i', 'm', 'n', 'o', 'r', 's', 'u', 'v', 'w', 'x', 'z' are placed up centered and down because they do not cover the ascenders or descenders areas.
- The letters: 'g', 'j', 'p', 'q', 'y' are placed centered and down because they cover the descenders area.
- Finally the letters: 'b', 'd', 'f', 'h', 'k', 'l', 't', 'A', 'B', 'C', 'D', 'E', 'F', 'G', 'H', 'I', 'J', 'K', 'L', 'M', 'N', 'O', 'P', 'Q', 'R', 'S', 'T', 'U', 'V', 'W', 'X', 'Y', 'Z' are placed up and centered because they cover the ascenders area.

Second, we move left or right the letter image and include in its sides (left, right or both) the beginning or end of another letter. In this way, similar training examples will be available to those that will be found when applied the model to a fragment of a word image. The borders of the characters (with 3 pixels of thickness) are extracted from the available sample to achieve this. These borders are added to the extremes of the existing characters in order to generate a new sample. In this process, the non-overlapping between the characters and the added borders is guaranteed.

5.3 Convolutional architectures

Convolutional Neural Networks (CNN) have been used for many years for the offline handwritten character recognition problem [135]. In our experiments, we apply three convolutional architectures with incremental complexity to identify those that may be the most suitable ones as the first component of the word recognition models described in Chapter 6.

- The first one is *LeNet* [225]. It was the first proposed for handwritten digit recognition in 1995. It is characterized by using 5×5 kernels, and having two sets of convolution and pooling layers, with an incremental number of feature maps.
- The second one is VGG [192], which is characterized by proposing a convolutionconvolution-pooling layers with 3×3 kernels, and by duplicating the number of feature maps in each module.

• The third one is the *ResNet* architecture [91], which is characterized by proposing direct connections (residuals) between the input and output of each convolutional component. Additionally, it employs the regularization strategy called *batch normalization*[105].

In every case, we use zero-padding to preserve the spatial size and a stride of ones in all the convolutional layers. All hidden layers include the *non-linear rectification units* (ReLU) [132]. Finally, we use the dropout regularization [95] in the dense layers with a ratio of 0.5. The configuration of each of these architectures used in the handwritten character recognition experiments is described below.

5.3.1 LeNet architecture

The LeNet architecture was originally proposed in 1995 by Yann LeCun and collaborators [225]. This model alternates convolutional layers and pooling layers, and ends with dense layers. The strategy of alternating convolutional layers and downsampling layers was initially proposed by Fukushima in 1975 in [73], and in [225] is adapted to the problem of recognizing isolated handwritten digits.

The original LeNet architecture was composed of four layers organized in two groups of a convolutional layer followed by a pooling layer. Then, in some cases, it was completed with dense layers at the end. Since this architecture was proposed precisely for the handwritten digit recognition problem, we decided not to modify it substantially for the present experiments. It has been taken into account that the number of categories in the current general character recognition problem is much larger. Additionally, the computational limitations of the computers at that time were conditioned by the total number of network parameters that could be handled. Therefore, it has been decided for the current experiments to increase the dense layers dimension and the number of feature maps. Fig. 5.3 shows the LeNet architecture selected in the present character recognition experiments. The architecture is characterized by employing two groups of convolutional layers with kernels of size 5×5 and pooling layers with kernels of size 2×2 that reduce the size of the output feature maps by half in each case. The number of feature maps in the first group is set to 20, and in the second group is set to 50. Three dense layers are added at the end. The first two ones are oriented to identify interactions between the output elements of the feature maps. They are sized so that the second is half the first and uses a ReLU type activation. The third one is a classification layer with an output size equal to the number of categories of the target, including a *softmax* activation that allows the network to return as output a probability distribution of the possible targets.

For the different databases used, the size of the dense layers have been modified to adapt them to the different number of categories in the target set. The simplest case is the MNIST database which has only 10 categories, since it is a digit database. For this case, we have set the size of the two dense layers to 256 and 128, respectively. The rest of the databases already have 26 or more categories as they include the letters of the alphabet in upper case (TICH) or upper and lower case (NIST and UNIPEN). In this case, the two dense layers have been configured with sizes of 512 and 256, respectively. Table 5.2 includes a detail of the configurations used of this architecture for the different databases and character sets used in the experiments and compares them with the original LeNet-1 and LeNet-5 architectures.


Figure 5.3: Proposed LeNet architecture schema for the MNIST database.

Database	Charset size	Model	Feature maps 1	Feature maps 2	Dense layer 1	Dense layer 2
MNIST	10	LeNet-1 [225]	4	12	0	0
MNIST	10	LeNet-5 [225]	6	16	120	84
MNIST	10	ours	20	50	256	128
TICH-NIST-COUT	+28	ours	20	50	512	256

Table 5.2: Configurations of the LeNet architecture used in the different databases.

Table 5.3 includes a detailed information of the layers for the MNIST database with an input image size of 28×28 and 10 categories. The layer type, the output tensor dimension, and the number of trainable parameters of each layer are indicated. The input tensor size of each layer is the one indicated as output in the previous layer. An input layer has been incorporated in the table indicating the size of model input data to obtain this information in all the layers. The total number of parameters is 687,142 for this case.

Layer type	Output Shape	Parameters
input	(28, 28, 1)	0
Convolution2D(ReLU) MaxPooling2D	(28, 28, 20) (14, 14, 20)	500 0
Convolution2D(ReLU) MaxPooling2D	$(14, 14, 50) \\ (7, 7, 50)$	$25,000 \\ 0$
Flatten Dense(ReLU) Dense(ReLU) Dense(Softmax)	$(2,450) \\ (256) \\ (128) \\ (10)$	$0\\627,456\\32,896\\1,290$

Table 5.3: Summary of layers, output shape and number of parameters for the proposed LeNet architecture applied over MNIST database.

5.3.2 VGG architecture

This second architecture is based in the VGG network [192] that is characterized by proposing a convolution-convolution-pooling layers module with 3×3 kernels and by duplicating the feature map number in each module. Thus, the main difference with the previous LeNet architecture is that the convolutional and pooling layer blocks have the following configuration:

- A first convolutional layer with a kernel of size 3×3 .
- A second convolutional layer of kernel 3×3 .
- A third pooling layer with a kernel of size 2×2 that reduces the output dimension to half the input dimension.

The original VGG architecture has between 16 and 19 layers and it is oriented to the more complex task of the ImageNet challenge [109] that is to classify a color image into 1,000 general categories. Our problem is more limited, and we do not need an architecture so complex. The proposed model has 9 trainable layers, grouped in 3 stacks of convolutional and subsampling layers, and 3 final dense layers similar to those proposed in the previous LeNet-type architecture. The detailed architecture is shown in Fig. 5.4.



Figure 5.4: VGG architecture model with the sizes of the layers and the sizes of the convolutional masks (in red). This model is instantiated for an input size of 64×64 and a final dense layer of 52 categories corresponding to the model applied on upper and lower case letters of the COUT database.

The options of 2, 3 and 4 convolutional blocks have been evaluated using the subset of images of upper and lower case letters of the new COUT database to determine the number of convolutional blocks with which to perform the experiments. Table 5.4 summarizes the results of accuracy in the validation and test partitions of the experiments performed, from which the configuration with 3 convolutional blocks is selected as the one that obtains the highest accuracy in the tests.

Finally, the Table 5.5 includes some detail of the layers for the case of the subset of upper and lower case letters of the COUT database with an input image size of 64×64 and 52 classes. The layer type, the output tensor size and the number of trainable parameters of

Num. VGG blocks	Accuracy	
	Validation(%)	Test(%)
2	84.77	85.62
3	86.83	87.41
4	87.12	87.20

Table 5.4: Accuracy of the VGG architecture over the COUT upper and lower case database. Comparative over the number of blocks.

each layer are detailed. The input tensor size of each layer is the one indicated as output in the previous layer. An input layer has been incorporated to the table indicating the size of the model input data to obtain this information in all layers. The total number of parameters is 4,625,492 for this case.

Layer type	Output Shape	Parameters
input	(64, 64, 1)	0
Convolution2D(ReLU) Convolution2D(ReLU) MaxPooling2D	$\begin{array}{c} (64,64,32)\\ (64,64,32)\\ (32,32,32) \end{array}$	$288 \\ 9,216 \\ 0$
Convolution2D(ReLU) Convolution2D(ReLU) MaxPooling2D	$\begin{array}{c} (32,32,64) \\ (32,32,64) \\ (16,16,64) \end{array}$	$18,432 \\ 36,864 \\ 0$
Convolution2D(ReLU) Convolution2D(ReLU) MaxPooling2D	$(16, 16, 128) \\ (16, 16, 128) \\ (8, 8, 128)$	$73,728 \\ 147,456 \\ 0$
Dense(ReLU) Dense(ReLU) Dense(Softmax)	$(512) \\ (256) \\ (52)$	$\begin{array}{r} 4,\!194,\!816 \\ 131,\!328 \\ 13,\!364 \end{array}$

Table 5.5: Summary of layers, output shape and number of parameters for the selected VGG architecture applied on upper and lower case letters of the COUT database.

5.3.3 ResNet architecture

The *Residual Networks* (ResNet) was proposed in 2015 by He el al. [91] as a strategy to overcome the limitations when trying to train models with a large number of layers. The strategy used is to define convolutional modules that have a built-in direct connection between the module input and output. The output of the convolutional module is combined with its input, and this combination is passed as input to the next convolutional module. In this way, in the mentioned paper 150-layer models are trained.

These networks are a particular case of the *Highway Networks* previously proposed by Srivastava et al. in [195]. The Highway Networks include in each convolutional module a gate that decides when the input to the module must be concatenated with its output. Thus, a ResNet is a Highway Net whose gates are always open.

The proposed architecture scheme is similar to the two previous cases. Several convolutional layer modules are followed by 3 dense layers. The difference is that in the case of the ResNet architecture, each convolutional module is more complex than those proposed before. Fig. 5.5 shows the detail of components of a ResNet module. The module is composed of the following layers:

- A first convolutional layer with a kernel of size 1×1 with 16 output feature maps.
- A second convolutional layer of kernel size 3×3 with 16 output feature maps.
- A third convolutional layer with a kernel of size 1×1 with 64 output feature maps.

All of the above convolutional layers are preceded by a batch normalization layer and employ ReLU activations. On top of the above, each block adds a direct connection between the input and output of the module, which is called *residual* and from which the name of the architecture derives.

In the original proposal, ResNet was oriented to the classification of natural images into 1,000 categories, and architectures of up to 150 layers were proposed. However, in our experiments with character databases, which are less complex, we decided to use only 3 blocks. The number of blocks has been fixed after performing experiments with a different number of blocks on the NIST database for classification of upper case letters. Table 5.6 summarizes the accuracy results in the validation and test partitions of the experiments performed.

Num. ResNet blocks	Accurac	ey
	Validation(%)	Test(%)
1	95.72	92.74
2	95.36	91.85
3	95.82	95.97
4	96.25	93.67
5	96.09	93.78

Table 5.6: Accuracy with the ResNet model over the NIST upper case database. Comparative over the number of blocks.

Fig. 5.5 shows the ResNet architecture used in the experiments for the handwritten character databases. On the left, the detail of each of the convolutional blocks is included, composed of convolutional layers and batch normalization layers [105] as indicated above. Additionally, the residual connection linking the input to the output is observed. The complete architecture is shown on the right, which includes 3 convolutional blocks.

Finally, Table 5.7 includes a detail of the layers for the case of the upper case letters image subset of the NIST database, with 64×64 input image size and 26 classes. The layer type, the output tensor dimension, and the number of trainable parameters of each layer are detailed. The size of the input tensor of each layer is the one indicated as output in the previous layer. The model is sequential except for the add-type layers that configure the residual connection. In these cases, the input layers to be added are indicated. The total number of model parameters is 8,531,242.



Figure 5.5: ResNet architecture: a) schema of a ResNet module and b) schema of the complete architecture with 3 modules.

5.4 Experimental results

This section details the experimental results obtained by applying the selected convolutional architectures on the considered databases. The results obtained are detailed next:

- Baseline results on the MNIST database, which allows having a first validation of the models on a well-known basic database.
- Comparative results. It includes the experiments carried out with the TICH and NIST databases. These databases are more complex, and published results are available to evaluate the models compared to models of other authors.
- Detailed results over the new COUT database to evaluate the behavior of the models with images of all types of characters, including special characters and punctuation marks. It allows us to evaluate the models in a scenario similar to that found in a sliding window strategy applied to the word recognition problem.

The proposed models have been applied to different character dictionaries. In particular, for the MNIST database, the 10 available digits have been used. We have also selected the available characters for the TICH database, which are digits and capital letters. In the case of the NIST database, separate models have been built for upper case, lower case and the combination of both types of letters to compare us with other authors. Finally, for the new COUT database, models have been built with the previous sets, plus one with the entire set of 93 characters, including letters, digits, punctuation marks, and other characters. Although the characters used in all cases are Latin, the methodology

Layes name	Layer type	Output Shape	Parameters
input	InputLayer	(64, 64, 1)	0
conv0	Convolution2D(ReLU)	(64, 64, 16)	144
bn0	BatchNormalization	(64, 64, 16)	64
conv1a	Convolution2D(ReLU)	(32, 32, 8)	128
bn1b	BatchNormalization	(32, 32, 8)	32
conv1b	Convolution 2D(ReLU)	(32, 32, 8)	576
bn1c	BatchNormalization	(32, 32, 8)	32
conv1c	Convolution2D(ReLU)	(32, 32, 16)	144
add1	Add conv1c and conv0	(32, 32, 16)	0
bn2a	BatchNormalization	(32, 32, 16)	64
conv2a	Convolution 2D(ReLU)	(32, 32, 8)	128
bn2b	BatchNormalization	(32, 32, 8)	32
conv2b	Convolution 2D(ReLU)	(32, 32, 8)	576
bn2c	BatchNormalization	(32, 32, 8)	32
conv2c	Convolution 2D(ReLU)	(32, 32, 16)	144
add2	Add conv2c and add1	(32, 32, 16)	0
bn3a	BatchNormalization	(32, 32, 16)	64
conv3a	Convolution 2D(ReLU)	(32, 32, 8)	128
bn3b	BatchNormalizationV1)	(32, 32, 8)	32
conv3b	Convolution 2D(ReLU)	(32, 32, 8)	576
bn3c	BatchNormalization)	(32, 32, 8)	32
conv3c	Convolution 2D(ReLU)	(32, 32, 16)	144
add3	Add conv3c and add2	(32, 32, 16)	0
bn4a	BatchNormalization	(32, 32, 16)	64
conv4a	Convolution 2D(ReLU)	(32, 32, 8)	128
bn4b	BatchNormalization	(32, 32, 8)	32
conv4b	Convolution 2D(ReLU)	(32, 32, 8)	576
bn4c	BatchNormalization	(32, 32, 8)	32
conv4c	Convolution 2D(ReLU)	(32, 32, 16)	144
add4	Add conv4c and add3	(32, 32, 16)	0
bnF	(BatchNormalization	(32, 32, 16)	64
Flatten	Flatten	(16384)	0
dense1	Dense(ReLU)	(512)	$8,\!389,\!120$
dense2	Dense(ReLU)	(256)	$131,\!328$
output	Dense(Softmax)	(26)	$6,\!682$

Table 5.7: Summary of layers, output shape and number of parameters for the ResNet architecture applied over the NIST upper case database.

presented can be used on other alphabets with minimal changes and starting with a character corpus for the considered language.

The training has been performed using a learning rate of 0.0001 and the optimization algorithm RMSProp [154]. A primary data augmentation strategy consisting of the following random transformations has also been employed: direct and inverse rotations up to 15 degrees, horizontal and vertical translations up to 20%, shear up to 15%, and zoom in and out up to 20%. A basic data augmentation strategy has been chosen because the objective of these experiments is to compare different architectures, not to obtain the best possible result. As indicated in [50], a more elaborate data augmentation strategy, including for example elastic deformations, would improve the overall accuracy of all models

evaluated.

The models are trained with an *early stopping* strategy until the validation error does not improve in 20 epochs. The best validating model is selected as the final model and it is used to evaluate the test data.

The main metric used to evaluate the results of character recognition models, given that they are multi-category classification models, is accuracy. Accuracy is defined as the percentage of cases correctly classified by the model over the total number of cases in an evaluation partition. A case is classified by a probabilistic model in the class in which it presents a higher output probability. Additionally, to better understand the errors made by each model, several confusion matrices described in Section 2.5.1 have been included. Since the number of target categories is large, only the rows and columns of the confusion matrix with significant errors will be shown.

5.4.1 Baseline results on MNIST database

The MNIST database is one of the most widely used in the analysis of Computer Vision models, and many publications use it in their benchmarks of new algorithms or methodologies (see for example [147], [217], [92], or [49]).

As indicated above, the particular configurations of the architectures for the experiments with this database have been adjusted, taking into consideration the reduced number of categories of the database. In particular, the first two dense layers have been set to 256 for the first one, and 128 for the second one.

The results obtained by the three architectures in terms of accuracy are included in the Table 5.8. It can be seen that the VGG architecture obtains the best result with an accuracy of 99.57%. In any case, the three models obtain good results, especially considering that no hyperparameter optimization has been performed to obtain them, and it can be considered that all these models are valid for this problem.

Architecture	Accuracy	
	Validation(%)	Test(%)
LeNet	99.15	99.28
VGG	99.51	99.57
ResNet	98.49	98.69

Table 5.8: Accuracy by architecture over the MNIST database.

The best results on the MNIST database with a convolutional architecture are reported in [217], which reports an accuracy of 99.79% using a new optimized dropout strategy called *dropconnect*. State-of-the-art is obtained by Mazzia et al. in [147] with an accuracy of 99,84% employing a type of model called *capsule networks*.

As shown in Fig. 5.6, the most frequent error is when confusing characters '9' and '4', and characters '3' and '5'. In the VGG model, these errors account each one for 14% of the total errors (6 out of 43 in both cases) in the test set. In the rest of the models, this error is also the most frequent one.



Figure 5.6: Confusion matrix (only errors) of the VGG model over the test partition of MNIST database.

5.4.2 Comparative results on NIST database

To determine whether the proposed models are competitive for character recognition over the whole vocabulary, we use the specific databases NIST and TICH. These two databases are different since NIST has the particularity that the images are binary, with only zero or one pixel values. Moreover, the TICH database contains grayscale images.

In the NIST database, experiments have been carried out with the three proposed architectures by setting the size of the first two dense layers to 512 and 256, respectively. The size has been doubled with respect to the experiments performed with the MNIST database, since the target size increases from 10 categories to a minimum of 26 categories. Results are provided for three subproblems summarized:

- Upper case characters recognition with 26 categories.
- Lower case characters recognition with 26 categories.
- Upper and lower case characters recognition with 52 categories.

The upper and lower case model that we have built to classify the character images has 52 letter categories (i.e., 26 upper case and 26 lower case ones). These categories are specially difficult because several letters like: 'c', 'f', 'i', 'j', 'k', 'o', 'p', 'u', 'v', 'w', 'x' or 'z' have similar shape for upper and lower case.

There exist several benchmarks for using the NIST database (2nd edition) [85]. One of the most accurate ones that use deep networks is [49], which provided recognition rates of 98.17% for upper case and 92.53% for lower case, respectively.

Other results are provided by Ciresan et al. [50], where a convolutional network architecture model with two convolution-pooling groups and a dense layer very similar to those proposed in this Thesis is used. In this work multiple trainings of the same architecture are executed, and committees of five trainings are used to analyze the error stability. Tables 5.9 and 5.10 included the results of this publication obtained on average by the committees and also the results of the best individual trainings.

In this database, the test sample does not follow the same distribution as the train and valid partitions. In the case of train and validation, there are certain characters that have a higher frequency, up to 10 times more than the less frequent ones, as is the case of the characters 'O' and 'S'. On the other hand, in the test partition, all characters have a similar frequency, about 400 cases per category. It makes the accuracy calculated for tests significantly lower than that obtained for the validation. In any case, the relative comparison between different test accuracy values is correct.

Results on NIST dataset for upper case letters

Table 5.9 includes the results of the upper case letter classification models from the NIST database. In this case, the best results are obtained by the VGG architecture, although the LeNet model obtains similar results. The ResNet model shows lower accuracy results.

Publication	Model	Accurac	ey
		Validation(%)	Test(%)
Ciresan 2011 [50]	Committee		98,09
Ciresan 2011 [50]	Best model		97.49
Ciresan 2012 [49]			98.17
Ours	LeNet	97.48	96.13
Ours	VGG	98.49	97.35
Ours	ResNet	95.37	91.85

Table 5.9: Accuracy by architecture over the NIST upper case dataset.

Fig. 5.7 shows a part of the confusion matrix (including only errors) of the model in which the most frequent errors of the model occur. It can be seen that the most common errors fall in the case of the upper case letters for VGG model occur when confusing the characters 'D' and 'O' (36 test cases), when confusing the characters 'U' and 'V' (31 test cases), and when assigning as 'C' 14 characters that are really 'L' ones.

In the upper case, the result of the VGG model is quite close to the published results [50] [49], even though the two works cited above seek to minimize the error. For this purpose, these works optimize the preprocessing of the database, apply a complex data augmentation, and adjust both the model design and the training algorithm. All these steps are applied in order to minimize the error. The objective of the present study is not to optimize the above parameters to obtain the best possible result but to evaluate three different architectures to determine their suitability for using them in continuous HTR models. It partly explains why the results obtained are a bit worse than those published.

Upper case letters are easier to identify than lower case letters, and upper case letter shapes have less variation because they are well defined and have smaller intra-class variability (because there is less stroke variations in the different writing styles).



Figure 5.7: Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST upper case letters.

Results on NIST dataset for lower case letters

Table 5.10 presents the results of the lower case letter classification models from the NIST database are included. The results are similar to the previous ones, with the VGG model obtaining the best result and the LeNet and ResNet models obtaining a worse results, especially ResNet.

Publication	Model	Accurac	ey
		Validation($\%$)	Test(%)
Ciresan 2011 [50]	committee		92,29
Ciresan 2011 [50]	best model		91.16
Ciresan 2012 [49]	Convolutional		92.53
Ours	LeNet	$95,\!05$	$87,\!30$
Ours	VGG	96.60	90.58
Ours	ResNet	92.99	83.12

Table 5.10: Accuracy by architecture over the NIST lower case dataset.

The best performing model (i.e. VGG) also achieves an accuracy relatively close to the results of the individual models of other authors. Fig. 5.8 shows the confusion matrix of this model. Again, only the rows or columns with relevant errors are shown.

The main source of error comes from classifying most of the images of the character 'i' as the character 'l' (el). It is also important to remark that there are the main errors when confusing 'g' and 'q', 'f' and 't', and 'u' and 'v'.



Figure 5.8: Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST lower case letters.

The full confusion matrix shows that practically all 'i' characters are classified as 'l'. This is because these characters are very similar and that in the train partition, there are 15,213 cases of the 'l' character and only 2,517 of the 'i' character. This difference in frequency is reflected in the biases associated with each target class of the last classification layer. This means that any image of the 'i' character on which the model does not achieve a precise identification ends up by classifying it as 'l'.

Results on NIST dataset for upper and lower case letters

Finally, Table 5.11 includes the results on NIST database for upper and lower case letters. This is the most complex case since there are several characters with similar upper and lower case spellings. In particular, the characters: 'C', 'F', 'I', 'J', 'K', 'L', 'M', 'O', 'P', 'S', 'U', 'V', 'W', 'X', 'Y', and 'Z' have that particularity and are usually distinguished by their relative size compared to other character.

The VGG architecture once again obtained the best results. In this case, the accuracy of models is further away from those obtained by other authors.

Fig. 5.9 shows that most frequent errors occur precisely when confusing the same character in upper and lower case. In particular, and by error volume, with the characters: 'I', 'C', 'M', 'U', 'O', 'S', 'V', 'P', 'W', 'J', 'X', and 'Z' respectively.

Where creating the NIST database, all characters are resized to the same size, and then a blank border is added to them. It eliminates the possibility of differentiating between characters of the same size in the upper and lower cases, and it explains the high number of errors in this confusion matrix.

Publication	Model	Accurac	ey
		Validation(%)	Test(%)
Ciresan 2011 [50]	committee		$78,\!59$
Ciresan 2011 [50]	best model		76.99
Ours	LeNet	85.70	68.13
Ours	VGG	87.56	70.86
Ours	ResNet	82.16	64.53

Table 5.11: Accuracy by architecture over the NIST upper and lower case letters dataset.



Figure 5.9: Truncated confusion matrix (only errors) of the VGG model over the test partition of NIST upper and lower case letters.

5.4.3 Comparative results on TICH database

The TICH database contains grayscale images of the upper case letter and digits characters with 36 categories. In Table 5.12 the results obtained by the three proposed architectures are detailed. A baseline proposed by the database author is also included in reference [143].

Using the TICH database [143], Van der Maaten provided a best error recognition result of $82.77\% \pm 0.82\%$ in recognition using 10-fold cross-validation procedure and a k-NN classifier. In this case, the comparative result corresponds to a baseline provided by the

Publication	Model	Accuracy	
		Validation(%)	Test(%)
Van der Maaten [143]	k-NN		82.77 ± 0.82
Ours	LeNet	94.91	95.35
Ours	VGG	95.94	96.54
Ours	ResNet	95.04	94.91

Table 5.12: Accuracy by architecture over the TICH database.

database author based on k-neighbors which is not optimized, and which the convolutional models outperform by far.

In the case of TICH dataset, the architecture with the lowest test error is the VGG with an accuracy of 95.54%. The results obtained with this database, which includes the 10 digits and 26 upper case letters, are very similar to those obtained for upper case letters in the NIST database detailed in Table 5.9. The fact that the results, in this case, are lower comes from the fact that classifying the digits attached to upper case letters introduces more sources of error. In particular, the confusion between the characters 'O' and '0' (i.e., letter O and digit zero) and between the characters 'I' and '1' (letter I and digit one) are the two primary sources of error, as seen in the confusion matrix in Fig. 5.10. This figure shows the section of the confusion matrix that includes the main errors of the model.



Figure 5.10: Truncated confusion matrix (only errors) of the VGG model over the test partition of TICH database.

5.4.4 Results on the new COUT database

In this section, the experimental results on several subsets of the new COUT database (Characters Offline from UNIPEN Trajectories database) are presented. This database is built by us from character stroke trajectory data extracted from the UNIPEN online database.

In particular, classification experiments have been performed on the following subsets of characters:

- Upper case letters: with 26 categories (results presented in Table 5.13).
- Lower case letters: with 26 categories (results presented in Table 5.14).
- Upper and lower case letters: with 52 categories (results presented in Table 5.15).
- All characters in the database: including upper and lower case letters, digit, punctuation and other symbols: with 93 categories (results presented in Table 5.16).

For each of the experiments, two runs of each experiment have been carried out. The first one (i.e. original data) uses the original images from the COUT database. In the second one (i.e. altered data), the training partition is transformed following the algorithm presented in Subsection 5.2.2, which modifies the images of each character by adding artifacts that simulate traces of previous and subsequent characters. This process simulates the effect of having been extracted them by a sliding window that runs through a handwritten word. In this way, it will be tested whether the proposed architectures are robust to the presence of these artifacts. In this second case, the algorithm is applied with a multiplicative factor of 8 that increases the training sample by this factor. That is, for each image of a character, 8 distorted versions of that image are generated. The validation and test partitions are not modified to obtain comparable results.

Results on COUT for upper caseletters

As shown in the Table 5.13, the test accuracy for upper case characters is similar in the models built on the original images and on the altered ones. It indicates that the ability to discriminate upper case letters is not reduced by the artifacts added to the left and right character images. The accuracy is also similar to that obtained in the experiments with the TICH and NIST databases.

The VGG model obtains the best results on the two datasets. The ResNet model performs well on the original image database. However, on the altered database, the performance drops significantly. The most frequent errors occur between the letters 'U' and 'V', and between the letters 'M' and 'N', respectively.

CHAPTER 5. DEEP ARCHITECTURES FOR ISOLATED OFFLINE CHARACTER HANDWRITING RECOGNITION

Model	Altered	Accuracy	
		Validation(%)	Test(%)
LeNet	Yes	94.66	95.44
	No	96.87	96.67
VGG	Yes	96.55	97.41
	No	98.31	98.34
ResNet	Yes	85.94	86.07
	No	96.55	97.23

Table 5.13: Accuracy by architecture over the new COUT upper case dataset.

Results on COUT for lower case letters

The results corresponding to the lower case character set are detailed in the Table 5.14.

Model	Altered	Accuracy	
		Validation(%)	Test(%)
LeNet	Yes	89.83	89.48
	No	95.20	95.09
VGG	Yes	93.03	92.82
	No	96.04	96.09
ResNet	Yes	68.70	69.58
	No	92.12	92.52

Table 5.14: Accuracy by architecture over the new COUT lower case dataset.

The model with the highest accuracy is again the VGG. Moreover, the difference with the other two models is quite significant in the modified database. The ResNet architecture model obtains worse results than the other two ones, as it was the case in the experiments performed on the lower case subset of the NIST database.

In the lower case letters, test results are superior to those obtained with the NIST database for the same character set. However, this happens because in the NIST database the test results were biased due to the difference of target distribution in that partition. Comparing with the results in the validation partition of the NIST database, which has a target distribution equal to that of the train, it is observed that the results for the two databases are very similar. For example, the VGG model in NIST obtains an accuracy of 96.6% in validation, and in the original COUT database, it obtains 96.04%.

There are more significant differences between the original and the modified database results in this lower case instance than in the upper case. It is especially important in the ResNet model, which drops its accuracy from 92.52% to 69.58%.

As it can be seen in the confusion matrix shown in Fig. 5.11 the main errors come from confusing the characters 'e' and 'i' with the character 'l', and confusing the characters 'u' and 'v'.



Figure 5.11: Truncated confusion matrix (only errors) of the VGG model over the test partition of COUT lower case letters.

Results on COUT for upper and lower case letters

The results corresponding to the charset of upper and lower case letters are detailed in the Table 5.15.

Model	Altered	Accuracy	
		Validation(%)	Test(%)
LeNet	Yes	76.74	77.47
	No	84.50	85.66
VGG	Yes	81.48	82.37
	No	87.03	87.20
ResNet	Yes	55.91	56.51
	No	80.89	82.56

Table 5.15: Accuracy by architecture over the new COUT upper and lower case dataset.

The conclusions are very similar to those obtained with the lower case model above. The VGG model is the one that obtains the best results. It also happens that the results in the validation partition are very similar to those obtained by each of the architectures on the same NIST database charset. Finally, also ResNet model is the one that shows more differences when trained with the distorted database.

As in the NIST database, the main errors come from confusing the characters with similar graphism for upper and lower case letters. However, in this case, the errors are less

frequent, probably because in COUT database construction the relative differences in the size of the characters have not been altered, as it was the case in the NIST database. It allows the model to better differentiate characters with identical spelling in upper and lower cases.

Results on COUT for all the character set

To conclude this subsection, the experimental results with the images of the 93 characters included in the database are summarized. Only results from the original database are included since the implemented data augmentation algorithm only applies to upper and lower case letters (i.e. the altered data is not available for this character set). The results are collected in the Table 5.16.

Model	Accuracy	
	Validation(%)	Test(%)
LeNet	83.86	83.62
VGG	86.22	86.17
ResNet	78.52	78.81

Table 5.16: Accuracy by architecture over the new COUT database

The differentiation of the 93 characters of the COUT database is probably the most complex problem among those addressed in this chapter. In addition to the similarities already analyzed between certain upper and lower case letters, there are also similarities between punctuation marks and other special characters such as: '(', '[', '{', ')', ']' or '}'.

The VGG model obtains the best outcome, and the results of LeNet and VGG models are very similar to those obtained in the previous experiment with the character set of upper and lower case letters. It shows that the recognition of special characters does not present significant errors.

A section of the confusion matrix of the VGG model for this problem is included in Fig. 5.12. It can be seen that the main errors are those appearing in the previous model on the upper and lower case letter charset when confusing characters with similar graphics in the two cases. In the case of special symbols, the most frequent error consists in confusing the addition symbol '+' with the letter 't'.

As a summary of the different experiments performed with this COUT database, it is indicated that the VGG architecture obtains the best results regardless of the subset of characters analyzed. It is also the architecture that presents the most similar results when comparing the augmented database versus the original one.

Input resolution influence on the COUT database

To determine the importance of the resolution of the input images, we built the model of the lower case charset with all architectures and the same training parameters but only changing the resolution of initial images to 32×32 pixels. The lower case model is selected because it deals with the most frequent characters in continuous writing, and the errors



Figure 5.12: Truncated confusion matrix (only errors) of the VGG model over the test partition of COUT database.

presented by this model will be the most common ones when dealing with the continuous HTR problem. A recognition accuracy comparative over the test images was included in Table 5.17 including accuracy for this new image resolution.

Model	Resolution	Parameters	Accuracy	
			Validation(%)	Test(%)
LeNet	64×64	0	95.20	95.09
	32×32	0	94.40	94.59
VGG	64×64	0	96.04	96.09
	32×32	0	96.08	96.43
ResNet	64×64	0	92.12	92.52
	32×32	0	93.72	94.56

Table 5.17: Accuracy by input resolution on the COUT test partition.

The models obtain very similar accuracy when applied to a 32×32 pixel image instead of a 64×64 pixel image. In the sense that it is closest in accuracy at the two resolutions, being the most stable model the VGG one.

The biggest difference is observed in the *ResNet* model, which obtains better results at

the lower resolution of 32×32 . It may be because the ResNet architecture needs detailed fine-tuning to obtain good results in each particular case. For example, when changing the size of the input image. This need for fine-tuning may also explain the very different behavior of this model in the tests performed with the different databases. It is recalled that for selecting the *ResNet* architecture for these experiments, basic fine-tuning of the architecture was done to determine the most appropriate number of *ResNet blocks*. This fine-tuning was done on the upper case letters of the NIST database. It is precisely in the upper case classification models, built for the different databases, where this model has a more similar result to the other two ones. It can be seen in Table 5.9 for the NIST database, Table 5.12 for the TICH database and Table 5.14 for the COUT database without data augmentation.

5.5 Discussion of results

The main results and conclusions obtained from these experiments are the following ones.

- The VGG architecture has proved to be the most efficient and the most versatile. It obtained the best results in all the experiments. It has not lost efficiency, always providing high accuracy results and close to those published in the literature, even though all the experiments have been performed with the same parameterization, and without any specific fine-tuning.
- LeNet architecture, despite being the simplest in terms of layer depth, has shown to work correctly in most of the experiments performed. Moreover, the number of parameters of this architecture is lower than the other two ones. This makes it a good candidate to obtain a suitable baseline for this type of HTR problem. It has only presented a decrease in accuracy with respect to the VGG architecture in the more complex models, with a higher number of classes. We assess that it is likely to be necessary to increase the number of network parameters by increasing the dimension of the layers in these cases.
- The ResNet architecture, which was the most complex one in terms of the number of layers, presents a lower accuracy than the other, in some cases much lower. It has been found that for the upper case dataset, on which the initial fine-tuning of this architecture was done, good results were obtained. However, for other cases, the performance dropped significantly. We conclude that this architecture may need specific fine-tuning to adapt it to each of the different experiments performed.
- In the experiments performed with the upper and lower case character set, it has been observed that the results improve if the database maintains the relative sizes between letters. This is because the model makes fewer errors when distinguishing between characters with the same spelling in the upper and lower case. This insight has been used in the normalization strategy of the word databases used in the next chapter.
- The results on extended character sets, combining digits, upper and lower case letters, and symbols, are the most relevant ones to evaluate the performance of each

architecture in the field of free text recognition. But precisely these results on isolated characters present obvious errors because several characters have very similar strokes and that people can only distinguish them by context. For example, the letter 'O' (upper case) and the digit '0'. As it is already known, these experiments do not allow to evaluate the ability of a model to differentiate characters by context, and only by spelling. When these architectures are used as part of a more general model of continuous text recognition, there will be such an answer. At that point, it will be possible to evaluate the ability of these models to take context into account when recognizing text.

Chapter 6

Deep architectures applied to continuous offline handwriting recognition

This chapter describes the models and experiments performed for recognizing continuous handwritten text present in images. The main objective is to develop a model capable of identifying the sequence of characters appearing in a continuous text image. For this task, a new model based on the *seq2seq* architecture, introduced by Sutskever et al. in [204], is proposed. Systematic experiments have been performed to evaluate the new model on multiple corpora, and the results have been compared to those of other authors.

In the next sections, we explain a novel architecture proposal for the offline handwriting recognition problem. This architecture combines a Convolutional Neural Network (CNN) with a seq2seq model. The CNN models the visual attributes of the handwritten words and provides a sequence of visual features from each part of the word image. This feature sequence is used as input to the seq2seq model. Across these inputs, the seq2seq network identifies the characters of the word using its encoder-decoder functionality. This functionality is designed to solve transcoding problems that need to transform one input sequence signal into another output sequence signal of different length. These architectures have been proven to successfully solve complex problems, such as machine language translation [204], speech recognition [45] or lip-reading [47].

For the evaluation of the proposed model, an experimental setup is defined, which includes a novel text image normalization strategy and a specified data augmentation algorithm. A model training strategy and a set of lexicons for dictionary-based decoding results are also proposed.

In addition, we analyze the effect of different parameterizations in the proposed architecture over the accuracy of the model. This analysis aims to check the stability of this model over changes in the parameters values, and conduct ablation analysis that measure the impact of the several components of the model on the error. It has been done using three different corpora in different languages to guarantee the generality of conclusions. Although the three selected databases are from languages with Latin characters, we think that the proposed solution is general enough and could be used on other character sets as Arabic or Chinese.

Finally, our results are compared to other related approaches to evaluate the quality of the

results concerning the state-of-the-art of the continuous HTR problem. Comparing results with different authors is complex due to the tremendous variability in the literature for existing data, dictionaries, and experimental strategies. These difficulties are analyzed, and solutions are proposed in specific sections.

The following section details the architecture of the model proposed. The necessary details of the experimental setup are provided in Section 6.2, and the results of the experiments performed in Section 6.3. In Section 6.4 the difficulties encountered in the comparison with other authors are analyzed. Based on the previous analysis, in Section 6.5, a framework for standardizing the results is proposed to solve these difficulties, and the best results obtained by the proposed model are compared with other approaches over the same databases and with the same lexicons. Finally, the results obtained in Section 6.6 are discussed.

6.1 Model architecture

This section describes the architecture of the new model proposed. The model transforms a variable-length handwritten word image into the variable-length sequence of the characters that conform the previous word. The model has three main components: a convolutional reader, an encoder, and a decoder with an attention mechanism. These components are combined into a seq2seq architecture as shown in Fig. 6.1.



Figure 6.1: Components of the proposed high-level model architecture for word recognition.

Following the notation of Fig. 6.1, it is possible to summarize the previous model by the sequence of equations 6.1:

$$x = C(I)
 (c, h) = E(x)
 a = A(h)
 l_t = D(a, c, l_{t-1})$$
(6.1)

The convolutional reader function C is a deep CNN that converts the input image I into a vector of visual features x. The input to this convolutional reader consists of the bi-dimensional array of pixel values of the word image. The convolutional model

extracts feature vectors of this image related to the letters that appear in it. Two types of convolutional readers have been defined. The first one employs a single convolutional model on the input image and uses a concatenation strategy of feature maps by horizontal positions to obtain an ordered sequence of output features. The second one divides the image into overlapping patches that run through the image from left to right, and applies the convolutional model on each of the defined patches. This second strategy naturally generates an ordered sequence of convolutional features.

Next, the encoder E function, a RNN of type LSTM or GRU, reads the sequence of convolutional features x and extracts the sequence relationships among these features. So, the encoder transforms x into a sequence of encoded output features h that model the relationships among the word letters. In addition, the final state c of the encoder RNN was preserved to use as an initial state in the RNN decoder.

Finally, a decoder D gets the previous encoder output transformed by the attention mechanism A to predict, letter by letter, the word. The decoder consists in a LSTM that generates each current letter l_t combining the state vector c, the encoder outputs transformed by the attention mechanism a and the previous letter l_{t-1} .

The attention mechanism A helps to focus on specific parts of the encoder output features h that are relevant to identify specific letters. For example, the first letters need to focus on the first elements of the encoder output features that represent the first part of the word image. On the other side, the last letters need to focus on the final part of the encoder features associated with the last part of the word image.

The following sections describe this architecture in detail. Subsection 6.1.1 comprises a detailed model schema in which the architecture of each of its components is included. The following subsections detail each of the components of the model: he convolutional reader is presented in Subsection 6.1.2; the encoder, in Subsection 6.1.3; the decoder, in Subsection 6.1.4; and the attention mechanism in Subsection 6.1.5.

6.1.1 Detailed schema of the model

In this subsection, we include the detailed schema of the proposed model. In the schema, we include the details of the different model components: convolutional reader, encoder, decoder, and attention mechanism. Additionally, we also include the details of how these components are connected. Fig. 6.2 shows the detailed schema of the complete model architecture. This figure follows the notation introduced previously.

The proposed solution is designed to model each letter l_i in the word $w = (l_1, l_2, ..., l_n)$ as a conditional probability of the input image I and the probability distribution of the previous letters of the word $l_{\leq i}$.

$$\mathbf{P}\left(l_i|I, l_{< i}\right) \tag{6.2}$$

Therefore we model the output probability distribution of a word given the sequence of patches, as follows:

$$\mathbf{P}(w|I) = \prod_{i} \mathbf{P}(l_i|I, l_{
(6.3)$$



Figure 6.2: Detailed model architecture.

In the following subsections, all the elements of the architecture that implements this model are detailed.

6.1.2 Convolutional reader

The objective of the convolutional reader is to extract the visual features which are relevant to identify the characters appearing in the input word image. The selection of the convolutional architecture is one of the most relevant aspects in the model design. It is also the component of the model that presents more variability in its configuration. In the previous chapter, on models for isolated characters identification, three different convolutional architectures were analyzed: LeNet, VGG, and ResNet, respectively. In that chapter, it was concluded that LeNet and VGG architectures obtained good results in character recognition, especially the second one, and that ResNet architecture only offered good results if it was properly configured for each specific target. Based on this, the VGG and LeNet architectures are the ones on which the current word recognition model will be evaluated. The details of the architectures and a diagram of them can be seen in Subsection 5.3.1 for LeNet and in Subsection 5.3.2 for VGG.

Since the output of the convolutional component is fed as input to it, the encoder must consist of a sequence of convolutional feature vectors. Furthermore, the features entering the encoder are expected to encode aspects of the input image sequentially. The first features encode characteristics related to the initial part of the image, and the last features encode the corresponding ones of the final part of the image of the handwritten text. The encoder will describe the sequential relationships in the image in the same way that humans read the text sequentially. Following the notation of Fig. 6.2, the output of this convolutional component is a sequence of vector features $x = (x_1, x_2, ..., x_n)$ that encapsulate the visual features of the handwritten word image.

One way to obtain the above representation is to extract from the image a sequence of vertical patches of fixed width in pixels, using a given pixel step size. From each patch, the convolutional component computes a vector that contains the visual features related to the character or characters included in it. The image patches are the inputs, and the features obtained from the last dense layer are the outputs. Alternatively, the convolutional component can be applied directly to the complete input image. In this case, a specific transformation must be applied to the output of the convolutional component to convert it into the input of the decoder RNN. The transformation must construct an ordered sequence of feature vectors that sequentially encode the image as indicated above.

Two types of convolutional components are used: the ones based on image patches and the ones using directly on the complete image, are described in detail below.

Convolutions over image patches

The first strategy for applying the convolutional architecture consists of using the same convolutional model to an ordered sequence of patches extracted from the original image with a sliding window that runs horizontally from left to right. In this way, we extract a sequence of horizontal overlapped slides for each original word image. These slides can include a part of a letter, one letter, or more than one letter. The slides are the inputs to our model, and they enable it to view the word image not globally but as a sequence of patches. These are characterized by two parameters: the patch size (i.e., its horizontal size) and the patch step (i.e., the number of pixels that we move to the right to extract the next patch). Fig. 6.3 shows an example of an original word image and its corresponding extracted patches.



Figure 6.3: Word patch extraction: (up) original image word, and (down) extracted input model patches.

The same convolutional model is applied on the previous slide images using a sharing parameters technique [236]. In this way, the model focuses on extracting the convolutional characteristics of the image region that appears in each patch, generating an ordered sequence of such feature vectors. The RNN of the decoder will model the sequential nature of these features.

The two convolutional architectures used are VGG and LeNet, that were selected in the previous Chapter 5. The only modification made is the removal of the last two dense layers.

These layers are not included because they were oriented to build the character classifier. In this case, what we seek to obtain as output are the convolutional characteristics of each patch.

Convolutions over complete images

In the case of convolutional architectures used directly on the input image without creating patches, an operation must be applied in order to adapt the output of the convolutional network to the input required by the RNN of the decoder. In addition, this operation must generate a sequence of vectors ordered concerning the columns of the input image; that is to say, the first vectors must collect features from the left side of the image and the last vectors, the ones from the right side.

For this purpose, the convolutional network output feature maps are resized into a twodimensional array used as input to the RNN component of the encoder. The resizing is done by concatenating the output feature maps of the convolutional component by columns. It ensures that the input sequence to the encoder RNN has the information ordered in the same way as in the input image but transformed by the convolutional component. Fig. 6.4 illustrates how columns traverse the feature maps to generate each vector of the output sequence by concatenation.



Figure 6.4: Residual connection to convert the convolutional output into the feature vectors sequence.

The convolutional architectures applied in this case are the same as in the previous patch application, LeNet and VGG. The only difference is that the previous transformation can only be applied to the output of a convolutional or pooling layer. Therefore, in this case, the dense layers have been removed from the convolutional component.

6.1.3 Encoder

The purpose of the encoder is to capture the input image characteristics corresponding to the sequential nature of the text. To do that, the encoder is implemented using RNN layers. The RNN encoder receives the sequence of vector features x_t generated by the convolutional reader and generates as output a sequence of output vectors h_t^e , each one for each encoder step t. The first element h_0^e of the RNN of the encoder is initialized randomly.

The encoder architecture is another essential component to be analyzed along with the convolutional component described above, since it must model the elements related to the sequential nature of the text. In our case, we chose to perform several experiments to determine if there is an encoder configuration that consistently provides better results in the different databases analyzed. Specifically, we have analyzed the type of RNN to be used, which can be LSTM or GRU, the number of RNN layers to be used, as well as their dimensions, and we have also experimented with simple forward RNNs or bidirectional RNNs. In Subsection 6.3.5 we detail the results of these experiments.

The output sequence of the RNN h^e will be passed as input to the attention mechanism. In bidirectional RNNs, where there are two output sequences, forward and backward, these are concatenated to generate the final input sequence to the attention mechanism. In addition, the encoder RNN returns a final state vector h^e_{last} , which is used as the initial state of the decoder RNN. In bidirectional RNNs, the state of the forward network is selected as the final state.

6.1.4 Decoder

The decoder is responsible for generating the output sequence formed by the ordered characters that make up the text of the input image, i.e., the transcription of the handwritten text image.

Like the encoder, the decoder is also based on an RNN network. In addition, in the decoder architecture proposed in this Thesis, it has been chosen to use the same type of RNN as the encoder. Thus, in the experiments in which the encoder uses an LSTM type network, the decoder will use an RNN of the same type. The same happens if the encoder uses a GRU type network.

The decoder operates so that each step generates one by one the characters of the transcription. During the training stage of the model, the decoder receives as input in each step the previous character of the transcript y_{t-1} and the context information provided by the attention mechanism a_t . With this information, it generates as output an estimate of the next character of the transcript \hat{y}_t . The decoder starts with the special signal $\langle GO \rangle$ as input to decode the first character of the word image. It ends when at the output appears the special character $\langle END \rangle$, by selecting the sequence of previous output characters as the decoded word. The initial state of the decoder RNN h_0 is initialized with the last state of the encoder h_{last}^e . This way of initializing the decoder means that the dimension of the decoder RNN must be the same as the dimension of the encoder RNN. Fig. 6.5 schematizes the operation of the decoder in the train stage.

Performing model training using the real y_t values as input to the RNN is called *teacher* forcing and was introduced by Williams et al. in [224]. This technique is described in detail in Subsection 4.8.3 of Chapter 4.



Figure 6.5: Decoder training diagram.

The probability distribution of the output character for each step is generated by a *dense* layer with the *softmax* activation function [35] applied over the output sequence of the RNN. Eq. 6.5 details the behavior corresponding to the decoder train process following the notation of Fig. 6.5.

$$y_{0} = \langle GO \rangle$$

$$h_{0}^{d} = h_{last}^{e}$$

$$h_{t}^{d} = RNN(concat(y_{t-1}, c_{t}), h_{t-1}^{d})$$

$$p(y_{t}) = softmax(dense(h_{t}))$$

$$\hat{y}_{t} = argmax(p(y_{t}))$$
(6.5)

The estimated output sequence $\hat{\mathbf{y}} = (\hat{y}_1, ..., \hat{y}_k)$ is built selecting the maximum probability values in $p(y_t)$ by argmax function for each t, as it is indicated in the equation 4.11.

Unlike the rest of the model, the decoder has a different behavior over the model train phase and the model evaluation or score phase. During the scoring phase, it is not possible to use the output character sequence y_t as input to the decoder, as is done during the training phase. Instead, as input for step t, the character estimated by the decoder for step t - 1 is used, i.e., y_{t-1} . Details of the scoring process are given in Fig. 6.6.

In the score or validation stage, the equation of the previous RNN is modified as shown in Eq. 6.6. Both real and estimated characters are used encoded with one-hot encoding.

$$h_t = RNN(concat(\hat{y}_{t-1}, c_t), h_{t-1}) \tag{6.6}$$

The training process can also be performed following the same scoring principle of using the prediction of the previous step as input to the next step. It is an alternative to the teacher forcing strategy mentioned above.



Figure 6.6: Decoder score diagram.

The objective of the decoder is to learn how to generate the output character sequence. For identifying the next character in the sequence, the RNN of the decoder receives as input the context information of the input image produced by the attention mechanism in addition to the previous character or the estimate of that character in the previous step. The estimation of the next character in a sequence from the previous characters through an RNN is precisely one of the types of language models analyzed in Section 4.10 of Chapter 4. It means that the decoder could be acting as a character-level language model. However, our model is trained at word level only with words present in the handwriting training database. This fact considerably limits the capacity of the decoder to act as a true language model. A true character language model typically needs characters of the previous words to capture the grammar and several orders of magnitude more text than is present in the handwriting databases.

6.1.5 Attention mechanism

The attention mechanism connects the sequence of feature vectors generated by the encoder $h_1^e, ..., h_n^e$ with each step of the decoder. The encoder output encodes the visual features obtained by the convolutional module and the sequential features obtained by the encoder RNN. It allows each decoder step to focus on a specific part of the relevant encoder output for this step. It is very important on the architecture because it allows the decoder to pay attention to the input data part that is more relevant to decode the current character. For example, if the decoder estimates the first letter of the word, it can focus on the features generated by the first part of the word image. And so on, with all the characters to be predicted.

Following the notation of the previous sections, to calculate the attention vector c_t at each output time t over a encoder output length N_e , the detailed equations in 6.7 are used:

$$e_{t,i} = w^T \tanh(Wh_i^e + Vh_{t-1}^d + b)$$

$$a_t = softmax(e_t)$$

$$c_t = \sum_{i=1}^{N_e} a_{t,i}h_i^e$$
(6.7)

where w and b are trainable weight and bias vectors, W and V are trainable matrices, h_i^e is the hidden state of the encoder at time $i \in [0, 1, ..., N_e]$ and h_{t-1}^d is the hidden state of the decoder at time t-1. The attention mechanism is implemented as a neural network layer that combines the states of the encoder and the previous states of the decoder with the hyperbolic tangent as an activation function. The vector e_t contains in the position i a score of how much attention we need to put the current decoder step t on the i-th encoder hidden state h_i . Then, we apply the softmax function to normalize it and create the attention masks a_t over all the encoder states. The final attention vector c_t is concatenated with the input of the decoder layer as we show in Fig. 6.2.

As indicated in Subsection 4.8.4 of Chapter 4 there are different types of attention mechanisms. In the case of the implemented model, a non-monotonic content base attention network characterized by the above equations 6.7 has been selected.

The selected attention mechanism is a content-based type, which does not encode positional information. This information is necessary to distinguish the same character at different locations in the input image. This is done because it is assumed that the RNN of the encoder already encodes this positional information, and therefore it is not necessary for the attention mechanism to do so as well.

The choice of a non-monotonic attention mechanism may be controversial since reading a handwritten word could be considered a monotonous task because it is needed to read the left letters before the right ones. However, this fact is not true for algorithms. Reading letters of a word, written using a Latin script, from left to right, is conventionalism, and it is possible reading them from right to left without loss of information. So, to allow the model to consider all the neighborhood information, the selected attention mechanism has no monotonicity restrictions.

To illustrate this assumption, we plot in Fig. 6.7 the outputs $a_{t,l}$ of the attention mechanism for two words of the validation partition. These figures present in the horizontal axis the image patch number and in the vertical axis the predicted character position. The figures show that, in general, the attention mechanism is working monotonically, but for some character positions, it uses information from patches at the left of what will be a monotonous behavior.

6.2 Experimental setup

A set of experiments has been designed to validate the usefulness of the proposed new model in the continuous HTR problem. In this section, the common elements of these experiments are defined. These elements are defined to ensure that the results of the experiments provide adequate information on the effectiveness of the proposed model. They also ensure that the results allow a correct comparison with results of other models from different authors.

In Subsection 6.2.1 we include a brief description of the software tools employed to build the different experiments. Next, Subsection 6.2.2 details the configuration defined for the



Figure 6.7: Example of attention mechanism outputs for two words and their respective original handwritten word images.

training of each model. In Subsection 6.2.3 describes in detail the image preprocessing algorithm used to normalize the handwritten text images before inserting them into the model. Subsection 6.2.4 describes the data augmentation algorithms implemented and used in some of the experiments. Then, in Subsection 6.2.5 we provide details about the different lexicons used for word decoding.

6.2.1 Source code and software

Different software tools are used to develop the experiments and to perform the different steps of the recognition process. Table 6.1 includes a version summary, download links, and application areas.

In order to facilitate that the results can be reproduced independently, all the software selected has open licenses and is freely available to download and use.

Name	Version	Link	Usage
Tensorflow	1.8	www.tensorflow.org	Modelling
OpenCV imgtxtenh	$\begin{array}{c} 3.0\\ 1.0\end{array}$	www.opencv.org www.github.com	Image preprocesing Image Preprocesing

Table 6.1: Software employed in the experiments.

Tensorflow [1] is a definition and execution of machine learning algorithms software. It has much flexibility to implement complex algorithms and is capable of scaling the execution of the same on large volumes of data. It is available for download from http://www.tensorflow.org.

Open CV (Open Source Computer Vision Library) is an open-source Computer Vision and machine learning software library. It is available for download from http: \www.opencv.org. It has a BSD-license that allows, among other things, its use for research purposes. The library provides C++, Python, Java, and MATLAB interfaces and supports the most common operating systems. This Thesis uses the Open CV Python interface over the Ubuntu operating system to develop some image transformations in the image preprocessing step.

imgtxtenh [214] is a tool to clean and enhance noisy scanned text images, which could

be printed or handwritten text. It is available for download from https://github.com/ mauvilsa/imgtxtenh (last access September 2020)

6.2.2 Implementation details

As trainer, we used the Adam stochastic optimization algorithm, with a batch size of 128 and with a dropout of 0.5 [164] in the dense layers of the convolutional part and in the cells of the encoder RNN layers. An initial learning rate of 0.001 was used, and it was decreased by 2% in each epoch. We applied stopping criteria of 20 epochs without any improvement in the Word Error Rate (WER) on the validation set. In general, the experiment configuration that has obtained a lower WER in the validation data is considered the most appropriate. In any case, the simple parsimony principle is applied [212], and when two configurations with similar results are presented, the one that produces a model with a smaller number of parameters is selected. The loss function used is the *weighted cross-entropy* loss for a sequence of logits. On this loss, it is applied on all the weights of the model a type L2 regularization with a parameter of 0.0001.

In all the experiments, the model weights are randomly initialized through the technique called *Glorot normal* [76] that initializes the weight of each node using a random value obtained from a Gaussian distribution of zero mean and standard deviation proportional to the number of connections of the node.

In each epoch, we randomized the order of the training data. The different experiments took between 60 and 280 epochs. The duration of the experiments varies greatly depending on the database, the model parameters volume, and whether or not data augmentation is used. In general terms, each experiment with the IAM and RIMES databases took between 10 and 20 hours, and the experiments with the Osborne database between 2 and 4 hours.

Once the model has been trained, it is applied to the evaluation and test datasets to obtain the accuracy metrics, producing three different groups of results: the raw accuracy of the visual model without any lexicon, the improvement obtained when using search over the standard lexicon of each database, and finally, the error obtained when eliminating the out-of-vocabulary (OOV) words using the test lexicon.

6.2.3 Image preprocessing

The data used in our experiments were preprocessed to correct some typical characteristics of the handwriting text that difficult the recognition task. For this purpose, a specific algorithm has been developed which, in summary, has the following steps. First, the image is cleaned, improving the image contrast and eliminating noise. Next, baseline and corpus line are identified and used to correct the line skew. Next, the slant is corrected, and the height of the characters is normalized based on baseline and corpus line. Finally, the input image size is adjusted to the model using a fixed height and width without modifying the aspect ratio of the text. Fig. 6.8 shows a schema of this algorithm's main steps.

This specific image pre-processing algorithm, which includes all the previous steps, has been applied to all the experiments performed, except for a specific one that measures the



Figure 6.8: Pre-processing algorithm steps.

impact of the normalization algorithm itself on the error. In addition, as in [46] we invert the images so that the traces are composed with the pixels of higher intensity on a black background of zero-valued pixels. It makes slightly easier to learn the activations of the CNN component. This inversion has been applied in all the models and results included in this Thesis.

The preprocessing performed is the same for the three databases used in the experiments. However, in the IAM database, it was applied at line-of-text level, and for the RIMES and Osborne databases, it is performed at word level. Fig. 6.9 shows one example of an original line of the IAM database and the result produced after the described normalization process.

MR. Macmillan	hus picked a strong
MR. Macmillan	has picked a strong

Figure 6.9: Line preprocessing example: (upper) original line and (lower) preprocessing results.

In the following paragraphs, we review this new normalization algorithm, detailing its use

in the processing carried out in this Thesis and outlined in Fig. 6.8.

Remove the noise of the image

The first step of the algorithm is to clean the image. The method proposed by Villegas et al. [214] has been used, which combines the advantages of local binarization methods and preservation of the grayscale. To do this, these authors modified Saviola algorithm [179] so that instead of binarizing each pixel based on the neighborhood, they convert it to a gray value through a linear transformation. An implementation of the algorithm is available at https://github.com/mauvilsa/imgtxtenh.

Fig. 6.10 shows two examples of applying this algorithm on one page of the IAM database and another page for the Osborne database.

Madrid Vi dengold Madrid Hellayold Mari MARA Willerundo e que labara me den is concor 11 comodarlo Chur carto de Canithe In a mande In a mande An the a ellet ertus que due à ellepo Ca forme Ud al favor a some la de far que una alpa que esta allina hora. tur heralo altima una. acarlo a ella 4 A/11 ella acolo a dento Edin to presenta Bredenta un alpic

Figure 6.10: Example of result of the enhanced contrast algorithm used in this Thesis applied on a Osborne database page.

Slope correction

The slope correction is carried out separately and before the slant correction described in the next section. For the experiments in this Thesis, we develop a new method similar to Gupta and Chanda [89], that uses a linear regression of the x, y positions of the pixels in the core region. We identify the slope angle from the estimate of the baseline obtained by applying the robust regression *RANdom SAmple Consensus* (RANSAC) [70] method on the vertical pixels that exceed a certain gray threshold and that are located in the lowest position of each pixel column. In Fig. 6.11 these pixels can be seen marked in red in a line of handwritten text from the RIMES database. The robust regression performs an elimination of outliers that mostly correspond to the extreme positions of the descenders.

of Radicalism, & the God knows what may happen Baseline of Radicalism, & the God knows what may happen Horizontal Baseline

Figure 6.11: Example of slope angle estimation used in this Thesis.

Slant correction

This Thesis develops a variant of the direct estimation methods provided in papers [228], [122] and [121], based on the analysis of 8-directional chain code of the thinned image. The text shear angle is estimated from the analysis of pixels larger than all right boundary pixels, i.e., black pixels with a white pixel on their right. For these pixels, the top three pixels, the upper-left, the upper-center, and the upper-right are checked so that evidence about the local inclination of the line to the left or the right is accumulated. The ratio between the difference of evidence to the right and the normalized evidence to the left provides an estimate of the slant angle tangent. The code is detailed in the algorithm 6.1.

Algorithm 6.1 Slant angle identification algorithm proposed.

```
def slant_angle(img, th_up=100, th_down=100):
    C = 0
    L = 0
    R = 0
    for w in range(1, img.shape[1]-1):
        for h in range(2,img.shape[0]-1):
            if img[h,w] > th_up and ( img[h, w+1] < th_down):
                if img[h-1, w-1] > th_up:
                     if img[h+1, w] > th_up:
                         L += 1
                         C += 1
                    if img[h+1, w+1] > th_up:
                         L += 2
                elif img[h-1, w] > th_up:
                    if img[h+1, w-1] > th_up:
                         R += 1
                         C += 1
                    if img[h+1, w] > th_up:
                         C += 2
                    if img[h+1, w+1] > th_up:
                         I. += 1
                         C += 1
                elif img[h-1, w+1] > th_up:
                     if img[h+1, w-1] > th_up:
                         R += 2
                    if
                       img[h+1, w] > th_up:
                         C += 1
                         R += 1
    return np.arctan2((R-L),(L+C+R))
```

Next, by knowing the slant angle α and the image width w, we corrected the slant using the affine transformation matrix defined by Eq. (6.8).

$$\left(\begin{array}{rrrr}
1 & -\alpha & 0.5 \cdot w \cdot \alpha \\
0 & 1 & 0
\end{array}\right)$$
(6.8)

The algorithm application implemented on two examples of line images can be seen in Fig. 6.12. The first one corresponding to the IAM database with a small slant, and the second one corresponding to the Osborne database with a very pronounced slant that makes it difficult to read the text even for humans. The transcription is "Estimado Madera".



Figure 6.12: Slant correction samples over lines of IAM and Osborne databases

Normalize ascenders and descenders regions

To normalize the size of the ascender and descender zones is necessary to identify them from the baseline and upperline lines. These lines have already been identified in the previous slope correction step 6.2.3, and the image has been transformed so that these lines are completely horizontal. However, the subsequent slant correction step may have modified them. Therefore, they are recalculated following the robust regression procedure described above but restricting that it must be a horizontal line with no slope angle. This constraint is easy to meet since the text slope has already been eliminated.

Once the horizontal lines of the baseline and upperline have been identified, the image is divided into three horizontal regions as shown in Fig. 2.6. A height rescaling is applied to the ascending and descending regions with a factor proportional to the ratio of the height of the region with respect to the height of the core region, as indicated in the equation 6.9.

$$r = \begin{cases} \frac{h_{cr}}{h_r}, & \text{if } h_r > h_{cr} \\ 1, & \text{otherwise} \end{cases}$$
(6.9)

Fig. 6.13 shows an example of a line from the IAM database on which the previous algorithm has been applied to normalize the ascending and descending regions.


Figure 6.13: Example of normalization of the ascenders and descenders regions.

Crop and resize input images

Finally, an image centering and resizing process is performed as follows. First, the left and right empty edges of each image are cut out, but the lower and upper edges are not changed. It is done to keep the relative vertical position of the baseline and upperline of the different images when this is possible. It happens when they are not already cropped in the original source, as is the case with Osborne's database. When the words top and bottom edges that have only ascenders or only descenders are removed, the relative positions of the baseline and the upperline are changed, sometimes significantly. Thus, a word-centered image that has only ascenders will have its core region at the bottom of the word, and, conversely, another word that has only descenders will have its core region at the top. Since we want to avoid this and keep the core region in the center of the image in all cases, no trimming of the top and bottom edges is done.

Secondly, the image is resized to a fixed height of 48 pixels, which has been visually checked to ensure that it does not alter text recognition by humans. In addition, empty columns up to a width of 192 pixels are added on the left. With this size of 48×192 pixels, it has been proven that it is possible to correctly transform more than 99.9% of the images in the databases used. For the few cases in which the word image is longer than 192 pixels after resizing, its size is adjusted to 192 by changing the aspect ratio of the word. In addition, we experiment with three different resolutions, 32×128 , 48×192 , and 64×256 , to evaluate the impact of this selection on the final result.

Fig. 6.14 shows several examples of images before and after the cropping and resizing described above. The images are printed, keeping the relative sizes between them.



Figure 6.14: Example of crop an resize word images: original at the left, and resized ones to fixed size at right.

6.2.4 Data augmentation

The high variability of HTR data, and the fact that existing databases for model development are limited in size, make difficult the training process of deep architecture models such as the one proposed. Therefore, we have experimented with applying data augmentation in training following the strategy of introducing the data augmentation process in each iteration. This integration makes it possible that a set of input data different from those of the previous epoch is obtained in each epoch of the model training algorithm. It allows performing the training with a theoretically infinite sample of data. Thus the training algorithm is defined next. It is the identical implementation of the data augmentation done in other HTR publications as, for example, in [170].

Algorithm 6.2 Train algorithm with data augmentation integrated.

```
def train(train_data, model, lr=learning_rate):
model.w = initialization()
for epoch in range(num_epochs):
    img_augmented = augmentation(train_data)
    y_pred = model.forward_pass(img_augmented)
    w_gradients = model.backward_pass(y, y_pred)
    model.w = model.w + (lr * w_gradients)
return model.w
```

The proposed algorithm performs random transformations of data that are invariant to the text content present in the image. The transformations used are: translations, resizings, slant modifications, elastic distortions, random projective transforms, and erode and dilate transformations.

The elastic distortion transformation is a technique based on the realization of small local distortions of the image. For this purpose it is defined a grid of points in the image, on which tiny random modifications of the points position are applied. The elastic distortion transformation is obtained by adapting the original image to the distorted grid of points. This transformation imitates the small oscillations in the stroke present in many handwritten calligraphies.

Random projective transform, also called homography, changes an image as if the point of view from which the image is viewed has been modified. Projective geometry is used to do this.

The erode and dilate transformations are another type of morphological transformations that allow to increase or decrease the size of the elements in the foreground of the image. The dilation operation adds points to the edges of the foreground image. If the foreground image is composed of handwritten text strokes, the size of these strokes is increased. In the case of dilation transfer, the thickness of the strokes will be reduced.

Fig. 6.15 show some examples of the word "room" augmented by the previous transformations.

The detailed algorithm is included in 6.3. The algorithm defines a level of randomness and an order of application of the above transformations. It places the smoothest transformations first, by assigning them a higher probability of application, and leaves for the last the most aggressive transformations, which are the erosion and dilation transformations with a lower probability of occurrence. For a given image, several transformations can be applied consecutively, producing a final composite transformation.



Figure 6.15: Examples of images generated by the data augmentation algorithm used in this Thesis.

Algorithm 6.3 Data augmentation algorithm proposed.

```
def augmentation(img):
if random_uniform(0, 1) < 0.5:
    img = ElasticDistortion(img)
if random_uniform(0, 1) < 0.5:
    img = RandomTransform(img)
# Skew
if random_uniform(0, 1) < 0.8 :
    angle = random_uniform(-0.15, 0.15)
    img = affineTransformation(img, angle)
#Resize
if random_uniform(0, 1) < 0.4:
    img = resize_down(img)
elif random_uniform(0, 1) < 0.4:
    img = resize_up(img)
#Erode - dilate
if random_uniform(0, 1) < 0.3:
    img = erode(img, iterations=1)
elif random_uniform(0, 1) < 0.3:
    img = dilate(img, iterations=1)
return img
```

6.2.5 Lexicons

For each considered handwriting word database (IAM, RIMES and Osborne, respectively), a standard lexicon is used in this chapter. It is selected considering that it is a lexicon mainly accepted in the literature or in its absence that it is easy to obtain or construct to facilitate the reproducibility and comparison of the results.

The application of each lexicon to decode the results is made by a systematic search of the lexicon word closest to the resulting string of characters in the visual model, in order to select the one that minimizes Levenshtein's distance [199].

In addition to the results obtained with the standard lexicons, validation metrics have been computed, by considering as a result the direct output of the visual model without any kind of subsequent decoding. It facilitates the direct comparison of different visual models, each one with each other.

Finally, results have also been provided using as lexicons the word sets present in the test partitions of the different corpora. Obviously, this does not provide a realistic result of the model's ability to recognize the text, but it allows the error impact produced by the OOV words to be measured when the standard lexicon is used. In addition, it provides an easy way to compare different models in the case of optimal decoding conditions.

Lexicons applied with the IAM database

The standard validation lexicon used was the set of 50,000 most frequent words in the union of Lancaster-Oslo/Bergen (LOB) [110] and the Brown [71] corpora when previously the IAM paragraphs from the Brown corpus were removed. Table 6.2 contains the complete list of excluded pages of the LOB corpus used in the IAM database.

A1	A2	A3	A4	A5	A6	-
B1	B2	B3	B4	B5	B6	-
C1	C2	C3	C4	-	C6	-
D1	-	D3	D4	D5	D6	D7
E1	E2	-	E4	-	E6	E7
F1	F2	F3	F4	-	-	F7
G1	G2	G3	G4	G5	G6	G7
H1	H2	-	H4	H5	H6	H7
J1	-	-	J4	-	J6	J7
K1	K2	K3	K4	-	-	$\mathbf{K7}$
L1	-	L3	L4	-	-	L7
M1	M2	M3	M4	-	M6	-
N1	N2	N3	N4	-	N6	-
P1	P2	P3	-	-	P6	-
-	R2	R3	-	-	$\mathbf{R6}$	-

Table 6.2: List of excluded pages of the LOB corpus used in the IAM database.

To identify the words in the previous corpora, we use a blank space tokenizer, and next, we isolate the punctuation marks because in the IAM database these characters are considered as single words.

Lexicons applied with the RIMES database

All the present words in the training and validation partitions are used as a standard lexicon in the RIMES database. It includes a total of 5,334 different words. Using the previous lexicon, the number of OOV words in the test partition is 410.

In this Thesis, for the RIMES database we use exactly the transcriptions of words provided with that database, differentiating between capital and small letters. However, other authors such Stuner et al. [199], or Menasri et al. [149] use only lowercase letters because they indicate that the labeling of certain uppercase characters, especially the letter 'j' (in the words 'je' or 'j') has errors.

Lexicons applied with the Osborne database

For the Osborne database, we also used all the words present in the train and validation partitions as a standard lexicon. In this case, we have a total of 2,708 different words, and the OOV words in the test set are 185.

6.3 Experiments

This section details the experimental results achieved with the proposed architecture for the offline continuous HTR problem. The experiments aim to explore this architecture capabilities and identify the components and parameterizations that provide the best results in terms of recognition capability.

The experiments have been repeated on three handwritten text databases of three different languages to validate the consistency of the experiment results. First, two well-known handwriting datasets were selected: IAM offline handwriting [146] in English, and RIMES databases [84] in French. These two databases have been extensively used by multiple authors and allow us to evaluate the proposed model compared to the results of other authors. In addition, we also provide results for the new Osborne database in Spanish. The Osborne database is especially challenging because it is a dataset of historical documents with very accentuated cursive handwriting and this dataset is smaller in size than the other two ones.

The handwritten text recognition ability in each experiment is measured using the metrics of Character Error Rate (CER) and Word Error Rate (WER) [72] metrics described in Section 2.5. In summary, CER measures the Levenshtein distance [138] between the predicted and the real character sequence of the word. WER represents the percentage of words correctly identified by the model and, therefore, defines the model accuracy. Since the goal of our experiments is to obtain conclusions about the performance of the visual model, WER and CER values provided in the experimental results are obtained by the direct application of the model, without using any lexicon in the decoding.

First, in Subsection 6.3.1, baseline results obtained using an initial configuration of the model selected, based on the results of the experiments carried out in the previous chapter (i.e., isolated character recognition and the most common configurations that this architecture usually has when applied to different problems) are presented [5] [106] [44]. Then, new experiments are carried out to explore the capabilities and limitations of the proposed architecture. The type of experiments performed are categorized as follows:

- Experiments about image segmentation strategy. To identify the impact of the patch width and the step size parameters of the image segmentation into the error. The results are detailed in subsections 6.3.2 and 6.3.3.
- Experiments for the different convolutional architectures analyzed in Chapter 5 regarding isolated character recognition. The results are detailed in Subsection 6.3.4.
- Experiments to select the most suitable encoder architecture in terms of the RNN type to be used, its dimension, and the number of layers. The results of these experiments are included in Subsection 6.3.5.
- Experiments on the impact of the input image resolution on the error, with detailed results in Subsection 6.3.6.
- Experiments on the importance of data normalization and data augmentation, detailed in Subsection 6.3.7.
- Experiment on the application of the teacher forcing strategy during training, with results in Subsection 6.3.8.

From the results of the different previous experiments, a more suitable model configuration is selected. The last experiment is performed with this configuration to obtain a final result of the proposed model that can be compared with the results of other authors on the same databases. These results are detailed in Subsection 6.3.9

In the case of the runs corresponding to the baseline and final models, these provide additional results using the selected lexicons in the decoding process. In particular, we offer three main results for each experiment:

- First, the results obtained by the direct application of the visual model (without any lexicon usage).
- Second, we process the outputs of the visual model by finding the closest word in terms of Levenshtein distance [138] over standard lexicons for IAM and RIMES databases, extensively used in the handwriting literature and described in Subsection 6.2.5.
- Third, we apply the same search using the test set lexicon.

Finally, we dedicate the final Subsection 6.3.10 to analyze the nature of the error by investigating the error types that we got in order to propose some possible improvements to the current model.

6.3.1 Baseline experiments

An initial base configuration of the model has been defined to obtain a reference result that will be used to contrast the subsequent results of the different experiments performed. The selection of the configuration is based on the criteria that this is not excessively complex and that, based on our experience, can obtain good a priori results on the proposed problem. This configuration is as follows:

- The strategy of converting the input image into a sequence of patches is selected as indicated in Subsection 6.1.2. The patch configuration is set up with an initial patch width of 10 pixels and a step size of 2 pixels.
- As convolutional architecture, LeNet applied on the patch sequence has been selected.
- The encoder is configured with two 256 LSTM layers.
- The height of the input size is set to 48
- The experiments are performed on the normalized data without any data augmentation in training and using the teacher forcing strategy.

With this configuration, experiments have been performed with the IAM, RIMES, and Osborne databases, and the following tables detail the results obtained for each of the decodings by lexicons used in all the experiments performed.

Table 6.3 includes CER and WER data for the validation and test partitions corresponding to the results obtained directly by the model, without using any lexicon to decode them.

Dataset	Validation		Test	
	CER	WER	 CER	WER
IAM	8.5	19.3	12.6	27.4
RIMES	6.0	18.0	6.1	17.8
Osborne	16.3	33.5	20.9	44.0

Table 6.3 \cdot	Baseline	results	with	no	lexicon
Table 0.9.	Dascinic	resurus	VV 1 U 1 1	110	ICAICOII.

If the standard lexicons to decode the results of the visual model are used, we obtain CER and WER data detailed in Table 6.4.

Dataset	Validation		Test	
	CER	WER	CER	WER
IAM	10.6	17.9	14.3	24.5
RIMES	7.0	14.2	6.9	14.3
Osborne	13.1	21.9	28.2	42.8

Table 6.4: Baseline results using standard lexicon.

Finally, the results in Table 6.5 are obtained by using the test lexicon to decode the model predictions. In this case, it only makes sense to apply this lexicon to the test data.

6.3.2 Image segmentation: step size

In this first set of experiments, we analyzed the effect over the metrics of CER and WER of the different parametrizations corresponding to the initial process of image patches

Dataset	Validation		Test	Test		
	CER	WER	CER	WER		
IAM	NA	NA	10.3	17.0		
RIMES	NA	NA	3.6	7.2		
Osborne	NA	NA	8.2	13.9		

Table 6.5: Baseline results using the test lexicon.

creation described in Subsection 6.1.2. We considered two main parameters: the patch width and the step size to create the patches. For example, a patch width of 5 and a step size of 2 means that the first patch is created with the columns from 1 to 5, the second patch with the columns from 3 to 7, and so on.

We started analyzing the step size. Training is performed on the baseline model by modifying only the step size parameter with the following values: 1, 2 (baseline), 4, 6, and 8. It is recalled that in the baseline model, the patch width is 10.

Table 6.6 shows the results of the experiments with the step size. It is observed that the results for step sizes 1, 2, and 4 are pretty similar in each database. From step sizes greater than 4, the error levels measured in terms of WER and CER increase with increasing the step size.

Dataset		Valida	tion	Test	
	Step size	CER	WER	CER	WER
	1	8.6	20.0	12.6	27.4
	2	8.5	19.3	12.6	27.4
IAM	4	8.8	20.3	12.9	28.0
	6	9.3	22.1	13.6	29.5
	8	10.2	23.2	14.9	31.1
	1	6.1	17.7	6.1	17.9
	2	6.0	18.0	6.1	$17,\!8$
RIMES	4	6.4	18.2	6.3	18.0
	6	6.6	19.0	6,4	18.4
	8	7.5	20.3	7.4	20.0
	1	17.6	36.1	23.2	46.4
	2	16.3	33.5	20.9	44.0
Osborne	4	16.4	35.1	21.0	43.5
	6	18.7	37.6	23.6	46.0
	8	21.3	41.0	27.4	51.7

Table 6.6: Step size experiments results (baseline in step size of 2). Best result, by database, is presented in bold.

The relative difference of using different step sizes can be seen in Fig. 6.16. It shows the relative increase or decrease of WER and CER concerning the baseline model, which in the graph shows the value 100.



Figure 6.16: Variation percentage of WER and CER in validation and test partitions for different step sizes over the baseline.

We observe the increasing trend of the error as a function of the step size, with the only exception of the step size of 1 applied to the Osborne database.

6.3.3 Image segmentation: patch width

The patch width, i.e., the width of each of the patches extracted from the image, is now analyzed. The patches are overlapped unless the step size and patch width coincide. Because we used a convolutional architecture over the patches, the patch width must have a minimal size to apply the convolutions. This minimal width is 5 because in LeNet architecture the initial convolutional layer has a filter of 5×5 . The baseline model uses a patch width of 10 with a step size of 2 applied on a LeNet architecture. The experiments performed analyze patch widths between 5 and 25 in intervals of 5.

Table 6.7 shows the results obtained in the experiments performed to analyze the effect of patch width on the error.

Table 6.7 shows that the use of different patch widths affects the model results relatively little. In terms of WER metric in validation, which is used in the training stop criterion, the minimum value is obtained for a patch width of 15. In any case, the results with patch widths between 5 and 20 are similar, and only with a patch width of 25, it starts to be seen an increase in the error metrics that is consistent between databases.

Fig. 6.17 shows the relative difference of using different patch widths. It presents the relative percentage increase or decrease of WER and CER concerning the baseline model, which corresponds to a patch width of 10, and which in the graph takes the reference value of 100.

It is consistently observed in the three databases that the patch width of 15 gives the best results and that smaller or larger patch widths increase the error. In any case, the relative differences are small with patch widths between 5 and 20.

Dataset		Valida	ation	Test	
	Patch width	CER	WER	CER	WER
	5	8.3	19.8	12.7	27.7
	10	8.5	19.3	12.6	27.4
IAM	15	8.5	19.6	12.5	27.3
	20	8.9	20.4	12.9	28.0
	25	9.3	21.2	13.4	28.7
	5	6.3	18.5	6.2	18.1
	10	6.0	18.0	6.1	$17,\!8$
RIMES	15	5.9	17.3	5.7	16.8
	20	6.4	18.8	6.2	18.4
	25	6.3	18.5	6.2	18.3
	5	17.7	35.1	24.2	46.0
	10	16.3	33.5	20.9	44.0
Osborne	15	15.9	32.0	19.9	43.8
	20	16.5	34.2	19.5	42.6
	25	19.2	38.2	23.6	48.3

Table 6.7: Patch width experiments results (baseline in patch width of 10). Best result by database in bold.



Figure 6.17: Variation percentage of WER and CER in validation and test partitions with different patch widths over the baseline.

6.3.4 Convolutional architectures

The different convolutional architectures proposed are discussed below. The two selected architectures are LeNet and VGG. In addition, each convolutional architecture can be used with an image patching strategy or directly on the model input image. In this way, we experiment with four possible configurations regarding convolutional architecture:

• LeNet architecture on image patches.

- VGG architecture on image patches.
- LeNet architecture on full image.
- VGG architecture on full image.

The results of these four experiments for each of the three considered databases are presented in Table 6.8.

Dataset			Valida	ntion	Test	
	Patches	Architecture	CER	WER	CER	WER
	Yes	LeNet	8.5	19.3	12.6	27.4
IAM	Yes	VGG	9.1	20.6	13.3	28.6
	No	LeNet	8.5	19.9	12.4	27.1
	No	VGG	8.3	19.6	12.0	26.6
	Yes	LeNet	6.0	18.0	6.1	$17,\!8$
RIMES	Yes	VGG	6.1	18.2	5.7	17.7
	No	LeNet	5.7	16.5	5.5	16.2
	No	VGG	5.4	16.6	5.4	16.3
	Yes	LeNet	16.3	33.5	20.9	44.0
Osborne	Yes	VGG	36.2	47.2	38.3	54.8
	No	LeNet	17.1	35.2	19.9	42.1
	No	VGG	NA	NA	NA	NA

Table 6.8: Convolutional architectures experiments results (baseline in Lenet and patches). Best result by database in bold.

The behavior of the model depending on the convolutional architectures used is quite different for each database. With IAM database, the results are similar using patches and full images, obtaining a lower WER in validation with LeNet architecture applied on patches. In the case of RIMES database, it is clearly observed that the error decreases when using the convolutional component directly on the image instead of on the patches. In the case of the Osborne database, the VGG architecture does not work correctly. In the case of using it on patches, the error increases notably, and when trying to apply it to the whole image, the training of the model does not converge to an optimal solution.

The graph in Fig. 6.18 shows the relative differences between the convolutional architectures proposed. The figure shows the relative increase or decrease percentage of WER and CER concerning the baseline model that corresponds to a LeNet architecture applied on patches that in the graph takes the reference value 100.

It is observed that, in general, the relative differences are less than 5%, except for the results of the full-image experiments performed on RIMES database. It is not easy to theorize on a possible cause of this behavior. It is proposed as a hypothesis that it could be related to the fact that the average word length in RIMES database is shorter than in the other two databases, and that may favor models applied on the whole image rather than on patches. This hypothesis should be tested in future work.



Figure 6.18: Variation percentage of WER and CER in validation and test partitions with different convolutional architectures over the baseline.

6.3.5 Encoder architecture

Next, we analyzed the effect of the encoder architecture on the final error. Four encoder characteristics were analyzed and hypothesized to be relevant for the model performance. The following characteristics have been analyzed by taking into account that in the base-line model, the encoder is defined as a bidirectional LSTM-type RNN with two 256 layers:

- First, the size of the RNN layers is analyzed, experimenting with values of 64, 118, 256, and 512.
- Second, the number of RNN layers is analyzed, considering both the number of layers and whether or not they are bidirectional. We experiment with values of 1, 2, and 3 layers for both the bidirectional and unidirectional cases.
- Finally, the RNN type is analyzed, experimenting with LSTM and GRU types.

RNN dimension

In the first experiment, we used a fixed baseline model with a two-layer bidirectional LSTM-type RNN encoder and built different experiments varying the size of the RNN layers. The results of the experiments for each of the three databases are included in Table 6.9.

It is observed that in the cases of IAM and RIMES databases, the experiments that set a large encoder size obtain better results. For Osborne database, the model with encoder size 512 does not converge properly and obtains a very high error. It is probably because IAM and RIMES databases are much more complex in size than Osborne database, and therefore the larger encoder sizes work better for these databases and worse for Osborne database.

In Fig. 6.19 we measure the percentage increase or decrease concerning the baseline of the validation CER and WER values. It allows to better observe the effect of the dimension change without being affected by differences in the complexity of the different databases.

Dataset		Valida	Validation		
	Encoder size	CER	WER	CER	WER
	64	9.7	22.8	13.9	30.3
IAM	128	8.3	19.6	12.8	27.9
	256	8.5	19.3	12.6	27.4
	512	8.0	18.6	11.7	26.0
	64	7.6	22.3	7.3	21.8
RIMES	128	6.3	18.1	6.0	17.6
	256	6.0	18.0	6.1	$17,\!8$
	512	7.8	17.6	8.0	17.4
	64	20.0	38.0	25.0	51.1
Osborne	128	15.4	33.7	21.5	46.7
	256	16.3	33.5	20.9	44.0
	512	51.2	56.1	54.1	63.3

CHAPTER 6. DEEP ARCHITECTURES APPLIED TO CONTINUOUS OFFLINE HANDWRITING RECOGNITION

Table 6.9: Encoder size experiments results (baseline in encoder size of 256). Best result by database in bold.



Figure 6.19: Variation percentage of WER and CER in validation and test partitions with different encoder RNN sizes over the baseline.

From Fig. 6.19 it is clear that the 64 size is not suitable for any of the databases. In terms of WER in the validation data, the best results are obtained with size of 512 for IAM and RIMES, and size of the 256 for Osborne. The results obtained with a size of 128 are relatively close to these optimal with a much smaller number of parameters. As a conclusion, it is proposed that a number of parameters between 128 and 512 are suitable for the large databases (IAM and RIMES), and between 128 and 256 for the smaller Osborne database.

Number of layers and bidirectionally

In the second set of experiments, we used a fixed baseline model with an RNN encoder of type bidirectional LSTM with dimension 256, and defined different experiments modifying the number of layers of the RNN and the fact that these layers are bidirectional or not. The results of the experiments for each of the three databases are included in Table 6.10. This Table shows that the best validation results are obtained with three bidirectional layers for IAM and RIMES databases and two bidirectional layers for Osborne database. As before, it is hypothesized that the models for the larger and more complex databases improve by increasing the decoder complexity, in this case by increasing the number of layers. It is also observed that models with bidirectional layers generally perform better than models with unidirectional layers. This is true regardless of the database and the number of layers.

Dataset			Valida	ation	Test	
	Bidirectional	num layers	CER	WER	CER	WER
	No	1	9.6	22.3	14.1	30.0
	No	2	9.1	20.7	13.2	28.3
IAM	No	3	9.3	20.7	13.2	28.0
	Yes	1	8.8	21.1	13.1	28.9
	Yes	2	8.5	19.3	12.6	27.4
	Yes	3	8.0	18.8	11.6	25.8
	No	1	7.7	21.1	7.5	21.2
	No	2	6.7	18.4	6.7	18.3
RIMES	No	3	6.4	17.3	6.2	17.3
	Yes	1	7.1	20.0	6.7	19.9
	Yes	2	6.0	18.0	6.1	$17,\!8$
	Yes	3	5.1	15.4	5.1	15.5
	No	1	23.8	43.8	27.9	51.1
	No	2	25.0	43.0	32.8	50.0
Osborne	No	3	27.5	41.9	31.3	52.5
	Yes	1	21.2	41.1	28.9	54.5
	Yes	2	16.3	33.5	20.9	44.0
	Yes	3	16.6	34.3	18.9	40.7

Table 6.10: Number of layers and bidirectionally experiment results (baseline in 2 bidirectional layers). Best result for each database in bold.

Fig. 6.20 presents the percentage increases or decreases concerning the baseline of the validation CER and WER values. It allows better observing the effect of changing the number and type of encoder layers without being affected by differences in the complexity of the different databases.

It is observed that, in general, WER tends to decrease as the number of layers increases, except that for Osborne database. The results with three layers are not good, and the optimum is achieved with two bidirectional layers. The conclusion is similar to the one obtained when analyzing each layer size: that the models of the more complex databases



Figure 6.20: Variation percentage of WER and CER in validation and test partitions with different encoder layers over the baseline.

benefit from an increase in the number of encoder parameters. It can be seen that increasing the encoder complexity by adding layers is more beneficial than increasing the encoder complexity by increasing the size of each layer when these data are compared with those obtained previously by experimenting with the size of the encoder layers, shown in Fig. 6.19. In RIMES database, by increasing the layer size from 256 to 512 means a WER improvement of less than 5%. However, for the same database, going from two bidirectional layers to three layers means an improvement of more than 15%.

Type of layers: LSTM vs GRU

The third set of experiments analyzes the impact on the error when changing the encoder RNN type, experimenting with LSTM and GRU types. In order not to draw conclusions about the type of architecture based on a single experiment per database, each type has been analyzed in the bidirectional and non-bidirectional cases. Table 6.11 compares the results of this experiment for each of the three databases. Finally, it is recalled that, as indicated in Subsection 6.1.4, for the model architecture proposed for this Thesis, the decoder has been chosen to use the same type of RNN as the encoder.

Table 6.11 shows that the best results for IAM and RIMES databases are obtained with the GRU type. In Osborne database, the results with GRU type networks are inferior to those obtained with bidirectional LTSM type networks. In the case of RIMES database, the improvement of results by using the GRU type appears with both unidirectional and bidirectional layers.

Fig. 6.21 measures the percentage increases or decreases for the baseline of the validation CER and WER values when changing the RNN type of the encoder and decoder. It allows to better observe the effect of the dimension change without being affected by the differences in the complexity of the different databases. This figure also shows how RIMES database is the most affected by the RNN type change. We have omitted the data from Osborne database, which had a huge percentage of change not to distort the graph.

Dataset	aset		Validation		Test	
	Bidirectional	Type	CER	WER	CER	WER
	No	LSTM	9.1	20.7	13.2	28.3
IAM	No	GRU	9.0	20.6	12.9	27.8
	Yes	LSTM	8.5	19.3	12.6	27.4
	Yes	GRU	8.0	18.9	11.6	26.0
	No	LSTM	6.7	18.4	6.7	18.3
RIMES	No	GRU	5.1	15.5	5.1	15.6
	Yes	LSTM	6.0	18.0	6.1	$17,\!8$
	Yes	GRU	5.3	16.5	5.3	16.3
	No	LSTM	25.0	43.0	32.8	50.0
Osborne	No	GRU	27.1	43.1	32.2	51.9
	Yes	LSTM	16.3	33.5	20.9	44.0
	Yes	GRU	41.2	52.5	44.9	59.3

Table 6.11: RNN encoder type experiments results (baseline in LSTM bidirectional). Best result by database in bold.



Figure 6.21: Variation percentage of WER and CER in validation and test partitions with different encoder types over the baseline.

In summary, IAM and RIMES databases benefit from an increase in encoder complexity greater than that established in the baseline. In addition, among the aspects analyzed, the most influential one in obtaining better WER and CER results in validation are increasing the number of layers from two to three and using GRU networks instead of LSTM type. In the case of Osborne database, the results indicate that the architecture proposed in the baseline would be the most appropriate one.

6.3.6 Image resolution

Experiments were also conducted to determine if the input image resolution affected the model performance. For this purpose, three different resolutions (height×width) were

selected for the input images of 32×128 , 48×196 (baseline), and 64×256 pixels. The changes in image sizes have been made proportionally to keep the aspect ratio of images. The results of these experiments for each of the three databases are included in Table 6.12.

Dataset	taset		ation	Test	
	Image Resolution	CER	WER	CER	WER
IAM	32x128	9.1	20.9	13.5	28.8
	48x192	8.5	19.3	12.6	27.4
	64x256	8.4	20.0	12.5	27.3
RIMES	32x128	6.5	18.3	6.4	18.1
	48x192	6.0	18.0	6.1	17,8
	64x256	6.3	18.2	6.2	18.1
Osborne	32x128	19.9	38.5	25.4	50.0
	48x192	16.3	33.5	20.9	44.0
	64x256	18.3	37.1	22.4	45.4

Table 6.12: Input image resolution experiments results (baseline in 48x192). Best result by database in bold.

It is observed that in all cases, the lowest WER in validation is provided by the 48×192 resolution. The database least affected by resolution changes is RIMES, and the most affected is Osborne. In general, by reducing the resolution to 32×128 increases the error more than increasing it with 64×256 resolution.

The graph in Fig. 6.22 shows the relative differences with respect to the baseline which defined an input resolution of 48×192 .



Figure 6.22: Variation percentage of WER and CER in validation and test partitions with different input image resolutions over the baseline.

As shown in Fig. 6.22, in IAM and RIMES databases the relative differences are less than 10%, especially when increasing the resolution with respect to the baseline. Therefore, it

is concluded that working with a resolution greater than or equal to 48×192 is a factor that has a relatively small impact on the final results.

6.3.7 Normalization and data augmentation

The following experiments analyze the impact of normalization and train data augmentation strategy on WER and CER metrics. Experiments are performed with original and normalized data. The normalization strategy is the one selected in the baseline. The normalization process performed with the images of the three databases is described in the previous Subsection 6.2.3 corresponding to Image Preprocessing. For each of the above cases, two experiments are performed, the first one with the data augmentation strategy activated and the second one without train data augmentation (baseline). The train data augmentation strategy used is described in the previous Subsection 6.2.4. The results of these four experiments for each of the three databases are included in Table 6.13. These results clearly show that both data normalization and data augmentation strategies provide a significant reduction of WER and CER in all three databases. In particular, it is the combination of the two activated strategies that provides a lower WER and CER in validation and test. It is also observed that the data augmentation strategy provides a greater improvement than the normalization strategy. For example, activating data augmentation on non-normalized data reduces the CER by 1.7 points and the WER by 4.2 points for IAM database. On the other hand, by applying normalization without data augmentation improves the CER by 1.5 and the WER by 3.8.

Dataset			Valida	ntion	Test	
	Normalization	Data augmentation	CER	WER	CER	WER
	Yes	Yes	7.3	17.0	10.1	22.6
IAM	Yes	No	8.5	19.3	12.6	27.4
	No	Yes	8.3	18.9	12.1	26.0
	No	No	10.0	23.1	15.2	31.3
	Yes	Yes	4.7	13.7	4.6	13.5
RIMES	Yes	No	6.0	18.0	6.1	$17,\!8$
	No	Yes	5.4	14.9	5.1	14.5
	No	No	6.7	18.3	6.5	17.8
	Yes	Yes	13.6	29.3	15.1	35.9
Osborne	Yes	No	16.3	33.5	20.9	44.0
	No	Yes	21.1	39.2	20.7	44.7
	No	No	29.3	48.2	28.4	51.2

Table 6.13: Normalization and data augmentation experiments results (baseline with normalization and no data augmentation). Best result by database in bold.

The graph in Fig. 6.23 shows the relative differences with respect to the baseline configuration that used normalization and did not use data augmentation.

When analyzing the relative differences in Fig.6.23, it is observed that these differences are important, especially for Osborne database. With a more limited size, this database



Figure 6.23: Variation percentage of WER and CER in validation and test partitions with normalization and data augmentation over the baseline.

takes more advantage of applying data augmentation in training. In the case of Osborne database, the effect of normalization also is greater. This is because this historical database has a very pronounced slant which, if not corrected by normalization, is likely to make text recognition very difficult.

6.3.8 Teacher forcing

Finally, experiments have been conducted to determine the importance of using the *teacher forcing* strategy during model training. As indicated in Subsection 4.8.3, teacher forcing [224] consists of train the seq2seq model by passing as input to the decoder the original values of the output sequence, but shifted to t-1. This training strategy can lead to problems in the evaluation phase because an error of one of the generation steps of the output sequence is propagated through the conditional probabilities. It causes a decrease in performance because the predicted sequence diverges from the real one from the point of error. Teacher forcing also has as negative effect the train overfit because, at each step t of the decoding process, the previous real value of the character sequence y_{t-1} is used. Table 6.14 shows the results of these experiments for each of the three databases.

It can be seen that the best WER validation results are obtained with the teacher forcing strategy activated, although the results are quite similar, especially in the cases of IAM and RIMES databases. Fig. 6.24 shows the relative differences per database.

In IAM and RIMES databases, the relative differences in WER and CER are less than 2%, so using or not this training strategy is not relevant in terms of accuracy. It has been observed that not using teacher forcing increases by approximately 50% the number of epochs that the algorithm needs to reach the defined stopping criterion. It indicates that this strategy significantly accelerates the convergence speed of the model training.

Dataset		Validation		Test	
	Teacher Forcing	CER	WER	CER	WER
IAM	Yes No	8.5 8.2	19.3 19.5	12.6 12.3	27.4 27.2
RIMES	Yes No	$\begin{array}{c} 6.0 \\ 6.0 \end{array}$	18.0 18.3	6.1 5.7	17,8 17.5
Osborne	Yes No	16.3 21.8	33.5 41.8	20.9 25.1	44.0 50.5

Table 6.14: Input image resolution experiments results (baseline with teacher forcing). Best result by database in bold.



Figure 6.24: Variation percentage of WER and CER in validation and test partitions with different training strategy over the baseline.

6.3.9 Final model

It has been observed in the experiments performed that the variations in model performance with the different configurations analyzed are relatively consistent between databases, especially between the IAM and RIMES databases. These experiments also allow identifying modifications on the baseline configuration that provide better results in terms of accuracy. From here, guidelines can be obtained to propose an optimized parameterization of the proposed model. The selected configuration does not necessarily include the settings that have provided the lowest validation error. It was decided to choose the configuration by defining a model with the smallest number of parameters, following the principle of parsimony [212], since we have two configurations that provide similar validation results. These configurations are the following ones:

• As a convolutional architecture, LeNet applied on the patch sequence is selected (as indicated in Subsection 6.1.2. The patch configuration is set with an initial patch size of 10 pixels and a step size of 2 pixels. In this case, the selection matches the one made in the baseline.

- The encoder is configured with three bidirectional layers of type GRU of size 256 for IAM and RIMES databases. For Osborne database, the encoder is configured with two bidirectional layers of type LSTM of size 256. The RNN of the decoder is set to the same type and size as used in the encoder.
- The size of the input image is set to 48×192 .
- The trainings are performed on the normalized data, with train data augmentation and teacher forcing strategy.

With this optimized configuration, experiments have been performed with IAM, RIMES, and Osborne databases, and the following tables detail the results obtained for each of the lexicon decodings used.

Table 6.15 includes the CER and WER values for the validation and test partitions, corresponding to the results obtained directly by the model, without using any lexicon to decode them.

Dataset	Valida	tion	Test	
	CER	WER	CER	WER
IAM	6.7	15.1	9.3	21.0
RIMES	4.6	13.1	4.5	13.2
Osborne	13.6	29.3	15.1	35.9

Table 6.15: Optimal results without lexicon.

In the case of using the standard lexicons to decode the results of the visual model, we obtain the CER and WER values detailed in Table 6.16.

Dataset	Valida	tion	Test	
	CER	WER	CER	WER
IAM	8.8	14.9	11.2	19.6
RIMES	6.2	12.3	5.9	12.2
Osborne	8.7	14.7	24.0	34.5

Table 6.16: Optimal results using a standard lexicon.

Finally, by using the test lexicon to decode the model predictions, the results are obtained from Table 6.17. It only makes sense to apply this lexicon to test data. It is possible to evaluate the error weight of the words of the test set not present in the standard lexicon (i.e., the OOV words) by comparing the results of this table with the previous one.

6.3.10 Error analysis

This subsection analyzes the errors of the models presented to identify possible lines of work for improvement. The first analysis was performed in the previous subsection by studying the influence of the OOV words in the error. These errors can be reduced by

Dataset	Valida	tion	Test	
	CER	WER	CER	WER
IAM	NA	NA	7.6	13.0
RIMES	NA	NA	2.7	5.4
Osborne	NA	NA	9.4	14.4

Table 6.17: Optimal results using test lexicon.

improving the lexicons and, especially, by adding to the system some language models at character level that can identify OOV words.

First, we analyze how the errors change in terms of word length. Fig. 6.25 shows the WER with the standard lexicons in IAM and RIMES databases regarding word length. We observe that the error increases with the word length. It is related to the previous analysis on the OOV influence in the errors because the words with fewer letters are articles, prepositions, and similar words that typically appear in the lexicons. Regarding errors in identifying words of length one (isolated characters), in the RIMES and Osborne databases, the number of cases is residual (5 errors in RIMES and 6 in Osborne). There is a large volume of isolated characters in the IAM database due to the separation into words of punctuation marks, including apostrophes, and 641 errors appear, which represents 11.8% of the total errors. The most common errors consist of confusing the characters '.' and ',' and the characters "' and "''.



Figure 6.25: Error analysis: WER in test vs word length.

To eliminate the influence of the OOV words in the analysis of the error versus the word length, we show in Fig. 6.26 the WER test error using the test lexicon. We observed that most of the errors in long words originate from the absence of the correct word in the lexicon. Now, the highest percent of errors appears for word length of 4 in the IAM database and word length of 5 in the RIMES database.

Next, we analyze the amount of errors originated by replacing only one character in a word. The real and predicted words have the same characters and the same length, but one character has been replaced by another in the predicted word. In this case, we observed that the 57.4% of errors of the visual model in the IAM database, the 59.1% in



Figure 6.26: Error analysis: WER in test vs word length with the test lexicon.

the RIMES database, and the 56.9% in the Osborne database are errors corresponding to only one character replaced. Probably, these errors could be solved in most cases using a character-level language model.

Finally, we analyze the errors originated by the confusion between the same character, one in lowercase and the other in uppercase. In this situation, we observe that a 3.7% in the IAM database, a 5.6% in the RIMES database and, a 0.9% in the Osborne database are errors originated by the confusion of some uppercase character and the same one in lowercase, or vice versa. As we expected, the characters where it this confusion appears are those with a similar appearance in uppercase and lowercase. For example: 'f', 'c', 'v', 'l', 'k' or 'w'. These errors happen almost always at the beginning of words because they correspond to the beginning of a paragraph. They can be solved by finding a way to detect the end of a paragraph; for example, by identifying that the previous character is a '.' (dot).

6.4 On the difficulties of comparing results

While there is a reasonably widespread consensus about datasets and reference metrics for comparing different approaches to the general handwriting recognition problem, several factors influence the final results and which the authors address in different ways. So it is very difficult to compare the different approaches with each other.

These factors are fundamentally related to the choice of data for training, validation, and testing, the set of valid characters considered, the use of training and assessment improvement strategies, and the use of different lexicons and language models in decoding.

Some of these factors directly prevent the comparison of results. It is not possible to compare the metrics of two outcomes that have been calculated on different evaluation datasets. It is also impossible to compare results that have eliminated certain characters, such as punctuation marks or capital letters in the decoding process.

Other factors, such as the use of data augmentation in training or testing or the use of different language models in decoding, do not prevent the direct comparison of the final

results of different publications. However, in these cases, it is essential to provide an ablation study that allows us to detail how much the different improvement strategies employed contribute to the accuracy level. It allows a more exact comparison with other authors who have not used these improvement strategies.

Next, these factors are reviewed in detail, indicating how they influence the final result and proposing recommendations that allow a comparison between different publications results that provide the maximum information about the efficiency of the different components existing in a handwritten text recognition algorithm.

6.4.1 Choice of train, validation and test partitions

The use of the same training, validation, and evaluation or test partitions of the reference datasets in the different investigations on any modeling problem is a fundamental aspect where there is a general consensus. The training partition is used in the parameter setting of the different experiments, the validation partition allows to compare the different experiments in terms of accuracy or error metrics to select the optimal parameterization. The evaluation or test partition allows to calculate the metrics values on the selected optimal parameterization experiment. In this sense, the test partition must contain precisely the same data along the different lines of research so that the comparison of metrics between different authors can be reliable.

However, in the case of the general handwritten text recognition problem, there are exceptions to the above, in one of the most widely used databases to develop and compare experimental results. In the IAM database [146] coexist, over time, two different partitions of the train, validation, and evaluation datasets, which are widely used by different authors that obtain results with one of the partition but compare them with other publications results obtained for the other partition.

The first partition (hereafter official) is provided on the official site of the database ¹ and separates the total of 13,353 database lines into 6,161 for train, 940 for validation (named validation 2 in the site), and 1,861 for test. Additionally, it provides a second validation partition with 900 lines (named validation 1 in the site) and excludes from the partitions a total of 3,491 lines.

Several authors have used the second partition (usually referred to in the literature as Aachen partition) from 2013 to the present. To the best of our knowledge, it was first used by Kozielski et al. [126] and include a total of 6,482 lines to train, 976 for validation and 2,915 for test. This partition excludes a total of 2,980 lines.

Table 6.18, we provide a cross table that includes the number of common lines in both partitions. It can be seen that all training lines of the official partition are included in the training set of the Aachen partition. However, in the validation and test partitions, there is no match between the two partitions and the data are mixed. It implies that it is not a good practice the comparison of results obtained with the two partitions. Regarding the excluded lines, there is overlap except that the Aachen partition has incorporated lines excluded by the official partition into its partitions. In particular, it has included 321 lines to its train partition, 56 to validation, and 134 to eval or test.

¹http://www.fki.inf.unibe.ch/databases/iam-handwriting-database Last access: April 2021

	Aachen			
official	excluded	Train	Test	Validation
excluded	2,980	321	134	56
train	0	6,161	0	0
test	0	0	1,313	548
validation	0	0	709	231
validation1	0	0	759	141

Table 6.18: Cross table of lines count between Official and Aachen partitions of IAM database.

The main difficulty lies in the fact that it is not evident which partition is used by which author, as it is rarely explicitly stated in the publication. The principal indicator to identify it is the line or word counts per partition that are usually included in the description sections of the IAM datasets of each of the publications. These counts almost always coincide with the counts of one or another partition. Analyzing the data for this publication, it can be identified, in most cases, which partition has been used.

Two additional particular cases can be observed analyzing this data. Firstly, it can be seen that in the publications before 2011, that claim to use the official partition that accompanies the IAM database, such as [81] or [69], declare different line counts in the validation and test partitions from those currently available. In particular, declare 920 lines for validation instead of 940, and 2,781 lines for test instead of 1,861. It is probably due to a modification of the official partition after 2010. Secondly, several authors like Puigcerver [170] and Ingle et al. [104] declare to use a number of lines for the train partition that matches the official partition count, but in validation and test they declare a number of lines corresponding to the Aachen partition. In this case, and since the matching of the test partition is the key factor for comparing results, the outcome of these publications can establish valid comparisons with other results using the Aachen partition, even if they use the official training partition.

The use of one or the other partition impacts the results in terms of CER and WER in the validation and test partitions. Table 6.19 includes a comparison of the results obtained using the two partitions on the model proposed in this Thesis, with the parameterization defined in the previous section, corresponding to the final model selected. CER and WER results are provided for the cases of not using any lexicon, using the standard lexicon, and using the test lexicon of the IAM database.

It is observed that in the test partition, the results at the CER level are pretty similar. However, the results at the WER level, the Official partition obtains approximately one point lower WER than the Aachen partition consistently to use different decoding lexicons. It is also observed that differences in results between validation and test of the official partition are much smaller than those of the Aachen partition.

Since the test data of the two partitions are not the same, and as shown in the Table 6.19 the results differ, it is recommended that a publication using one of the partitions to be compared with publications using the same partition. It has not always been the case; there are several cases in which a publication has performed the experiments with one partition and compared their results with publications that have used the other partition. Below are some cases that have been identified in the literature:

Lexicon		Validation		Test	
	IAM Partition	CER	WER	CER	WER
No lexicon	Official Aachen	$7.4 \\ 6.7$	$17.8 \\ 15.1$	9.4 9.3	$19.8 \\ 21.0$
Standard lexicon	Official Aachen	9.4 8.8	$\begin{array}{c} 16.8\\ 14.9\end{array}$	11.3 11.2	$\begin{array}{c} 18.7 \\ 19.6 \end{array}$
Test Lexicon	Official Aachen	NA NA	NA NA	$7.4 \\ 7.6$	$12.0 \\ 13.0$

Table 6.19: Results comparative between the Official and Aachen IAM partitions.

- Publications [126] of 2013, and [164], both use the Aachen partition (with 2915 lines in test) and compares they results with [81] of 2009 and [69] of 2010 that use the older version of the Official partition (with 2781 lines in test). To our understand, the first usage of the Aachen partition in a publication is in [126].
- Publication [64] use the Aachen partition and compares its results with, among others, [69] and [66], which use the Official partition.
- Publication [169] use the standard split that comes with the database (Official partition) and compares its results with, among others, [126], [164], [64] and [26], which use the Aachen partition.
- Publication [128] use the Official partition and compares its results with, among others, [64] and [26], which use the Aachen partition.
- Publication [68] use the Official partition and compares its results with, among others, [170] and [164], which use the test Aachen partition.

The above cases do not modify the valuable contributions of each paper, but they do make it difficult to interpret the tables of comparative results provided in these publications.

In the case of the RIMES database, there are also two partitions used by different authors. The partition corresponding to the ICDAR2009 competition (with 51,739 words for train and 7,464 words for test), and the one corresponding to the ICDAR2011 competition (the most common). It uses the same train set as the previous partition (51,739 words), converts the previous test set to validation (7,464 words), and adds a new test set with 7,776 new words. The test set is different in both partitions, so it is considered that the comparison of results obtained with these two partitions is not valid. In this case, the two partitions have different names and are well differentiated in the literature.

6.4.2 Filter transcription errors in IAM database

The IAM database provides images and transcripts at different levels, including line and word level. Word level transcripts include a column indicating whether the transcript 'can be bad' or is correct. This transcription error mark is defined at the line level. If a line contains an inadequately segmented word, all words in the line are marked as incorrect. As explained in Subsection 3.2.1 this is because the segmentation and annotation of the IAM database at the word level were done with a procedure that manually validated the results at line level. There are approximately 15% of words marked as wrong in the database, although only a part of them are really segmentation errors.

In most publications, there is no reference to this fact from the IAM database, we understand that in these cases the correct transcription mark is not considered, and all the images available in the selected partition are used, although we cannot be entirely sure. If all the images are not used, a part of the errors accounted in these publications comes from the segmentation error and not necessarily from an error in the proposed model. In any case, any two publications using all available data are perfectly comparable. Several publications explicitly eliminate these cases, such as [200] or [222]. In the particular case of [222] it is indicated that the erroneous data have been excluded from the test partition but have been considered in the train and validation partitions, arguing that the additional cases included compensate for the actual transcription errors in terms of training improvement.

In terms of evaluation metrics, the elimination of cases marked as erroneous from all partitions has the advantage that transcription errors are eliminated from the test partition. However, it has the disadvantage that the training dataset is reduced, which makes the model difficult to train, especially in the case of deep architectures, for which having a larger volume of train data is essential. In any case, the results to be compared should be calculated on the same set of test images, whether they have these errors or not.

6.4.3 Character set selected for decoding

The IAM database has 79 characters that include the upper and lower case letters of the English language, the 10 digits, various punctuation marks, the character white space (ASCII 32), and a special character for garbage (#). Most publications use the complete character set or ignore the garbage character and the white space character. The frequency of these last two characters in the database is very low (33 cases of each character), and there are no significant differences between considering them or not in the results.

Most authors perform the evaluation using the entire set of characters indicated. However, certain authors limit the characters used in the transcription. For example:

- In reference [128], it is indicated that the evaluation of the performance is done in a case-insensitive manner, and includes comparisons of that evaluation with references [2], [26] and [64], which in our understanding use the full charset with a distinction between upper and lower case characters.
- In reference [169] the author indicates that the character set employed contains the lower and upper case of the Latin alphabet. Digits were not included but are considered in the accounting of errors. Punctuation marks are ignored. The publication includes a table comparing test score with multiple authors (including [2], [26] or [64]) who are understood to use the complete character set including punctuation marks.
- Another example appears in [68] which uses a charset that includes only lowercase letters and compares it to reference works [201], [170] or [164] which use the full character set with capitalization and punctuation.

• In the case of [222] the authors use different charsets, including or not capital letters and punctuation marks. It provides results for the different cases and compares them with other authors who have used the same character set. Thus, valid comparisons are obtained with multiple authors and for the different sets of characters used.

In the case of the RIMES database for the ICDAR2011 partition, it has a total of 81 characters. Most authors use the above character set, but some authors use a character set that ignores uppercase letters.

- In reference [149], the authors indicate that the evaluation is done over ICDAR2011 competition dataset, to which the publication refers, is case-insensitive.
- In publication [199], the evaluation is carried out using lowercase letters for comparison purposes and indicating that there are transcription errors, especially for the letter 'j', in words 'je' and 'j'. This last publication establishes comparisons with [149] which also provides results for the lower case character set, and with [169] which declares to use the full character set with upper and lower case.
- In reference [222] the authors use different char sets, and establish valid comparisons in different tables. However, the final results are obtained with a reduced character set and these are cited in comparative tables of other authors who use the full character set (for example [116]).

The use of a reduced character set in decoding has a significant impact on WER and CER metrics. In Wigington et al. [222] a 23% improvements are declared in WER (from 3.69 to 2.85) and a 20% in CER (from 1.69 to 1.34) over the RIMES database.

6.4.4 Data augmentation

There are two data augmentation strategies that can be applied separately or together: data augmentation during the training process and test-time data augmentation.

Due to the variability of configuration strategies and the introduction of data augmentation in the training process, the different strategies and algorithms can be considered as part of the solution proposed by each author. Therefore, from the point of view of comparing the final results of two different approaches, it is perfectly correct to compare two approaches that use different data augmentation strategies in train or one that uses it with another that does not use it. There are references in the literature (such as [151], or [57]) that indicate that data augmentation techniques improve the training of deep network architectures, especially in areas where the dataset size is not very large, reducing overfitting and improving the generalizability of the model. However, in the case of HMM applied to handwritten text recognition, it is not a commonly performed practice.

The test-time data augmentation tends to improve significantly the accuracy metrics by depending on the type of model used and on the size of the training dataset [188]. For example, Wigington et al. [222] report a 20% of WER reductions for IAM word level (official partition) and a 25% reductions for RIMES at word level based on a specific experiment to evaluate the test-time data augmentation strategy.

The test-time data augmentation is a procedure that can be applied to any model built, which will generally improve the evaluation metrics. In this sense, its use is commonly conditioned on whether such a procedure is a common practice in the problem literature. In handwritten text recognition, it is not a common practice to provide the evaluation metrics results using test-time data augmentation. In cases where it is used, as in [169], [68] or in [222], direct comparison of the results with other authors who do not use test-time data augmentation makes it difficult to interpret the data since it is not possible to determine whether the differences of results between methods are due to larger or smaller efficiency, or to the use of test-time data augmentation by one of them and not by the other. Ideally, if this technique is used, we suggest the inclusion of an ablation study that provides comparative results of the algorithm with and without this technique.

6.4.5 Data normalization

A common practice in the various approaches to offline handwritten text recognition problem is to perform an image normalization pre-processing to reduce the inherent variability of the handwritten text. In fact, in older publications, the text normalization step is a practice almost always present (see, for example, references [81] or [23]). In recent years, especially with deep neural network architectures, some authors such as Huang et al. [99] or Kang et al. [114], propose recognition algorithms in which no prior normalization of the input images is applied to the model, except for a rescaling and adjustment to a fixed size of all images. This is usually due to some specific requirements of models that need that the input data have a particular structure.

In this Thesis, experiments have been carried out both with original images without normalization and with normalized images following the image preprocessing algorithm described in Subsection 6.2.3.

As far as the comparison of different results is concerned, data normalization plays a similar role to data augmentation during training. Since specific models only work correctly if the data are normalized in a certain way, it is understood that data normalization is part of each of the problem-solving strategies proposed by each author. Therefore, comparing results from two approaches that apply different normalizations or comparing with solutions that do not include a normalization step provides interpretable conclusions about the compared results.

If the normalization step is optional in the proposed approach, it is recommended to include, as part of the results, an ablation study comparing the results with and without the use of data normalization.

6.4.6 Decoding with lexicons and language models

As detailed in Section 2.4, in the HTR problem is a common practice to apply a decoding process to the result, provided by the visual model that encodes the input image so that the context and language information of the text to be decoded are taken into consideration. For this purpose, language models (LM) or lexicon search are usually used, especially in cases where an architecture based on Hidden Markov Models (HMM) was proposed.

As discussed in the previous section on data augmentation, it is correct to compare different publications that address the problem with or without decoding. It is especially true when the decoding process is integrated as a part of the model that is trained together. It happens, for example, in models based on HMM that need subsequent decoding through an LM to obtain results.

In cases where decoding with lexicons or LM is an optional subsequent step applied to the output of the visual model, comparisons are more controversial. If it is done to compare with another that also does it, the most appropriate is to use the same lexicon or the same LM, but for that, there should be a standard way to use or share the language model. If there are licensing issues to share the object, instructions or the source code to create the object should be shared. If comparing with another approach that does not employ an LM, it is most appropriate to compare the results of the visual models without using the LM.

The objective is to measure and compare the different publication contributions separately to determine the relevance of each one in terms of error. For example, a publication proposes a new visual model, uses a new language model, and only provides a single error measure that improves the state-of-the-art. In that case, it is impossible to determine whether the improvement comes from the visual model or the LM or the combination of both.

In the case of this Thesis, we employ two lexicons by database to evaluate the models. First, a standard lexicon commonly used in the literature for this database, and second, the test lexicon, to get an error measure without out-of-vocabulary (OOV) words. In our case, whenever error results are provided using a lexicon, the same result is also provided without using any lexicon.

6.5 Comparative of results

This section compares the results obtained by the proposed model with results from other publications, given the difficulties and considerations made in the previous section. First of all, some recommendations aimed at comparing results easily interpretable are proposed in Subsection 6.5.1. Subsequently, the results are compared on RIMES database in Subsection 6.5.2 and on IAM database in Subsection 6.5.3, respectively.

6.5.1 Best practices recommendations to results comparative

The main objective of these best practices suggested is to be able to quantify as precisely as possible each of the contributions of a publication providing different error measures for each of the innovations introduced, and detailing all the elements necessary to guarantee the reproducibility of the results.

First of all, it is always recommended to publish the results for the standard metrics of CER and WER calculated on a well-identified test data set. In general, it is recommended not to alter the databases used, i.e., not to filter any data and to use the complete character set of the databases, differentiating between uppercase, lowercase, digits, and special characters.

When including tables of comparative results with other authors, make sure that they can be compared without bias because they are calculated not only on the same datasets but using the same set of valid characters and with the same image selection for training and testing (the same error cases are always filtered out). In general, these tables compare the best final results of each publication, including all the contributions made together so the recognition method can be evaluated entirely. This data is usually easily identifiable in all available publications. If the publication makes relevant contributions in different aspects of the problem, it is proposed to publish comparative tables with other authors that focus on each of these aspects separately. For example, it is recommended to construct an additional comparative table using only the visual model and comparing it with results from other authors that have also been obtained using only visual models, if contributions are made on both the visual model and the language model.

The presentation of partial results to measure the relative impact on the error of the different inputs is also proposed. It is suggested first to provide the results obtained directly by the visual model exclusively. Defining a baseline model and then providing results on the absolute and relative improvement of the inclusion of each of the published inputs to the baseline model.

It is also recommended to publish sufficient details of the database used, the proposed model, the lexicons or language models used, and the experimental setup implemented to facilitate the reproducibility of the results obtained. Ideally, it is proposed to publish the source code used in the experiments.

The following subsections include detailed tables of results from various publications for IAM and RIMES databases at both word and line levels. In the case of IAM database, the results obtained with the two available partitions have been separated. Other results not included in the tables are also detailed, explaining the reasons and the differences between these cases and those appearing in the tables.

6.5.2 Historical results comparative for RIMES database

The following are the published results for the HTR continuous problem using RIMES database at word level. Values are provided for the CER and WER metrics, calculated on the test partition defined for the ICDAR2011 competition when available. This test partition is composed of a total of 100 lines and 7,464 words.

Our results included in the tables correspond to those provided in Subsection 6.3.9, corresponding to the final model selected. The most appropriate parameterization has been selected according to the results in the validation partitions obtained in the previous experiments.

The results obtained using the complete set of characters available in the database for decoding are included in the tables. Publications using a reduced character set and publications using test time data augmentation are included separately.

Differentiated tables are provided for three cases depending on the use of lexicons or language models to obtain the results:

- Without using lexicons or LMs.
- Using lexicons or LMs that do not include the test partition words (so, there are OOV words usually).

• Using lexicons or LMs that do include the test partition words (so, there are no OOV words).

The following abbreviations are used in the notes to the different tables.

- DA: train data augmentation is used
- Synt: A pretrain with synthetic data is performed.
- Norm: Normalization is used or not (No norm.).
- LM: Language model is used to decoding.

Table 6.20 includes the WER and CER values obtained on the test partition of RIMES database corresponding to the ICDAR2011 competition. The results have been obtained by directly applying a visual model and are not decoded by any lexicon or language model.

Reference	Year	Model type	Notes	CER	WER
Bluche et al. [19]	2015	MLP		17.8	59.5
Bluche et al. [19]	2015	RNN		5.6	20.9
Bluche et al. $[20]$	2015			2.9	12.6
Kang et al. $[116]$	2021	seq2seq	No norm, DA, Synt	2.65	8.71
Ours	2021	seq2seq	Norm, DA	4.5	13.2

Table 6.20: RIMES word level results. Visual model. No lexicon or LM.

Table 6.21 includes the WER and CER values obtained on the test partition of the RIMES database corresponding to the ICDAR2011 competition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that does not use the test word set.

Reference	Year	Model type	Notes	CER	WER
Bluche et al. $[19]$	2015	MLP		7.2	26.1
Bluche et al. $[19]$	2015	RNN		4.3	16.4
Kang et al. $[116]$	2021	seq2seq	No norm, DA, Synt, LM	2.32	7.47
Ours	2021	seq2seq	Norm, DA	5.9	12.2

Table 6.21: RIMES word level results. Lexicon or LM not including test data.

Table 6.22 includes the WER and CER values obtained on the test partition of the RIMES database corresponding to the ICDAR2011 competition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that explicitly uses the set of test words.

Error metrics from other publications are listed below, and the reason for not including them in the above tables of RIMES results is given:

• Menasri et al. [149], in 2012, use a character set case insensitive. It obtains a WER of 4.75 on the ICDAR2011 test partition using a lexicon that includes the test words.

CHAPTER 6. DEEP ARCHITECTURES APPLIED TO CONTINUOUS OFFLINE HANDWRITING RECOGNITION

Reference	Year	Model type	Notes	CER	WER
Bluche et al. [23] Kang et al. [116] Ours	2013 2021 2021	CNN + HMM seq2seq seq2seq	No norm, DA, Synt, LM Norm, DA	$\begin{array}{c} \textbf{1.45} \\ 2.7 \end{array}$	9.2 3.75 5.4

Table 6.22: RIMES word level results. Test data included in lexicon or LM.

- Pham et al. [164], in 2014, use the RIMES database partition corresponding to the ICDAR2009 competition (instead of the ICDAR2011 partition). Since the test data of these two partitions do not match, comparisons between the two partitions cannot be established. The model implemented is MDLSTM type and obtains a WER of 27.01 and a CER of 8.62 without using lexicons or LM.
- Poznanshi et al. [169], in 2016, indicate that the character set employed for RIMES contains the lower and upper case Latin alphabet, digits and accented letters. Punctuation marks are ignored. The model used are CNN-based and achieved a CER of 1.90 and a WER of 3.90 for RIMES using the test lexicon for decoding.
- Wigington et al. [222], in 2017, use an MDLSTM with CTC, employing test time data augmentation and a reduced character set without punctuation marks and upper case. It obtained a WER of 11.29 and a CER of 3.09 without any lexicon or LM, and a WER of 2.85, and a CER of 1.36 using the test lexicon.
- Dutta et al. [68], in 2018, use a charset that includes only lowercase letters. They implement a pre-trained seq2seq type model with synthetic images and obtained a WER of 7.04 and a CER of 2.32 for the RIMES test partition without employing any lexicon or LM. When using a lexicon that includes the test words they obtained a WER of 1.86 and a CER of 0.65.

6.5.3 Historical results comparative for IAM database

The results obtained in the word recognition problems on the IAM database are detailed in the tables below. The results that are comparable in the terms discussed in preceding subsections are grouped.

These results are shown differentiated by partition (Official and Aachen). If any publication uses a test set different from those of these partitions, it is listed separately. Results obtained in papers using the entire set of characters available in the database for decoding are included in the table. Papers using a reduced character set and publications using test time data augmentation are also listed separately.

For each of the two partitions, separate tables are provided for three cases depending on the use of lexicons or language models:

- The result has been obtained without using lexicons or LMs.
- Lexicons or LMs that **do not** include the test sample have been used.
- Lexicons or LM that **do** include the test sample have been used.

The publication year, the model type, and the necessary notes to identify the particularities of a specific result are detailed in the tables for each reference. The CER and WER metrics are included when available as results. In the case that data are not known, these are left blank.

Table 6.23 includes the WER and CER values obtained on the test data of IAM database corresponding to the *Aachen* partition. The results have been obtained by directly applying a visual model and are not decoded by any lexicon or language model.

Citation	Year	Model type	Notes	CER	WER
Pham et al. [164]	2014	MDLSTM	No LM	13.92	31.44
Bluche et al. $[19]$	2015	MLP		15.6	54.2
Bluche et al. $[19]$	2015	RNN		7.3	24.7
Bluche et al. $[20]$	2015			7.9	24.6
Kang et al. $[116]$	2021	seq2seq	No norm, DA, Synt	5.79	15.15
Ours	2021	seq2seq	Norm, DA	9.3	21.0

Table 6.23: IAM-Aachen test partition results. Word level results. Visual model, no lexicon or LM.

Table 6.24 includes the WER and CER values obtained on the test data of IAM database corresponding to the *Aachen* partition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that does not use the set of test words.

Citation	Year	Model type	Notes	CER	WER
Kang et al. [116]	2021	seq2seq	No norm, DA, Synt, LM	5.74	15.11
Ours	2021	seq2seq	Norm, DA	11.2	19.6

Table 6.24: IAM-Aachen test partition results. Word level results. Lexicon or LM not include test data.

Table 6.25 includes the WER and CER values obtained on the test data of IAM database corresponding to the *Aachen* partition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that explicitly uses the set of test words. In this case, we have not been able to identify another publication that meets the conditions identified for inclusion in the table. Several cases could be included, but they use a reduced character set or use test time data augmentation.

Citation	Year	Model type	Notes	CER	WER
Kang et al. [116]	2021	seq2seq	No norm, DA, Synt, LM	4.27	8.36
Ours	2021	seq2seq	Norm, DA	1.0	13.0

Table 6.25: IAM-Aachen test partition results. Word level results. Test data included in lexicon or LM.

Listed below are the results obtained in the publications using the official partition of the IAM database. As indicated in Subsection 6.4.1 this partition has undergone variations over time. For example, publications prior to 2010 declare a different number of lines

in the official test partition. When including results from different publications in the tables, we have followed the criterion that the number of lines or test words match the data published in the official IAM database ² which corresponds to a total of 1,861 lines, or 17,616 words. In some cases, as in [170], the counts of the train or validation partitions do not match exactly with those available on the database official website, although the test count does. These cases have also been included in the tables because, as previously argued, the comparison is considered valid if it has been calculated on the same test data, regardless of whether there are differences in the other partitions.

Table 6.26 includes the WER and CER values obtained on the test data of IAM database corresponding to the *Official* partition. The results have been obtained by directly applying a visual model and are not decoded by any lexicon or language model.

Citation	Year	Model type	Notes	CER	WER
Bluche et al. [23]	2013	CNN + HMM			23.7
Ours	2021	seq2seq	Norm, DA	9.4	19.8

Table 6.26: IAM-Official test partition results. Word level results. Visual model, no lexicon or LM.

Table 6.27 includes the WER and CER values obtained on the test data from the IAM database corresponding to the *Official* partition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that does not use the set of test words.

Citation	Year	Model type	Notes	CER	WER
Bluche et al. $[23]$	2013	CNN + HMM	LM		20.5
Ours	2021	seq2seq	Norm, DA	11.3	18.7

Table 6.27: IAM-Official test partition results. Word level results. Lexicon or LM not include test data.

Table 6.28 includes the WER and CER values obtained on the test data of IAM database corresponding to the *Official* partition. The results have been obtained by decoding the output of the visual model by a lexicon or language model that explicitly uses the set of test words.

Citation	Year	Model type	Notes	CER	WER
Bluche et al. [23]	2014	CNN + HMM	Norm, DA	9.2	20.5
Almazan et al. [2]	2014	k-NN		11.27	20.01
Ours	2021	seq2seq		7.4	12.0

Table 6.28: IAM-Official test partition results. Word level results. Test data included in lexicon or LM.

Error metrics from other publications are listed below, and the reason for not including them in the above tables of IAM results is given:

²http://www.fki.inf.unibe.ch/databases/iam-handwriting-database (last access: April 2021)

- Graves et al. [81], in 2009, use the previous official partition of IAM databases, with 2,781 lines in the test set. The authors use a LSTM with the CTC decoder and provides a WER of 25.9 and a CER of 18.2 using a standard lexicon that not include the test words.
- Krishnan et al. [128], in 2016, indicate that the evaluation of the performance is done in a case-insensitive manner. This work provides a WER of 6.61 and a CER of 3.72 decoding with the test lexicon over the official IAM partition.
- Poznanski and Wolf [169], in 2016, indicate that the character set employed contains the lower and upper case Latin alphabet. They also point out that digits were not included but are considered in the accounting of errors. Punctuation marks are ignored. The model used are CNN based and get a CER of 3.44 and a WER of 6.45 using the official partition and the test lexicon for decoding.
- Wigington et al. [222], in 2017, use a MDLSTM with CTC, employing test time data augmentation and a reduced character set without punctuation and case. For IAM dataset, they obtain a WER of 19.07 and a CER of 6.07 without Lexicon or LM, and a WER of 5.71 and a CER of 3.03 using the test lexicon. It is not possible to determine which partition has been used.
- Dutta et al. [68], in 2018, use a charset that includes only lowercase letters. They implement a pre-trained seq2seq type model with synthetic images and obtains a WER of 12.61 and a CER of 4.88 for the official IAM test partition without employing any lexicon or LM. Using the test lexicon, they obtain a WER of 4.07 and a CER of 2.17.
- Kang et al. [113], in 2018, propose a seq2seq model and they achieve a CER of 6.88 and a WER of 17.45 without using any lexicon or LM using the Aachen test partition, but exclude cases not marked as 'OK', using 20,305 words out of 25,920 in the original approval.

6.6 Discussion

In this chapter, we have presented a system for recognizing offline handwritten words. The proposed architecture is inspired in the way that the humans try to identify handwritten words and it is based in the seq2seq architecture with the addition of a convolutional network. A detailed set of experiments has been performed to analyze the possibilities of this new architecture. The main results and conclusions obtained from these experiments are the following ones:

• The analysis performed at the character recognition level concluded that VGG convolutional architecture was the most suitable one for this problem. However, in the experiments carried out at the word level, it is observed that LeNet and VGG architectures analyzed obtain different results depending on the database and whether or not the image partitioning strategy is applied in patches. From the above, it is not easy to establish a conclusion that favors one architecture over the other, especially in the standard IAM and RIMES databases. What is observed is that LeNet architecture performs better than VGG in the smaller Osborne database.
CHAPTER 6. DEEP ARCHITECTURES APPLIED TO CONTINUOUS OFFLINE HANDWRITING RECOGNITION

- From the results obtained, it is concluded that dividing the original image into patches does not have a significant advantage over the strategy of directly applying the convolutional component to the original image without dividing it. In any case, it is considered that this patching strategy provides a more robust way to connect the output of the convolutional component with the RNN of the encoder. It is true that the patching strategy requires an adjustment of the patch size and step size parameters to work optimally, but it has been observed in the experiments that results very close to the optimum can be obtained for wide ranges of patch sizes and step sizes. It is therefore considered to be a robust architecture concerning these two parameters.
- In general, the optimal parameterization identified is quite similar for the IAM and RIMES databases. In the case of Osborne database, there are significant differences, especially in the encoder parameterization. It is considered that the smaller size of this database contributes to the fact that the results of the experiments performed with it have less stability than those performed with the other two databases. Therefore, conclusions concerning Osborne database should be taken with caution.
- Both the pre-normalization of the text images and the train data augmentation strategy significantly improve the model accuracy. In the experiments carried out, it has been observed that the combination of both techniques is the most influential factor in reducing the error consistently in the three databases analyzed.
- The error analysis indicates that the model performs worse with long texts. It could be caused by the fact that the decoder identifies the characters present in the image sequentially (i.e., for identifying the next character, the prediction of the previous characters is used as input). However, the attention mechanism allows access to each decoder association to the complete information encoded in the input image. It is also likely that the low frequency of long text images in the databases is a factor that explains the low accuracy of the recognition of these long texts.
- The main advantage of the proposed model is the integrated management of local aspects of a word image related to the characters that compose the word (convolutional network), the sequential horizontal aspects of the word image (encoder), and the sequence of characters in the word (decoder). When applying our visual model on two standard databases, the obtained results are competitive with those others published in the literature.

Chapter 7

Conclusions

In this Thesis, we have focused on investigating the problem of offline continuous handwritten text recognition, consisting of transforming cursive text images into their transcription. In particular, we have worked with word images for which no character segmentation is available.

The motivation and specific objectives of this thesis are detailed in the Chapters 1 and 2. In addition, an introduction to the HTR problem in general and a more detailed definition of the offline continuous HTR problem, the specific subject of this thesis, are provided. It also introduces the general concepts of the problem, such as image processing techniques, decoding alternatives, or performance evaluation metrics.

Chapter 3 provides detailed descriptions of the databases available to address the problems of recognizing isolated handwriting characters and continuous HTR. By studying the databases available for isolated characters and words, we identified that none of the character databases include examples of special characters or punctuation marks present in the continuous text databases. To fill this gap, we presented the new database COUT of handwritten isolated character images that includes a wide range of up to 93 different characters, including all printable characters of the ASCII standard. To our best knowledge, it is the database of isolated handwritten characters with more different categories. Additionally, an algorithm is provided for altering the original characters by incorporating artifacts that emulate the context in which the character is found when it is part of a word. This facilitates the study of isolated character recognition models that can be incorporated into continuous word recognition models.

Chapter 4 provides an extensive review of the HTR problem research over the years. We detail the various contributions that have made significant advances in resolving the problem by putting them in context, and we review in particular detail the aspects most related to the work completed in this Thesis. This review allows us to have a clear perspective of the main research lines in the continuous HTR problem.

Next, in Chapter 5 we explore several models with different convolutional architectures applied to the problem of recognition of isolated handwritten characters. We provide a detailed analysis of the capabilities and limitations of each model when are used on different character sets, including upper and lower case letters, special characters, and punctuation marks. We have consistently identified that the VGG architecture as produced the lowest error in recognition of isolated handwriting characters. In Chapter 6 we propose a novel model architecture for the continuous offline HTR at the word level. It is based on the combination of a Convolutional Neural Network (CNN) and the sequence-to-sequence (seq-to-seq) algorithm. We provide an extensive set of experiments analyzing the impact of each of the model components on the model error. This proposed model has demostrated that it obtains competitive results comparable to the state-of-the-art with the main offline continuous HTR databases.

We found that the strategies of applying the proposed model directly on the image of each word or applying it on a sequence of patches extracted from each image provide results that favor the chunk partitioning strategy as the length of the image text increases.

We found that normalizing the handwritten text images improves the accuracy of the proposed recognition model. We proposed a novel normalization algorithm for handwriting images that significantly reduces the variability of handwritten text. As part of this algorithm, we presented a new slant angle identification algorithm and a new ascenders and descenders normalization height method.

We found that applying an integrated data augmentation strategy in the trial process significantly improves the results of the proposed model. We also propose a new data augmentation pipeline that includes the most relevant handwritten text image transformations from the last few years. We found that data normalization and data augmentation strategies are not mutually exclusive and that the combination of both provides better results than the use of each one separately.

We analyzed the contributions of other authors and identified the particularities of each published paper that make it difficult to compare the results between different models. We proposed a comparative framework that avoids the previous problems and that facilitates the fair comparison of results in the clearest possible way. In this line, we presented a wide set of results that facilitate comparing the proposed model with those of other authors, eliminating the identified biases. Numerous tables with partial results that allow us to evaluate the individual contribution of each of the proposed improvements in the results of the new model have also been provided.

Finally, we have publicly distributed ¹ all the source code used in the experiments performed in this Thesis under an open-source license. We have also provided the instructions and scripts necessary to replicate all the experiments performed in this Thesis.

7.1 Future work

We identified the following lines of future work on the construction of **isolated character recognition** models:

- Experiment with other more recent network architectures that give good results on other Computer Vision problems, such as the EfficientNet [208] models.
- Experiment with other data augmentation techniques to improve the generality of the models built. For example, morphological erode and dilate, or elastic distortion techniques such as those suggested in [50].

¹https://github.com/sueiras

- Carry out cross-validations between the databases to better understand their generalization capacity. For this purpose, it should be taken into account that the different databases in origin have essential differences in terms of image size, border size, and whether they are binarized or not.
- Analyze the generalization capacity of the models built in real cases. One of the applications of recognizing isolated handwritten characters is the automatic reading of forms with writing templates that separate each character. In these cases, once the forms have been digitized and the characters have been isolated, their images can present artifacts coming from the template. Additionally, they may be cropped by the segmentation process. Analyzing the performance of models built on these templates can be of industrial relevance.
- It has been observed that the significant difference in distribution between the train and test partitions of the NIST database can distort the mean error. It is considered that balancing the training sample could correct these distortions. This balancing could be done by stratified sampling or by data augmentation applied on the less frequent characters.

The lines of future work identified on the construction of **continuous HTR models** at word level are the following:

- Data analysis in this Thesis has shown that normalization and data augmentation processes are relevant in improving the accuracy. However, these are transformations on the text image that act in the opposite way. Thus data augmentation alters the previously normalized images at each training step. It is suspected that a data augmentation strategy that keeps as much as possible the characteristics of the normalized text may be more effective when used in conjunction with normalization. So further analysis of this interaction may be useful.
- In analyzing the different strategies of how to include the convolutional component in the model, it was hypothesized that dividing the image into patches might be more beneficial for large words. This hypothesis could be verified in future work.
- The use of pre-trained convolutional components on which training performs finetuning is a strategy that has worked well in other cases [207] [113]. Therefore, it will be explored as an alternative to the full training that has been used in this Thesis.
- The use of an RNN type network in the decoder opens the door to integrating RNNbased language models as part of the decoder. It would have advantages over the usual option of applying the language model to the output of the visual model, as the visual and language models would be trained simultaneously.
- The model needs also to be tested at line level. It will also allow us to analyze the behavior of the patching strategy using online images. The modeling of the databases at line level also has the advantage that it allows us to compare the results with other authors who have only published results at that level, especially in the last few years [223], [104], [115].

- The limitation observed in the performance of the current models due to the small size of the available annotated databases will be analyzed. Therefore, synthetic data generation strategies that have worked well in other Computer Vision problems could be explored. Based on the initial tests that have been performed, the most promising approach is the use of digital typographic fonts that emulate handwritten text as a mean of generating infinite samples of annotated handwriting data. This line of work is one of the most promising ones.
- It is planned to test if this model can also work with other languages that do not use Latin characters, like for example, Chinese or Arabian ones. The proposed model can be applied directly to languages that read from right to left, or with minimal changes on vertically typed languages.

7.2 Publications

The research carried out in this thesis has produced the following publications:

- Sueiras, J., Ruiz, V., Sanchez, A., and Velez, J. F. (2018). Offline continuous handwriting recognition using sequence to sequence neural networks. Neurocomputing, 289, 119-128.
- Sueiras, J., Ruiz, V., Sánchez, Á., and Vélez, J. F. (2016, December). Using a Synthetic Character Database for Training Deep Learning Models Applied to Offline Handwritten Recognition. In International Conference on Intelligent Systems Design and Applications (pp. 299-308). Springer, Cham.
- Ruiz, V., de Lena, M. T. G., Sueiras, J., Sanchez, A., and Velez, J. F. (2017, June). A Deep Learning Approach to Handwritten Number Recognition. In International Work-Conference on the Interplay Between Natural and Artificial Computation (pp. 193-202). Springer, Cham.

Bibliography

- M. Abadi et al. TensorFlow: Large-Scale Machine Learning on Heterogeneous Distributed Systems. 2016. arXiv: 1603.04467 (cited on page 115).
- J. Almazan et al. "Word spotting and recognition with embedded attributes". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 36.12 (2014), pages 2552–2566. ISSN: 01628828. DOI: 10.1109/TPAMI.2014.2339814 (cited on pages 68, 147, 155).
- G. J. Babu and A. Bose. "Bootstrap confidence intervals". In: Statistics and Probability Letters 7.2 (1988), pages 151–160. ISSN: 01677152. DOI: 10.1016/0167-7152(88)90042-9 (cited on page 22).
- [4] D. Bahdanau, K. H. Cho, and Y. Bengio. "Neural machine translation by jointly learning to align and translate". In: 3rd International Conference on Learning Representations, ICLR 2015 - Conference Track Proceedings (2015). arXiv: 1409. 0473 (cited on pages 6, 63, 68, 72).
- [5] D. Bahdanau et al. "End-to-end attention-based large vocabulary speech recognition". In: ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing Proceedings. Volume 2016-May. 2016, pages 4945–4949. ISBN: 9781479999880. DOI: 10.1109/ICASSP.2016.7472618. arXiv: 1508.04395 (cited on pages 38, 68, 125).
- [6] L. B. Baum and J. A. Eagon. "An inequality with applications to statistical estimation for probabilistic functions of markov processes and to a model for ecology". In: *Bulletin of the American Mathematical Society* 73.3 (1967), pages 360–363. ISSN: 02730979. DOI: 10.1090/S0002-9904-1967-11751-8 (cited on page 45).
- [7] L. E. Baum et al. "A Maximization Technique Occurring in the Statistical Analysis of Probabilistic Functions of Markov Chains". In: *The Annals of Mathematical Statistics* 41.1 (1970), pages 164–171. ISSN: 0003-4851. DOI: 10.1214/aoms/1177697196 (cited on page 45).
- [8] S. Bengio et al. "Scheduled sampling for sequence prediction with recurrent neural networks". In: Advances in Neural Information Processing Systems. Volume 2015-January. 2015, pages 1171–1179. arXiv: 1506.03099. URL: https://proceedings. neurips.cc/paper/2015/file/e995f98d56967d946471af29d7bf99f1-Paper. pdf (cited on page 71).
- Y. Bengio et al. "LeRec: a NN/HMM hybrid for on-line handwriting recognition." In: Neural computation 7.6 (1995), pages 1289–1303. ISSN: 08997667. DOI: 10. 1162/neco.1995.7.6.1289 (cited on page 59).

- [10] Y. Bengio, Y. LeCun, and D. Henderson. "Globally Trained Handwritten Word Recognizer using Spatial Representation Convolutional Neural Networks and Hidden Markov Models". In: Advances in Neural Information Processing Systems 6 (1994), pages 937–944. URL: https://proceedings.neurips.cc/paper/1993/ file/3b5dca501ee1e6d8cd7b905f4e1bf723-Paper.pdf (cited on page 52).
- [11] Y. Bengio, P. Simard, and P. Frasconi. "Learning Long-Term Dependencies with Gradient Descent is Difficult". In: *IEEE Transactions on Neural Networks* 5.2 (1994), pages 157–166. ISSN: 19410093. DOI: 10.1109/72.279181 (cited on pages 49, 53).
- S. K. Bera et al. "A One-Pass Approach for Slope and Slant Estimation of Tri-Script Handwritten Words". In: *Journal of Intelligent Systems* 29.1 (2020), pages 688– 702. ISSN: 03341860. DOI: 10.1515/jisys-2018-0105 (cited on page 42).
- [13] S. Bercu and G. Lorette. "On-Line Handwritten Word Recognition: An Approach Based on {Hidden Markov Models}". In: Proc. Third Int'l Workshop Frontiers in Handwriting Recognition. 1993, pages 385–390 (cited on page 46).
- [14] R. Bertolami and H. Bunke. "Hidden Markov model-based ensemble methods for offline handwritten text line recognition". In: *Pattern Recognition* 41.11 (2008), pages 3452–3460. ISSN: 00313203. DOI: 10.1016/j.patcog.2008.04.003 (cited on pages 18, 40, 45–47, 77, 78).
- [15] Bharadwaj, K. B. Prakash, and G. R. Kanagachidambaresan. Pattern Recognition and Machine Learning. springer, 2006, pages 105–144. DOI: 10.1007/978-3-030-57077-4_11 (cited on page 5).
- [16] A. Białecki, R. Muri, and G. Ingersoll. "Apache Lucene 4". In: Proceedings of the SIGIR 2012 Workshop on Open Source Information Retrieval. 2012, pages 17–24. ISBN: 978-0-473-22026-6. URL: http://www.cs.otago.ac.nz/homepages/andrew/involvement/2012-SIGIR-OSIR.pdf?origin=publication%78%5C_%7Ddetail%78%5C#%7Dpage=22 (cited on page 17).
- [17] G. Bideault et al. "Spotting handwritten words and REGEX using a two stage BLSTM-HMM architecture". In: *Document Recognition and Retrieval XXII*. Volume 9402. International Society for Optics and Photonics. 2015, 94020G. ISBN: 9781628414929. DOI: 10.1117/12.2075796 (cited on page 38).
- G. M. Binmakhashen and S. A. Mahmoud. "Document layout analysis: A comprehensive survey". In: ACM Computing Surveys 52.6 (2019). ISSN: 15577341. DOI: 10.1145/3355610 (cited on page 10).
- [19] T. Bluche. "Neural Networks for Large Vocabulary Handwritten Text Recognition". PhD thesis. Université Paris Sud-Paris XI, 2015, page 246. URL: https: //tel.archives-ouvertes.fr/tel-01249405 (cited on pages 22, 41, 45, 48, 152, 154).
- [20] T. Bluche. "Joint line segmentation and transcription for end-to-end handwritten paragraph recognition". In: Advances in Neural Information Processing Systems. 2016, pages 838–846. arXiv: 1604.08352 (cited on pages 63, 67, 152, 154).

- [21] T. Bluche, J. Louradour, and R. Messina. "Scan, Attend and Read: End-To-End Handwritten Paragraph Recognition with MDLSTM Attention". In: *Proceedings* of the International Conference on Document Analysis and Recognition, ICDAR 1 (2017), pages 1050–1055. ISSN: 15205363. DOI: 10.1109/ICDAR.2017.174. arXiv: 1604.03286 (cited on pages 29, 38, 39, 57, 60, 62–64, 67, 74).
- [22] T. Bluche and R. Messina. "Gated Convolutional Recurrent Neural Networks for Multilingual Handwriting Recognition". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 1. IEEE. 2017, pages 646–651. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.111 (cited on pages 57, 59, 63, 64, 67).
- [23] T. Bluche, H. Ney, and C. Kermorvant. "Tandem HMM with convolutional neural network for handwritten word recognition". In: *ICASSP*, *IEEE International Conference on Acoustics, Speech and Signal Processing Proceedings.* IEEE. 2013, pages 2390–2394. ISBN: 9781479903566. DOI: 10.1109/ICASSP.2013.6638083 (cited on pages 13, 30, 39, 40, 47, 52, 59, 149, 153, 155).
- [24] T. Bluche et al. "The A2iA arabic handwritten text recognition system at the Open HaRT2013 evaluation". In: *Proceedings - 11th IAPR International Workshop* on Document Analysis Systems, DAS 2014. IEEE. 2014, pages 161–165. ISBN: 9781479932436. DOI: 10.1109/DAS.2014.40 (cited on page 62).
- [25] T. Bluche et al. "Framewise and CTC training of Neural Networks for handwriting recognition". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 2015-November. IEEE. 2015, pages 81–85. ISBN: 9781479918058. DOI: 10.1109/ICDAR.2015.7333730 (cited on pages 66, 67).
- [26] T. Bluche, H. Ney, and C. Kermorvant. "A comparison of sequence-trained deep neural networks and recurrent neural networks optical modeling for handwriting recognition". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 8791. Springer. 2014, pages 199–219. DOI: 10.1007/978-3-319-11397-5_15 (cited on pages 30, 40, 47, 67, 78, 146, 147).
- [27] B. E. Boser, I. M. Guyon, and V. N. Vapnik. "Training algorithm for optimal margin classifiers". In: *Proceedings of the Fifth Annual ACM Workshop on Computational Learning Theory*. 1992, pages 144–152. ISBN: 089791497X. DOI: 10. 1145/130385.130401 (cited on page 49).
- [28] O. Boudraa, W. K. Hidouci, and D. Michelucci. "An improved skew angle detection and correction technique for historical scanned documents using morphological skeleton and progressive probabilistic hough transform". In: 2017 5th International Conference on Electrical Engineering - Boumerdes, ICEE-B 2017. Volume 2017-January. IEEE. 2017, pages 1–6. ISBN: 9781538606865. DOI: 10.1109/ICEE-B. 2017.8192043 (cited on page 41).
- [29] A. Boukharouba. "A new algorithm for skew correction and baseline detection based on the randomized Hough Transform". In: Journal of King Saud University Computer and Information Sciences 29.1 (2017), pages 29–38. ISSN: 22131248. DOI: 10.1016/j.jksuci.2016.02.002 (cited on page 41).

- [30] N. G. Bourbakis, C. Koutsougeras, and A. Jameel. "Handwriting recognition using a reduced character method and neural nets". In: *Nonlinear Image Processing VI*. Volume 2424. International Society for Optics and Photonics. 1995, pages 592–601.
 DOI: 10.1117/12.205260 (cited on page 57).
- [31] H. Bourlard and N. Morgan. Connectionist Speech Recognition A Hybrid Approach. Volume 247. Springer Science \ Business Media, 1994 (cited on pages 47, 51).
- [32] A. P. Bradley. "The use of the area under the ROC curve in the evaluation of machine learning algorithms". In: *Pattern Recognition* 30.7 (1997), pages 1145– 1159. ISSN: 00313203. DOI: 10.1016/S0031-3203(96)00142-2 (cited on page 20).
- [33] A. Brakensiek and G. Rigoll. "Handwritten address recognition using hidden markov models". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 2956. Springer, 2004, pages 103–122. DOI: 10.1007/978-3-540-24642-8_7 (cited on page 38).
- [34] T. M. Breuel. "High Performance Text Recognition Using a Hybrid Convolutional-LSTM Implementation". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 1. IEEE. 2017, pages 11–16. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.12 (cited on page 64).
- [35] J. S. Bridle. "Probabilistic Interpretation of Feedforward Classification Network Outputs, with Relationships to Statistical Pattern Recognition". In: *Neurocomputing*. Springer, 1990, pages 227–236. DOI: 10.1007/978-3-642-76153-9_28 (cited on pages 70, 112).
- [36] T. Brox. "Maximum Likelihood Estimation". In: Computer Vision. Edited by K. Ikeuchi. Boston, MA: Springer US, 2014, pages 481–482. ISBN: 978-0-387-31439-6.
 DOI: 10.1007/978-0-387-31439-6_674 (cited on page 45).
- [37] F. Camastra and A. Vinciarelli. "Markovian models for sequential data". In: Advanced Information and Knowledge Processing 2.9781447167341 (2015), pages 295–340. ISSN: 21978441. DOI: 10.1007/978-1-4471-6735-8_10 (cited on page 57).
- [38] M. Chen et al. "Generative Pretraining From Pixels". In: Proceedings of the 37th International Conference on Machine Learning. Volume 119. PMLR. 2020, pages 1691– 1703 (cited on page 74).
- [39] S. F. Chen and J. Goodman. "Empirical study of smoothing techniques for language modeling". In: *Computer Speech and Language* 13.4 (1999), pages 359–394.
 ISSN: 08852308. DOI: 10.1006/csla.1999.0128. arXiv: 9606011 [cmp-lg] (cited on page 18).
- [40] Z. Chen et al. "Simultaneous Script Identification and Handwriting Recognition via Multi-Task Learning of Recurrent Neural Networks". In: *Proceedings of the International Conference on Document Analysis and Recognition*, *ICDAR*. Volume 1. IEEE. 2017, pages 525–530. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.92 (cited on pages 62, 63, 67).
- [41] M. Cheriet et al. "Handwriting recognition research: Twenty years of achievement... and beyond". In: *Pattern Recognition* 42.12 (2009), pages 3131–3135. ISSN: 00313203. DOI: 10.1016/j.patcog.2009.03.014 (cited on page 37).

- [42] C. C. Chiu and C. Raffel. "Monotonic chunkwise attention". In: arXiv. 2017. arXiv: 1712.05382. URL: https://openreview.net/forum?id=Hko85plCW (cited on page 75).
- [43] K. Cho et al. "Learning phrase representations using RNN encoder-decoder for statistical machine translation". In: EMNLP 2014 2014 Conference on Empirical Methods in Natural Language Processing, Proceedings of the Conference. Doha, Qatar: Association for Computational Linguistics, 2014, pages 1724–1734. ISBN: 9781937284961. DOI: 10.3115/v1/d14-1179. arXiv: 1406.1078. URL: https://www.aclweb.org/anthology/D14-1179 (cited on pages 56, 72, 75).
- [44] J. Chorowski and N. Jaitly. "Towards better decoding and language model integration in sequence to sequence models". In: Proceedings of the Annual Conference of the International Speech Communication Association, INTERSPEECH. Volume 2017-August. 2017, pages 523–527. DOI: 10.21437/Interspeech.2017-343. arXiv: 1612.02695 (cited on pages 72, 125).
- [45] J. Chorowski et al. "Attention-based models for speech recognition". In: Advances in Neural Information Processing Systems. Volume 2015-January. 2015, pages 577– 585. arXiv: 1506.07503 (cited on pages 2, 6, 7, 48, 68, 73, 105).
- [46] A. Chowdhury and L. Vig. "An efficient end-to-end neural model for handwritten text recognition". In: arXiv. 2018. arXiv: 1807.07965 (cited on pages 30, 67, 75, 117).
- [47] J. S. Chung and A. Zisserman. "Lip reading in the wild". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 10112 LNCS. Springer. 2017, pages 87–103. ISBN: 9783319541839. DOI: 10.1007/978-3-319-54184-6_6 (cited on pages 68, 70, 105).
- [48] J. Chung et al. "Empirical Evaluation of Gated Recurrent Neural Networks on Sequence Modeling". In: NIPS 2014 Workshop on Deep Learning, December 2014. 2014. arXiv: 1412.3555 (cited on page 56).
- [49] D. Ciregan, U. Meier, and J. Schmidhuber. "Multi-column deep neural networks for image classification". In: *Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition*. 2012, pages 3642–3649. ISBN: 9781467312264. DOI: 10.1109/CVPR.2012.6248110. arXiv: 1202.2745 (cited on pages 80, 91–94).
- [50] D. C. Cireşan et al. "Convolutional neural network committees for handwritten character classification". In: *Proceedings of the International Conference on Document Analysis and Recognition*, *ICDAR*. IEEE. 2011, pages 1135–1139. ISBN: 9780769545202. DOI: 10.1109/ICDAR.2011.229 (cited on pages 90, 93, 94, 96, 160).
- [51] L. Cohen. "Generalized phase-space distribution functions". In: *Journal of Mathematical Physics* 7.5 (1966), pages 781–786 (cited on page 42).
- [52] D. Coquenet, C. Chatelain, and T. Paquet. "Recurrence-free unconstrained handwritten text recognition using gated fully convolutional network". In: Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR. Volume 2020-September. IEEE, 2020, pages 19–24. ISBN: 9781728199665. DOI: 10. 1109/ICFHR2020.2020.00015 (cited on page 57).

- [53] D. Coquenet, C. Chatelain, and T. Paquet. "End-to-end Handwritten Paragraph Text Recognition Using a Vertical Attention Network". In: (2020). arXiv: 2012.
 03868 (cited on page 67).
- [54] D. Coquenet et al. "Have convolutions already made recurrence obsolete for unconstrained handwritten text recognition ?" In: arXiv. Volume 5. IEEE. 2020, pages 65–70. DOI: 10.1109/icdarw.2019.40083. arXiv: 2012.04954 (cited on page 43).
- [55] I. G. Courville, Y. Bengio, and Aaron. *Deep learning*. Volume 29, 7553. MIT Press, 2016, pages 1–73 (cited on pages 6, 49).
- [56] E. D. Cubuk et al. "AutoAugment: Learning augmentation strategies from data". In: arXiv. 2018, pages 113–123 (cited on page 44).
- [57] X. Cui, V. Goel, and B. Kingsbury. "Data Augmentation for Deep Neural Network Acoustic Modeling". In: *IEEE Transactions on Audio, Speech and Language Processing* 23.9 (2015), pages 1469–1477. ISSN: 15587916. DOI: 10.1109/TASLP. 2015.2438544 (cited on page 148).
- T. Dao et al. "A kernel theory of modern data augmentation". In: 36th International Conference on Machine Learning, ICML 2019. Volume 2019-June. PMLR. 2019, pages 2755–2775. ISBN: 9781510886988. arXiv: 1803.06084 (cited on page 16).
- [59] J. Das Gupta and B. Chanda. "Novel methods for slope and slant correction of offline handwritten text word". In: *Proceedings - 2012 3rd International Conference* on Emerging Applications of Information Technology, EAIT 2012. IEEE. 2012, pages 295–298. ISBN: 9781467318259. DOI: 10.1109/EAIT.2012.6407927 (cited on page 42).
- [60] A. P. Dempster, N. M. Laird, and D. B. Rubin. "Maximum Likelihood from Incomplete Data Via the EM Algorithm ". In: Journal of the Royal Statistical Society: Series B (Methodological) 39.1 (1977), pages 1–22. DOI: 10.1111/j.2517-6161.1977.tb01600.x (cited on page 45).
- [61] L. Deng. "The MNIST database of handwritten digit images for machine learning research". In: *IEEE Signal Processing Magazine* 29.6 (2012), pages 141–142. ISSN: 10535888. DOI: 10.1109/MSP.2012.2211477 (cited on page 80).
- [62] J. Devlin et al. "BERT: Pre-training of deep bidirectional transformers for language understanding". In: arXiv. Minneapolis, Minnesota: Association for Computational Linguistics, 2018, pages 4171–4186. DOI: 10.18653/v1/N19-1423. URL: https: //www.aclweb.org/anthology/N19-1423 (cited on page 74).
- [63] G. Dimauro et al. "Automatic bankcheck processing: A new engineered system". In: International Journal of Pattern Recognition and Artificial Intelligence 11.4 (1997), pages 467–504. ISSN: 02180014. DOI: 10.1142/S0218001497000214 (cited on page 17).
- [64] P. Doetsch, M. Kozielski, and H. Ney. "Fast and Robust Training of Recurrent Neural Networks for Offline Handwriting Recognition". In: Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR. Volume 2014-December. IEEE. 2014, pages 279–284. ISBN: 9781479943340. DOI: 10.1109/ ICFHR.2014.54 (cited on pages 29, 30, 38–40, 45, 46, 77, 146, 147).

- [65] P. Dreuw, G. Heigold, and H. Ney. "Confidence- and margin-based MMI/MPE discriminative training for off-line handwriting recognition". In: *International Journal* on Document Analysis and Recognition 14.3 (2011), pages 273–288. ISSN: 14332833. DOI: 10.1007/s10032-011-0160-x (cited on page 47).
- [66] P. Dreuw et al. "Hierarchical hybrid MLP/HMM or rather MLP features for a discriminatively trained Gaussian HMM: A comparison for offline handwriting recognition". In: *Proceedings International Conference on Image Processing, ICIP.* IEEE. 2011, pages 3541–3544. ISBN: 9781457713033. DOI: 10.1109/ICIP.2011.
 6116480 (cited on pages 40, 45, 51, 52, 146).
- [67] R. O. Duda and P. E. Hart. "Use of the Hough Transformation to Detect Lines and Curves in Pictures". In: *Communications of the ACM* 15.1 (1972), pages 11–15. ISSN: 15577317. DOI: 10.1145/361237.361242 (cited on page 41).
- [68] K. Dutta et al. "Improving CNN-RNN hybrid networks for handwriting recognition". In: Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR. Volume 2018-August. IEEE. 2018, pages 80–85. ISBN: 9781538658758. DOI: 10.1109/ICFHR-2018.2018.00023 (cited on pages 13, 30, 40, 43, 44, 57, 64, 67, 76, 146, 147, 149, 153, 156).
- [69] S. España-Boquera et al. "Improving offline handwritten text recognition with hybrid HMM/ANN models". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 33.4 (2011), pages 767–779. ISSN: 01628828. DOI: 10.1109/ TPAMI.2010.141 (cited on pages 51, 52, 145, 146).
- [70] M. A. Fischler and R. C. Bolles. "Random sample consensus: A Paradigm for Model Fitting with Applications to Image Analysis and Automated Cartography". In: *Communications of the ACM* 24.6 (1981), pages 381–395. ISSN: 15577317. DOI: 10.1145/358669.358692 (cited on page 118).
- [71] W. N. Francis and H. Kuçera. Manual of Information to Accompany a Standard Corpus of Present-day Edited American English, for use with Digital Computers. Brown University, Department of Linguistics, 1964 (cited on page 124).
- [72] V. Frinken and H. Bunke. Continuous handwritten script recognition. 2014. DOI: 10.1007/978-0-85729-859-1_12 (cited on pages 21, 125).
- [73] K. Fukushima. "Neural Network Model for a Mechanism of Pattern Recognition Unaffected By Shift in Position-Neocognitron." In: *Electronics & communications* in Japan 62.10 (1979), pages 11–18. ISSN: 04248368 (cited on page 84).
- M. J. Gales. "Maximum likelihood linear transformations for HMM-based speech recognition". In: Computer Speech and Language 12.2 (1998), pages 75–98. ISSN: 08852308. DOI: 10.1006/csla.1998.0043 (cited on page 44).
- Y. Ganin and V. Lempitsky. "Unsupervised domain adaptation by backpropagation". In: 32nd International Conference on Machine Learning, ICML 2015. Edited by F. Bach and D. Blei. Volume 2. Proceedings of Machine Learning Research. Lille, France: PMLR, 2015, pages 1180–1189. ISBN: 9781510810587. arXiv: 1409.7495. URL: http://proceedings.mlr.press/v37/ganin15.html (cited on page 77).
- [76] X. Glorot and Y. Bengio. "Understanding the difficulty of training deep feedforward neural networks". In: *Journal of Machine Learning Research*. Volume 9. JMLR Workshop and Conference Proceedings. 2010, pages 249–256 (cited on pages 63, 116).

- T. Goswami, Z. Barad, and N. P. Desai. "Text Detection and Recognition in images: A survey". In: arXiv 37.7 (2018), pages 1480–1500. ISSN: 23318422. arXiv: 1803.07278 (cited on page 11).
- [78] A. Graves, S. Fernández, and J. Schmidhuber. "Multi-dimensional recurrent neural networks". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 4668 LNCS. PART 1. 2007, pages 549–558. ISBN: 9783540746898. DOI: 10.1007/978-3-540-74690-4_56. arXiv: 0705.2011 (cited on pages 39, 60, 61).
- [79] A. Graves and J. Schmidhuber. "Offline handwriting recognition with multidimensional recurrent neural networks". In: Advances in Neural Information Processing Systems 21 - Proceedings of the 2008 Conference. 2009, pages 545–552. ISBN: 9781605609492 (cited on pages 3, 29, 38, 39, 48, 53, 61, 67).
- [80] A. Graves et al. "Connectionist temporal classification: Labelling unsegmented sequence data with recurrent neural networks". In: ACM International Conference Proceeding Series. Volume 148. ACM. 2006, pages 369–376. ISBN: 1595933832. DOI: 10.1145/1143844.1143891 (cited on pages 38, 39, 61, 66).
- [81] A. Graves et al. "A novel connectionist system for unconstrained handwriting recognition". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 31.5 (2009), pages 855–868. ISSN: 01628828. DOI: 10.1109/TPAMI.2008.137 (cited on pages 6, 40, 47, 57, 66, 67, 78, 145, 146, 149, 156).
- [82] A. Graves et al. "Unconstrained online handwriting recognition with recurrent neural networks". In: Advances in Neural Information Processing Systems 20 -Proceedings of the 2007 Conference. 2009. ISBN: 160560352X (cited on pages 9, 13, 53).
- [83] K. Greff et al. "LSTM: A Search Space Odyssey". In: *IEEE Transactions on Neural Networks and Learning Systems* 28.10 (2017), pages 2222–2232. ISSN: 21622388.
 DOI: 10.1109/TNNLS.2016.2582924. arXiv: 1503.04069 (cited on page 56).
- [84] E. Grosicki et al. "Results of the RIMES evaluation campaign for handwritten mail processing". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. 2009, pages 941–945. ISBN: 9780769537252. DOI: 10.1109/ICDAR.2009.224 (cited on pages 15, 31, 125).
- [85] P. Grother. (Latin letters) NIST special database 19 handprinted forms and characters database. Technical report. 1995. URL: http://www.nist.gov/srd/upload/ nistsd19.pdf (cited on pages 23, 92).
- [86] N. Gruber and A. Jockisch. "Are GRU Cells More Specific and LSTM Cells More Sensitive in Motive Classification of Text?" In: *Frontiers in Artificial Intelligence* 3 (2020), page 40. ISSN: 2624-8212. DOI: 10.3389/frai.2020.00040 (cited on page 56).
- [87] C. Gulcehre et al. "On Using Monolingual Corpora in Neural Machine Translation". In: arXiv e-prints (2015), arXiv:1503.03535. arXiv: 1503.03535 (cited on page 75).
- [88] S. Günter and H. Bunke. "HMM-based handwritten word recognition: On the optimization of the number of states, training iterations and Gaussian components". In: *Pattern Recognition* 37.10 (2004), pages 2069–2079. ISSN: 00313203. DOI: 10.1016/j.patcog.2004.04.006 (cited on page 46).

- [89] J. D. Gupta and B. Chanda. "An efficient slope and slant correction technique for off-line handwritten text word". In: *Proceedings - 4th International Conference* on Emerging Applications of Information Technology, EAIT 2014. IEEE. 2014, pages 204–208. ISBN: 9781479942725. DOI: 10.1109/EAIT.2014.19 (cited on pages 42, 118).
- [90] I. Guyon et al. "UNIPEN project of on-line data exchange and recognizer benchmarks". In: Proceedings of the 12th IAPR International Conference on Pattern Recognition, Vol. 3-Conference C: Signal Processing (Cat. No. 94CH3440-5). Volume 2. IEEE. 2002, pages 29–33. DOI: 10.1109/icpr.1994.576870 (cited on pages 26, 80).
- [91] K. He et al. "Deep residual learning for image recognition". In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. Volume 2016-December. 2016, pages 770–778. ISBN: 9781467388504. DOI: 10.1109/CVPR.2016.90. arXiv: 1512.03385 (cited on pages 59, 80, 84, 87).
- [92] M. Heidarysafa et al. "An improvement of data classification using random multimodel deep learning (RMDL)". In: *arXiv.* 2018, pages 19–28 (cited on page 91).
- [93] H. Hermansky, D. P. Ellis, and S. Sharma. "Tandem connectionist feature extraction for conventional HMM systems". In: *ICASSP*, *IEEE International Conference* on Acoustics, Speech and Signal Processing - Proceedings. Volume 3. IEEE. 2000, pages 1635–1638. ISBN: 0780362934. DOI: 10.1109/ICASSP.2000.862024 (cited on pages 47, 52).
- [94] G. E. Hinton, S. Osindero, and Y. W. Teh. "A fast learning algorithm for deep belief nets". In: *Neural Computation* 18.7 (2006), pages 1527–1554. ISSN: 08997667.
 DOI: 10.1162/neco.2006.18.7.1527 (cited on page 49).
- [95] G. E. Hinton et al. "Improving neural networks by preventing co-adaptation of feature detectors". In: CoRR abs/1207.0580 (2012). arXiv: 1207.0580 (cited on pages 62, 84).
- [96] S. Hochreiter and J. Schmidhuber. "Long Short-Term Memory". In: Neural Computation 9.8 (1997), pages 1735–1780 (cited on page 55).
- [97] K. Hornik, M. Stinchcombe, and H. White. "Multilayer feedforward networks are universal approximators". In: *Neural Networks* 2.5 (1989), pages 359–366. ISSN: 08936080. DOI: 10.1016/0893-6080(89)90020-8 (cited on page 48).
- J. Hu, M. K. Brown, and W. Turin. "HMM based on-line handwriting recognition". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 18.10 (1996), pages 1039–1045. ISSN: 01628828. DOI: 10.1109/34.541414 (cited on page 46).
- [99] X. Huang et al. "End-to-end sequence labeling via convolutional recurrent neural network with a connectionist temporal classification layer". In: International Journal of Computational Intelligence Systems 13.1 (2020), pages 341–351. ISSN: 18756883. DOI: 10.2991/ijcis.d.200316.001 (cited on pages 30, 40, 65, 149).
- [100] R. Hussain et al. "A comprehensive survey of handwritten document benchmarks: structure, usage and evaluation". In: Eurasip Journal on Image and Video Processing 2015.1 (2015), pages 1–24. ISSN: 16875281. DOI: 10.1186/s13640-015-0102-5 (cited on page 23).

- [101] H. Hyyrö. Explaining and extending the bit-parallel approximate string matching algorithm of Myers. Technical report. Citeseer, 2001. URL: http://citeseerx. ist.psu.edu/viewdoc/download?doi=10.1.1.19.7158%7B%5C&%7Drep=rep1% 7B%5C&%7Dtype=pdf (cited on page 21).
- [102] S. Impedovo, L. Ottaviano, and S. Occhinegro. "Optical character recognition—a survey". In: International Journal of Pattern Recognition and Artificial Intelligence 5.01n02 (1991), pages 1–24 (cited on page 38).
- [103] S. Impedovo, P. S. P. Wang, and H. Bunke. Automatic Bankcheck Processing. Volume 28. World Scientific, 1997, page 382. ISBN: 978-981-02-3223-8 (cited on pages 3, 38).
- [104] R. R. Ingle et al. "A scalable handwritten text recognition system". In: *arXiv*. IEEE. 2019, pages 17–24 (cited on pages 65, 145, 161).
- S. Ioffe and C. Szegedy. "Batch normalization: Accelerating deep network training by reducing internal covariate shift". In: 32nd International Conference on Machine Learning, ICML 2015. Volume 1. ICML'15. JMLR.org, 2015, pages 448–456. ISBN: 9781510810587. arXiv: 1502.03167 (cited on pages 59, 64, 75, 84, 88).
- [106] M. Jaderberg et al. "Synthetic Data and Artificial Neural Networks for Natural Scene Text Recognition". In: CoRR abs/1406.2227 (2014). arXiv: 1406.2227 (cited on pages 76, 125).
- [107] M. Jaderberg et al. "Spatial transformer networks". In: Advances in Neural Information Processing Systems. Volume 2015-January. NIPS'15. Cambridge, MA, USA: MIT Press, 2015, pages 2017–2025. arXiv: 1506.02025 (cited on pages 3, 44, 64).
- [108] K. Jarrett et al. "What is the best multi-stage architecture for object recognition?" In: Proceedings of the IEEE International Conference on Computer Vision. IEEE. 2009, pages 2146–2153. ISBN: 9781424444205. DOI: 10.1109/ICCV.2009.5459469 (cited on page 50).
- [109] Jia Deng et al. "ImageNet: A large-scale hierarchical image database". In: 2009 IEEE conference on computer vision and pattern recognition. Ieee. 2009, pages 248– 255. DOI: 10.1109/cvprw.2009.5206848 (cited on page 86).
- [110] S. Johansson. "The tagged LOB Corpus: User's Manual". In: (1986), page 149 (cited on pages 29, 124).
- [111] P. Kahle et al. "Transkribus A Service Platform for Transcription, Recognition and Retrieval of Historical Documents". In: *Proceedings of the International Conference on Document Analysis and Recognition*, *ICDAR*. Volume 4. IEEE. 2018, pages 19–24. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.307 (cited on page 4).
- [112] N. Kalchbrenner and P. Blunsom. "Recurrent continuous translation models". In: EMNLP 2013 - 2013 Conference on Empirical Methods in Natural Language Processing, Proceedings of the Conference. 2013, pages 1700–1709. ISBN: 9781937284978 (cited on page 72).

- [113] L. Kang et al. "Convolve, Attend and Spell: An Attention-based Sequence-to-Sequence Model for Handwritten Word Recognition". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 11269 LNCS. Springer. 2019, pages 459–472. ISBN: 9783030129385. DOI: 10.1007/978-3-030-12939-2_32 (cited on pages 39, 57, 70, 72, 74, 75, 80, 156, 161).
- [114] L. Kang et al. "Unsupervised adaptation for synthetic-to-real handwritten word recognition". In: arXiv. 2019, pages 3502–3511 (cited on pages 40, 77, 149).
- [115] L. Kang et al. "Pay attention to what you read: Non-recurrent handwritten textline recognition". In: arXiv (2020). ISSN: 23318422. arXiv: 2005.13044 (cited on pages 57, 75–78, 161).
- [116] L. Kang et al. "Candidate fusion: Integrating language modelling into a sequenceto-sequence handwritten word recognition architecture". In: *Pattern Recognition* 112 (2021), page 107790. ISSN: 00313203. DOI: 10.1016/j.patcog.2020.107790. arXiv: 1912.10308 (cited on pages 3, 37, 75, 76, 78, 148, 152–154).
- [117] E. Kavallieratou, N. Fakotakis, and G. Kokkinakis. "Skew angle estimation for printed and handwritten documents using the Wigner-Ville distribution". In: *Image and Vision Computing* 20.11 (2002), pages 813–824. ISSN: 02628856. DOI: 10. 1016/S0262-8856(02)00091-4 (cited on pages 41–43).
- [118] J. D. Kelleher. "A Brief History of Deep Learning". In: *Deep Learning* (2020). DOI: 10.7551/mitpress/11171.003.0007 (cited on page 48).
- [119] C. Kermorvant et al. "From isolated handwritten characters to fields recognition: There's many a slip twixt cup and lip". In: *Proceedings of the International Conference on Document Analysis and Recognition, ICDAR.* IEEE. 2009, pages 1031– 1035. ISBN: 9780769537252. DOI: 10.1109/ICDAR.2009.91 (cited on page 17).
- [120] A. Khan et al. "A survey of the recent architectures of deep convolutional neural networks". In: *arXiv* (2019), pages 1–62. ISSN: 23318422 (cited on page 58).
- F. Kimura et al. "Lexicon directed algorithm for recognition of unconstrained handwritten words". In: *IEICE Transactions on Information and Systems*. Volume E77-D. 7. IEEE. 1994, pages 785–793 (cited on pages 42, 119).
- [122] G. Kirn and V. Govindaraju. "A lexicon driven approach to handwritten word recognition for real-time applications". In: *IEEE Transactions on Pattern Analysis* and Machine Intelligence 19.4 (1997), pages 366–379. ISSN: 01628828. DOI: 10. 1109/34.588017 (cited on page 119).
- [123] R. Kneser and H. Ney. "Improved backing-off for n-gram language modeling". In: 1995 international conference on acoustics, speech, and signal processing. Volume 1. IEEE. 1995, pages 181–184 (cited on page 77).
- [124] A. L. Koerich, R. Sabourin, and C. Y. Suen. "Large vocabulary off-line handwriting recognition: A survey". In: *Pattern Analysis and Applications* 6.2 (2003), pages 97–121. ISSN: 14337541. DOI: 10.1007/s10044-002-0169-3 (cited on page 17).
- [125] J. F. Kolen and S. C. Kremer. Gradient Flow in Recurrent Nets: The Difficulty of Learning LongTerm Dependencies. 2010. DOI: 10.1109/9780470544037.ch14 (cited on page 74).

- M. Kozielski, P. Doetsch, and H. Ney. "Improvements in RWTH's system for offline handwriting recognition". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. IEEE. 2013, pages 935–939. DOI: 10.1109/ICDAR.2013.190 (cited on pages 13, 18, 30, 38, 40, 78, 144, 146).
- [127] M. Kozielski et al. "Open vocabulary handwriting recognition using combined word-level and character-level language models". In: *ICASSP*, *IEEE International Conference on Acoustics, Speech and Signal Processing - Proceedings*. IEEE. 2013, pages 8257–8261. ISBN: 9781479903566. DOI: 10.1109/ICASSP.2013.6639275 (cited on page 52).
- P. Krishnan, K. Dutta, and C. V. Jawahar. "Deep feature embedding for accurate recognition and retrieval of handwritten text". In: *Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR.* Volume 0. IEEE. 2016, pages 289–294. ISBN: 9781509009817. DOI: 10.1109/ICFHR.2016.0062 (cited on pages 59, 68, 146, 147, 156).
- P. Krishnan, K. Dutta, and C. V. Jawahar. "Word spotting and recognition using deep embedding". In: Proceedings 13th IAPR International Workshop on Document Analysis Systems, DAS 2018. IEEE. 2018, pages 1–6. ISBN: 9781538633465. DOI: 10.1109/DAS.2018.70 (cited on page 44).
- [130] P. Krishnan and C. V. Jawahar. "Generating Synthetic Data for Text Recognition". In: ArXiv abs/1608.04224 (2016). arXiv: 1608.04224 (cited on page 76).
- P. Krishnan and C. V. Jawahar. "Matching handwritten document images". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 9905 LNCS. Springer. 2016, pages 766–782. ISBN: 9783319464473. DOI: 10.1007/978-3-319-46448-0_46. arXiv: 1605.05923 (cited on pages 59, 68).
- [132] A. Krizhevsky, I. Sutskever, and G. E. Hinton. "ImageNet classification with deep convolutional neural networks". In: *Communications of the ACM*. Volume 60. 6. 2017, pages 84–90. DOI: 10.1145/3065386 (cited on pages 6, 7, 48, 80, 84).
- [133] Y. Le Cun, L. Bottou, and Y. Bengio. "Reading checks with multilayer graph transformer networks". In: ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing - Proceedings. Volume 1. IEEE. 1997, pages 151–154. DOI: 10.1109/icassp.1997.599580 (cited on page 59).
- Y. LeCun et al. "Backpropagation Applied to Handwritten Zip Code Recognition". In: Neural Computation 1.4 (1989), pages 541–551. ISSN: 0899-7667. DOI: 10.1162/ neco.1989.1.4.541 (cited on pages 37, 48, 57, 59).
- Y. LeCun et al. "Gradient-based learning applied to document recognition". In: Proceedings of the IEEE 86.11 (1998), pages 2278–2323. ISSN: 00189219. DOI: 10. 1109/5.726791 (cited on pages 25, 37, 59, 83).
- [136] C. Y. Lee and S. Osindero. "Recursive Recurrent Nets with Attention Modeling for OCR in the Wild". In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. Volume 2016-December. 2016, pages 2231–2239. ISBN: 9781467388504. DOI: 10.1109/CVPR.2016.245. arXiv: 1603.03101 (cited on page 68).

- S. Lessmann et al. "Benchmarking state-of-the-art classification algorithms for credit scoring: An update of research". In: *European Journal of Operational Research* 247.1 (2015), pages 124–136. ISSN: 03772217. DOI: 10.1016/j.ejor.2015.
 05.030 (cited on page 6).
- [138] V. Levenshtein. "Binary codes capable of correcting deletions, insertions, and reversals". In: *Soviet Physics Doklady*. Volume 10. 8. 1966, pages 707–710 (cited on pages 17, 18, 21, 125, 126).
- [139] H. Li et al. "Reading car license plates using deep neural networks". In: Image and Vision Computing 72 (2018), pages 14-23. ISSN: 02628856. DOI: 10.1016/ j.imavis.2018.02.002. URL: https://www.sciencedirect.com/science/ article/pii/S0262885618300155 (cited on page 64).
- [140] T. Y. Lin et al. "Focal loss for dense object detection". In: arXiv. 2017, pages 2980– 2988 (cited on page 75).
- [141] L. Liu et al. "Deep Learning for Generic Object Detection: A Survey". In: International Journal of Computer Vision 128.2 (2020), pages 261–318. ISSN: 15731405.
 DOI: 10.1007/s11263-019-01247-4. arXiv: 1809.02165 (cited on page 11).
- [142] Q. Liu, L. Wang, and Q. Huo. "A Study on Effects of Implicit and Explicit Language Model Information for DBLSTM-CTC Based Handwriting Recognition". In: arXiv. IEEE. 2020, pages 461–465 (cited on page 22).
- [143] L. van der Maaten. "A new benchmark dataset for handwritten character recognition". In: Technical Report. Tilburg University, Tilburg, The Netherlands (2009), pages 2-5. URL: http://www.tilburguniversity.edu/research/institutesand-research-groups/ticc/research/technicalreports/TR2009002.pdf (cited on pages 26, 80, 96, 97).
- [144] N. Manisha, E. Sreenivasa, and Y. K. "Role of Offline Handwritten Character Recognition System in Various Applications". In: International Journal of Computer Applications 135.2 (2016), pages 30–33. DOI: 10.5120/ijca2016908349 (cited on page 3).
- [145] U. V. Marti and H. Bunke. "Using a statistical language model to improve the performance of an HMM-based cursive handwriting recognition system". In: *International Journal of Pattern Recognition and Artificial Intelligence*. Volume 15.
 1. World Scientific, 2001, pages 65–90. DOI: 10.1142/S0218001401000848 (cited on pages 42, 46, 77).
- [146] U. V. Marti and H. Bunke. "The IAM-database: An English sentence database for offline handwriting recognition". In: International Journal on Document Analysis and Recognition 5.1 (2003), pages 39–46. ISSN: 14332833. DOI: 10.1007/ s100320200071 (cited on pages 29, 38, 76, 125, 144).
- [147] V. Mazzia, F. Salvetti, and M. Chiaberge. "Efficient-CapsNet: Capsule Network with Self-Attention Routing". In: ArXiv abs/2101.12491 (2021). arXiv: 2101.
 12491 (cited on page 91).
- [148] W. S. McCulloch and W. H. Pitts. "A logical calculus of the ideas immanent in nervous activity". In: Systems Research for Behavioral Science: A Sourcebook 5.4 (2017), pages 93–96. ISSN: 00928240. DOI: 10.7551/mitpress/12274.003.0011 (cited on pages 47, 48).

- [149] F. Menasri et al. "The A2iA French handwriting recognition system at the Rimes-ICDAR2011 competition". In: *Document Recognition and Retrieval XIX*. Volume 8297. International Society for Optics and Photonics. 2012, 82970Y. ISBN: 9780819489449.
 DOI: 10.1117/12.911981 (cited on pages 61, 62, 125, 148, 152).
- [150] J. Michael et al. "Evaluating sequence-to-sequence models for handwritten text recognition". In: *arXiv*. IEEE. 2019, pages 1286–1293 (cited on pages 43, 75).
- [151] A. Mikołajczyk and M. Grochowski. "Data augmentation for improving deep learning in image classification problem". In: 2018 International Interdisciplinary PhD Workshop, IIPhDW 2018. IEEE. 2018, pages 117–122. ISBN: 9781538661437. DOI: 10.1109/IIPHDW.2018.8388338 (cited on page 148).
- [152] M. Minsky and S. Papert. "Perceptrons." In: (1969) (cited on page 48).
- B. A. Moore and G. J. McLachlan. Discriminant Analysis and Statistical Pattern Recognition. Volume 157. 1. John Wiley \& Sons, 1994, page 162. DOI: 10.2307/ 2983518 (cited on page 52).
- [154] M. C. Mukkamala and M. Hein. "Variants of RMSProp and adagrad with logarithmic regret bounds". In: arXiv. PMLR. 2017, pages 2545–2553 (cited on page 90).
- [155] A. Newell. "A Tutorial on Speech Understanding Systems". In: Speech Recognition: Invited Papers Presented at the 1974 IEEE Symposium (1975), pages 3–54 (cited on page 13).
- [156] O. Okun, M. Pietikinen, and J. Sauvola. "Document skew estimation without angle range restriction". In: International Journal on Document Analysis and Recognition 2.2-3 (1999), pages 132–144. ISSN: 1433-2833. DOI: 10.1007/s100320050043 (cited on page 42).
- [157] N. Otsu. "A Threshold Selection Method from Gray-Level Histograms". In: *IEEE Transaction on Systems, Man and Cybernetics* 20.1 (1979), pages 62–66 (cited on page 41).
- U. Pal and B. B. Chaudhuri. "An improved document skew angle estimation technique". In: *Pattern Recognition Letters* 17.8 (1996), pages 899–904. ISSN: 01678655.
 DOI: 10.1016/0167-8655(96)00042-6 (cited on page 41).
- [159] A. Papandreou and B. Gatos. "Slant estimation and core-region detection for hand-written Latin words". In: *Pattern Recognition Letters* 35.1 (2014), pages 16–22. ISSN: 01678655. DOI: 10.1016/j.patrec.2012.08.005 (cited on page 41).
- [160] M. Pastor, A. Toselli, and E. Vidai. "Projection profile based algorithm for slant removal". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 3212. Springer. 2004, pages 183–190. ISBN: 3540232400. DOI: 10.1007/978-3-540-30126-4_23 (cited on pages 41–43).
- [161] M. Pastor et al. "Improving handwritten off-line text slant correction". In: Proceedings of the 6th IASTED International Conference on Visualization, Imaging, and Image Processing, VIIP 2006. 2006, pages 389–394. ISBN: 0889865981 (cited on page 42).

- [162] J. Pastor-Pellicer et al. "A combined Convolutional Neural Network and Dynamic Programming approach for text line normalization". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 2015-November. IEEE. 2015, pages 341–345. ISBN: 9781479918058. DOI: 10.1109/ ICDAR.2015.7333780 (cited on page 43).
- [163] H. Pesch et al. "Analysis of preprocessing techniques for Latin handwriting recognition". In: Proceedings International Workshop on Frontiers in Handwriting Recognition, IWFHR. IEEE. 2012, pages 280–284. ISBN: 9780769547749. DOI: 10. 1109/ICFHR.2012.179 (cited on pages 13, 41).
- [164] V. Pham et al. "Dropout Improves Recurrent Neural Networks for Handwriting Recognition". In: Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR. Volume 2014-December. IEEE. 2014, pages 285–290. ISBN: 9781479943340. DOI: 10.1109/ICFHR.2014.55. arXiv: 1312.4569 (cited on pages 53, 62, 67, 116, 146, 147, 153, 154).
- [165] R. Plamondon and S. N. Srihari. "On-line and off-line handwriting recognition: A comprehensive survey". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 22.1 (2000), pages 63–84. ISSN: 01628828. DOI: 10.1109/34.824821 (cited on pages 2, 3, 11, 37, 44).
- J. Poulos and R. Valle. "Character-based handwritten text transcription with attention networks". In: Neural Computing and Applications (2021). ISSN: 14333058.
 DOI: 10.1007/s00521-021-05813-1. arXiv: 1712.04046 (cited on page 37).
- [167] D. Povey et al. "The Kaldi Speech Recognition". In: IEEE 2011 Workshop on Automatic Speech Recognition and Understanding. January. IEEE Signal Processing Society. 2011. ISBN: 978-1-4673-0366-8. URL: http://kaldi.sf.net/ (cited on page 45).
- [168] D. M. W. Powers. "Evaluation: from precision, recall and F-measure to ROC, informedness, markedness and correlation". In: arXiv preprint arXiv:2010.16061 (2020). arXiv: 2010.16061 (cited on page 19).
- [169] A. Poznanski and L. Wolf. "CNN-N-Gram for Handwriting Word Recognition". In: The IEEE Conference on Computer Vision and Pattern Recognition (CVPR). 2016 (cited on pages 30, 44, 59, 68, 146–149, 153, 156).
- [170] J. Puigcerver. "Are Multidimensional Recurrent Layers Really Necessary for Handwritten Text Recognition?" In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 1. IEEE. 2017, pages 67–72. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.20 (cited on pages 13, 15, 22, 48, 57, 59, 63, 64, 67, 80, 122, 145–147, 155).
- [171] G. Pulford. "The viterbi algorithm". In: *IET Seminar Digest* 2006.11359 (2006), pages 53-65. DOI: 10.1049/ic:20060556 (cited on page 39).
- [172] L. R. Rabiner. "A Tutorial on Hidden Markov Models and Selected Applications in Speech Recognition". In: *Proceedings of the IEEE* 77.2 (1989), pages 257–286.
 ISSN: 15582256. DOI: 10.1109/5.18626 (cited on page 45).
- [173] C. Raffel et al. "Online and linear-time attention by enforcing monotonic alignments". In: arXiv. PMLR. 2017, pages 2837–2846 (cited on page 75).

- [174] D. E. Rumelhart, G. E. Hinton, and R. J. Williams. Learning Internal Representations by Error Propagation. Technical report. California Univ San Diego La Jolla Inst for Cognitive Science, 2013, pages 399–421. DOI: 10.1016/B978-1-4832-1446-7.50035-2 (cited on page 48).
- [175] D. E. Rumelhart, G. E. Hinton, and R. J. Williams. "Learning representations by back-propagating errors". In: *Nature* 323.6088 (1986), pages 533–536. ISSN: 00280836. DOI: 10.1038/323533a0 (cited on pages 48, 52).
- [176] A. L. Samuel. "Some studies in machine learning using the game of checkers". In: *IBM Journal of Research and Development* 44.1-2 (2000), pages 207–219. ISSN: 00188646. DOI: 10.1147/rd.441.0206 (cited on page 5).
- [177] J. A. Sanchez et al. "Handwritten text recognition for historical documents in the transcriptorium project". In: ACM International Conference Proceeding Series. ACM. 2014, pages 111–117. ISBN: 9781450325882. DOI: 10.1145/2595188.2595193 (cited on page 3).
- B. Sankaran et al. "Temporal Attention Model for Neural Machine Translation". In: CoRR abs/1608.02927 (2016). arXiv: 1608.02927 (cited on page 75).
- [179] J. Sauvola and M. Pietikainen. "Adaptive document image binarization". In: *Pattern Recognition* 33.2 (2000), pages 225–236. ISSN: 00313203. DOI: 10.1016/S0031-3203(99)00055-2 (cited on pages 41, 118).
- K. M. Sayre. "Machine recognition of handwritten words: A project report". In: *Pattern Recognition* 5.3 (1973), pages 213–228. ISSN: 00313203. DOI: 10.1016/ 0031-3203(73)90044-7 (cited on page 3).
- [181] H. Scheidl, S. Fiel, and R. Sablatnig. "Word beam search: A connectionist temporal classification decoding algorithm". In: *Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR*. Volume 2018-August. IEEE. 2018, pages 253–258. ISBN: 9781538658758. DOI: 10.1109/ICFHR-2018.2018.00052 (cited on pages 67, 75).
- [182] J. Schenk and G. Rigoll. "Novel Hybrid NN / HMM Modelling Techniques for Online Handwriting Recognition". In: Int'l Workshop on Frontiers in Handwriting Recognition. 2006, pages 619–623 (cited on page 57).
- [183] M. Schenkel, I. Guyon, and D. Henderson. "On-line cursive script recognition using time delay neural networks and hidden markov models". In: *ICASSP*, *IEEE International Conference on Acoustics, Speech and Signal Processing - Proceedings* 2.4 (1994), pages II637–II640. ISSN: 15206149. DOI: 10.1109/ICASSP.1994.389575 (cited on page 46).
- [184] J. Schmidhuber. "Deep Learning in neural networks: An overview". In: Neural Networks 61 (2015), pages 85-117. ISSN: 18792782. DOI: 10.1016/j.neunet. 2014.09.003. arXiv: 1404.7828. URL: https://www.sciencedirect.com/ science/article/pii/S0893608014002135 (cited on page 49).
- [185] L. R. Schomaker and L. Vuurpijl. Forensic Writer Identification: A Benchmark Data Set and a Comparison of Two Systems. NICI (Nijmegen Institute of Cognitive Information), Katholieke Universiteit, 2000, pages 1–22 (cited on page 26).
- [186] M. Schuster and K. K. Paliwal. "Bidirectional recurrent neural networks". In: *IEEE Transactions on Signal Processing* 45.11 (1997), pages 2673–2681. ISSN: 1053587X.
 DOI: 10.1109/78.650093 (cited on pages 53, 60).

- [187] A. W. Senior and A. J. Robinson. "An off-line cursive handwriting recognition system". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 20.3 (1998), pages 309–321. ISSN: 01628828. DOI: 10.1109/34.667887 (cited on page 57).
- [188] D. Shanmugam et al. "When and why test-time augmentation works". In: *arXiv* (2020), arXiv-2011. ISSN: 23318422. arXiv: 2011.11156 (cited on page 148).
- [189] B. Shi, X. Bai, and C. Yao. "An End-to-End Trainable Neural Network for Image-Based Sequence Recognition and Its Application to Scene Text Recognition". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 39.11 (2017), pages 2298–2304. ISSN: 01628828. DOI: 10.1109/TPAMI.2016.2646371. arXiv: 1507.05717 (cited on pages 3, 63, 64).
- B. Shi et al. "Robust Scene Text Recognition with Automatic Rectification". In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. Volume 2016-December. 2016, pages 4168–4176. ISBN: 9781467388504. DOI: 10.1109/CVPR.2016.452. arXiv: 1603.03915 (cited on page 68).
- P. Y. Simard, D. Steinkraus, and J. C. Platt. "Best practices for convolutional neural networks applied to visual document analysis". In: *Proceedings of the International Conference on Document Analysis and Recognition, ICDAR*. Volume 2003-January. 2003. Citeseer. 2003, pages 958–963. ISBN: 0769519601. DOI: 10.1109/ICDAR.2003.1227801 (cited on page 44).
- [192] K. Simonyan and A. Zisserman. "Very deep convolutional networks for large-scale image recognition". In: 3rd International Conference on Learning Representations, ICLR 2015 - Conference Track Proceedings. Edited by Y. Bengio and Y. LeCun. 2015. arXiv: 1409.1556 (cited on pages 59, 68, 75, 80, 83, 86).
- [193] A. Slomson. "Digital Computers Applied to Games". In: Alan Turing: His Work and Impact (2013), pages 623–650. DOI: 10.1016/B978-0-12-386980-7.50024-1 (cited on page 5).
- S. N. Srihari. "Recognition of handwritten and machine-printed text for postal address interpretation". In: *Pattern Recognition Letters* 14.4 (1993), pages 291–302. ISSN: 01678655. DOI: 10.1016/0167-8655(93)90095-U (cited on page 17).
- [195] R. K. Srivastava, K. Greff, and J. Schmidhuber. "Training very deep networks". In: Advances in Neural Information Processing Systems. Edited by C. Cortes et al. Volume 2015-January. Curran Associates, Inc., 2015, pages 2377-2385. arXiv: 1507.06228. URL: https://proceedings.neurips.cc/paper/2015/file/215a71a12769b056c3c32e7299f1c5ed-Paper.pdf (cited on pages 60, 87).
- T. Starner et al. "On-line cursive handwriting recognition using speech recognition methods". In: ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing Proceedings. Volume 5. IEEE. 1994, pages V-125. DOI: 10. 1109/icassp.1994.389432 (cited on page 46).
- [197] T. Strauß et al. "CITlab ARGUS for keyword search in historical handwritten documents". In: CEUR Workshop Proceedings 1609 (2016), pages 399–412. ISSN: 16130073 (cited on pages 62, 63).

- [198] T. Street. "An Online Cursive Script Recognition System Using Recurrent Error Propagation Networks". In: International Workshop on Frontiers in Handwriting Recognition. 1993, pages 132–141 (cited on page 57).
- B. Stuner, C. Chatelain, and T. Paquet. "Handwriting recognition using cohort of LSTM and lexicon verification with extremely large lexicon". In: *Multimedia Tools and Applications* 79.45-46 (2020), pages 34407–34427. ISSN: 15737721. DOI: 10.1007/s11042-020-09198-6. arXiv: 1612.07528 (cited on pages 124, 125, 148).
- [200] J. Sueiras et al. "Using a synthetic character database for training deep learning models applied to offline handwritten recognition". In: Advances in Intelligent Systems and Computing. Volume 557. Springer. 2017, pages 299–308. ISBN: 9783319534794. DOI: 10.1007/978-3-319-53480-0_30 (cited on page 147).
- [201] J. Sueiras et al. "Offline continuous handwriting recognition using sequence to sequence neural networks". In: *Neurocomputing* 289 (2018), pages 119–128. ISSN: 18728286. DOI: 10.1016/j.neucom.2018.02.008 (cited on pages 30, 31, 39, 57, 70, 75, 147).
- [202] C. Y. Suen et al. "Computer Recognition of Unconstrained Handwritten Numerals". In: *Proceedings of the IEEE* 80.7 (1992), pages 1162–1180. ISSN: 15582256.
 DOI: 10.1109/5.156477 (cited on page 38).
- [203] D. Suryani, P. Doetsch, and H. Ney. "On the benefits of convolutional neural network combinations in offline handwriting recognition". In: Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR. Volume 0. IEEE. 2016, pages 193–198. ISBN: 9781509009817. DOI: 10.1109/ICFHR.2016.0046 (cited on page 63).
- [204] I. Sutskever, O. Vinyals, and Q. V. Le. "Sequence to sequence learning with neural networks". In: Advances in Neural Information Processing Systems. Volume 4. January. 2014, pages 3104–3112. arXiv: 1409.3215 (cited on pages 6, 7, 68, 69, 105).
- [205] C. Syms. Principal components analysis. 69. Sage, 2018, pages 566–573. ISBN: 9780444641304. DOI: 10.1016/B978-0-12-409548-9.11152-2 (cited on pages 47, 52).
- [206] C. Szegedy et al. "Rethinking the Inception Architecture for Computer Vision". In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. Volume 2016-December. 2016, pages 2818–2826. ISBN: 9781467388504. DOI: 10.1109/CVPR.2016.308. arXiv: 1512.00567 (cited on pages 72, 75).
- [207] N. Tajbakhsh et al. "Convolutional Neural Networks for Medical Image Analysis: Full Training or Fine Tuning?" In: *IEEE Transactions on Medical Imaging* 35.5 (2016), pages 1299–1312. ISSN: 1558254X. DOI: 10.1109/TMI.2016.2535302. arXiv: 1706.00712 (cited on page 161).
- [208] M. Tan and Q. V. Le. "EfficientNet: Rethinking model scaling for convolutional neural networks". In: *arXiv.* PMLR. 2019, pages 6105–6114 (cited on page 160).

- [209] A. H. Toselli et al. "Integrated handwriting recognition and interpretation using finite-state models". In: International Journal of Pattern Recognition and Artificial Intelligence 18.4 (2004), pages 519–539. ISSN: 02180014. DOI: 10.1142/ S0218001404003344 (cited on page 43).
- [210] E. Trentin and M. Gori. "A survey of hybrid ANN/HMM models for automatic speech recognition". In: *Neurocomputing* 37.1-4 (2001), pages 91–126. ISSN: 09252312. DOI: 10.1016/S0925-2312(00)00308-8 (cited on pages 45, 46).
- [211] Unesco. *Hamburg Declaration*. 2021. URL: http://www.unesco.org/education/ uie/confintea/pdf/con5eng.pdf' (cited on page 1).
- [212] J. Vandekerckhove, D. Matzke, and E.-J. Wagenmakers. Model comparison and the principle of parsimony. eScholarship, University of California, 2014, pages 1– 29. ISBN: 978-0-19-995799-6. URL: http://www.ejwagenmakers.com/inpress/ VandekerckhoveEtAlinpress.pdf (cited on pages 116, 140).
- [213] A. Vaswani et al. "Attention is all you need". In: arXiv. Edited by I. Guyon et al. Volume 30. Curran Associates, Inc., 2017. URL: https://proceedings.neurips. cc/paper/2017/file/3f5ee243547dee91fbd053c1c4a845aa-Paper.pdf (cited on pages 57, 74).
- [214] M. Villegas, V. Romero, and J. A. Sánchez. "On the modification of binarization algorithms to retain grayscale information for handwritten text recognition". In: Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). Volume 9117. Springer. 2015, pages 208–215. ISBN: 9783319193892. DOI: 10.1007/978-3-319-19390-8_24 (cited on pages 41, 115, 118).
- [215] A. Vinciarelli and J. Luettin. "A new normalization technique for cursive hand-written words". In: *Pattern Recognition Letters* 22.9 (2001), pages 1043–1050. ISSN: 01678655. DOI: 10.1016/S0167-8655(01)00042-3 (cited on page 42).
- [216] P. Voigtlaender, P. Doetsch, and H. Ney. "Handwriting recognition with large multidimensional long short-term memory recurrent neural networks". In: *Proceedings of International Conference on Frontiers in Handwriting Recognition, ICFHR.* Volume 0. IEEE. 2016, pages 228–233. ISBN: 9781509009817. DOI: 10.1109/ICFHR. 2016.0052 (cited on pages 30, 39, 62, 63, 67).
- [217] L. Wan et al. "Regularization of neural networks using DropConnect". In: 30th International Conference on Machine Learning, ICML 2013. PART 3. PMLR. 2013, pages 2095–2103 (cited on page 91).
- [218] J. Wang and X. Hu. "Gated recurrent convolution neural network for OCR". In: Advances in Neural Information Processing Systems. Volume 2017-December. 2017, pages 335–344 (cited on page 65).
- [219] K. Wang, B. Babenko, and S. Belongie. "End-to-end scene text recognition". In: Proceedings of the IEEE International Conference on Computer Vision. IEEE. 2011, pages 1457–1464. ISBN: 9781457711015. DOI: 10.1109/ICCV.2011.6126402 (cited on pages 3, 11).
- [220] S. Wang. "A comprehensive survey of data mining-based accounting-fraud detection research". In: 2010 International Conference on Intelligent Computation Technology and Automation, ICICTA 2010. Volume 1. 2010, pages 50–53. ISBN: 9780769540771. DOI: 10.1109/ICICTA.2010.831 (cited on page 6).

- [221] P. Werbos. "Beyond Regression : New Tools for Prediction and Analysis in the Behavioral". In: *Ph. D. dissertation, Harvard University* January 1974 (2018) (cited on page 48).
- [222] C. Wigington et al. "Data Augmentation for Recognition of Handwritten Words and Lines Using a CNN-LSTM Network". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR. Volume 1. IEEE. 2017, pages 639–645. ISBN: 9781538635865. DOI: 10.1109/ICDAR.2017.110 (cited on pages 43, 44, 147–149, 153, 156).
- [223] C. Wigington et al. "Start, follow, read: End-to-end full-page handwriting recognition". In: Proceedings of the European Conference on Computer Vision (ECCV). 2018, pages 367–383 (cited on pages 60, 64, 67, 161).
- [224] R. J. Williams and D. Zipser. "A Learning Algorithm for Continually Running Fully Recurrent Neural Networks". In: *Neural Computation* 1.2 (1989), pages 270– 280. ISSN: 0899-7667. DOI: 10.1162/neco.1989.1.2.270 (cited on pages 71, 111, 139).
- [225] S. Wu, W. Wei, and L. Zhang. "Comparison of machine learning algorithms for handwritten digit recognition". In: *Communications in Computer and Information Science*. Volume 874. Perth, Australia. 2018, pages 532–542. DOI: 10.1007/978– 981-13-1651-7_47 (cited on pages 49, 59, 80, 83–85).
- [226] K. Xu et al. "Show, attend and tell: Neural image caption generation with visual attention". In: 32nd International Conference on Machine Learning, ICML 2015. Volume 3. PMLR. 2015, pages 2048–2057. ISBN: 9781510810587. arXiv: 1502.03044 (cited on page 68).
- [227] A. El-Yacoubi et al. "An HMM-based approach for off-line unconstrained handwritten word modeling and recognition". In: *IEEE Transactions on Pattern Anal*ysis and Machine Intelligence 21.8 (1999), pages 752–760. ISSN: 01628828. DOI: 10.1109/34.784288 (cited on page 46).
- [228] D. You and G. Kim. "Slant correction of handwritten strings based on structural properties of Korean characters". In: *Proceedings - International Workshop on Frontiers in Handwriting Recognition, IWFHR*. IEEE. 2002, pages 467–472. ISBN: 0769516920. DOI: 10.1109/IWFHR.2002.1030954 (cited on pages 42, 119).
- [229] S. Young et al. "The HTK book". In: Cambridge University Engineering Department 3.July 2000 (2002), page 175. ISSN: 1098-6596. arXiv: arXiv:1011.1669v3 (cited on page 45).
- [230] M. Yousef and T. E. Bishop. "OrigamiNet: Weakly-Supervised, Segmentation-Free, One-Step, Full Page Text Recognition by learning to unfold". In: arXiv. 2020, pages 14710–14719 (cited on pages 60, 67).
- [231] M. Yousef, K. F. Hussain, and U. S. Mohammed. "Accurate, data-efficient, unconstrained text recognition with convolutional neural networks". In: arXiv 108 (2018), page 107482. ISSN: 23318422. arXiv: 1812.11894 (cited on pages 44, 59).
- [232] A. Yuan et al. "Offline handwritten English character recognition based on convolutional neural network". In: Proceedings - 10th IAPR International Workshop on Document Analysis Systems, DAS 2012. IEEE. 2012, pages 125–129. ISBN: 9780769546612. DOI: 10.1109/DAS.2012.61 (cited on pages 3, 80).

- [233] A. Zahour et al. "Text Line segmentation of historical Arabic documents". In: Proceedings of the International Conference on Document Analysis and Recognition, ICDAR 1.2-4 (2007), pages 138–142. ISSN: 15205363. DOI: 10.1109/ICDAR.2007. 4378691 (cited on page 11).
- [234] N. M. Zaitoun and M. J. Aqel. "Survey on Image Segmentation Techniques". In: Procedia Computer Science 65 (2015), pages 797-806. ISSN: 18770509. DOI: 10. 1016/j.procs.2015.09.027. URL: https://www.sciencedirect.com/science/ article/pii/S1877050915028574 (cited on page 11).
- [235] F. Zamora-Martínez et al. "Neural network language models for off-line handwriting recognition". In: *Pattern Recognition* 47.4 (2014), pages 1642–1652. ISSN: 00313203. DOI: 10.1016/j.patcog.2013.10.020 (cited on pages 47, 78).
- [236] D. Zhang et al. "Learning to share: Simultaneous parameter tying and sparsification in deep learning". In: 6th International Conference on Learning Representations, ICLR 2018 - Conference Track Proceedings. 2018 (cited on page 109).
- [237] X. Zhang, J. Zhao, and Y. Lecun. "Character-level convolutional networks for text classification". In: Advances in Neural Information Processing Systems. Volume 2015-January. 2015, pages 649–657. arXiv: 1509.01626 (cited on page 80).
- [238] Y. Zhang et al. "Sequence-to-sequence domain adaptation network for robust text image recognition". In: Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition. Volume 2019-June. 2019, pages 2735– 2744. ISBN: 9781728132938. DOI: 10.1109/CVPR.2019.00285 (cited on page 77).
- [239] Y. T. Zhou and R. Chellappa. "Computation of optical flow using a neural network". In: *ICNN*. 1988, pages 71–78. DOI: 10.1109/icnn.1988.23914 (cited on page 58).
- [240] M. Zimmermann, J. C. Chappelier, and H. Bunke. "Offline grammar-based recognition of handwritten sentences". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* 28.5 (2006), pages 818–821. ISSN: 01628828. DOI: 10.1109/ TPAMI.2006.103 (cited on page 46).
- [241] M. Zimmermann and H. Bunke. "Automatic segmentation of the IAM off-line database for handwritten english text". In: *Proceedings - International Conference* on Pattern Recognition. Volume 16. 4. IEEE. 2002, pages 35–39. DOI: 10.1109/ icpr.2002.1047394 (cited on page 31).

Index

Adam algorithm 116
Affine transformation 16
Ascenders and descenders regions 15, 43
Automatic speech recognition 6, 44, 46
Back-propagation through time 53
Baum-Welch algorithm 45
Bidirectional recurrent neural network 53,
60, 134
BPTT (see Back-propagation through
time)
BRNN (see Bidirectional recurrent neural
networks)
Calligraphy zones 13, 43
Canonical correlation analysis 59
CER (see Metrics: Character error rate)
CNN (see Convolutional neural network)
Company
A2IA 4, 62
Google 4
Microsoft 4
Computer vision 43, 61
Conferences
DAS 2
ICDAR 2, 32
ICFHR 2, 32
Connectionist temporal classification 39,
61, 63, 65
Convolutional neural network 38, 57, 63,
83, 130
Corpus
Brown 124
LOB 29, 124
CTC (see Connectionist temporal
classification)
Data augmentation 11, 16, 43, 122, 138
Databases
COUT 26, 98
IAM 29, 38, 64

MNIST	25,38,91
NIST	23, 92
Osborne	32
RIMES	31
TICH	26, 96
UNIPEN	26
Deep learning	6
Dropout	62, 116
Expectation-maximization	45
Fourier transformation	42
Gabor filter	42
Gated recurrent unit	56, 135
Gaussian mixture models	46, 52
Glorot normal	116
GMM (see Gaussian mixture GRU (see Gated recurrent un	model) it)
Hidden Markov model	39, 44, 51
Histograms of oriented gradie	nts 38
HMM (see Hidden Markov me	odel)
Image denoising	14, 40, 118
Image preprocessing	13, 39
Image segmentation	11,127,129
Journals	
IJDAR	2
Lavout analysis	10
LDA (see Linear discriminant	analysis)
LeNet	83. 127
Levenshtein distance	17. 18. 21. 124
Lexicon	12. 17. 124
Linear discriminant analysis	52
Long short-term memory netw	vork 55, 61,
135	
LSTM (see Long short-term r network)	nemory
Machine learning	5
0	-

Maximum likelihood estimation	45
MDLSTM (see Multidimensiona	l LSTM)
MDRNN (see Multidimensional	recurrent
neural network)	
Metrics	
Accuracy	18, 91
AUC	20
Character error rate	21,125
Confusion matrix	20
Error rate	18
F1 score	19
False positive rate	20
Recall	19
True positive rate	20
Word error rate	21,64,125
MLP (see Multilayer preceptron))
Monotonic chunkwise attention	75
Multidimensional LSTM	61,67
Multidimensional recurrent neur	al network
39, 53, 60	
Multilayer perceptron	15,39,49
Multinomial distribution	18
n-gram language models	18,77
Object detection	11
Offline handwriting recognition	9
Online handwriting recognition	9
Out of vocabulary	17,68
PCA (see Principal component a	analysis)
PNG	24, 29
Principal component analysis	38,47,52

Progressive	probabilistic	Hough	$\operatorname{transform}$
41			

Random sample consensus	118			
RANSAC (see Random sample consensus)				
Recurrent neural network	15, 38, 63			
ResNet	84			
RMSProp algorithm	91			
RNN (see Recurrent neural netw	vork)			
ROC curve	20			
Scene text recognition	3			
seq2seq (see Sequence-to-sequen	ce)			
Sequence learning	61			
Sequence-to-sequence	6, 7, 39, 68			
Slant correction	15, 42, 119			
Slope correction	14, 41, 118			
Softmax	70, 112			
Software				
Apache Lucene	17			
Imgtxtenh	116			
Open CV	115			
Scikit-learn	25			
Tensorflow	25,115			
Spatial transformer network	44, 64			
Speech recognition	45			
Support vector machines	49			
SVM (see Support vector machi	nes)			
Transkribus	4			
VGG	83			
Viterbi algorithm	39, 45			
0	,			

WER (see Metrics: Word error rate)